# UC Berkeley UC Berkeley Electronic Theses and Dissertations

### Title

Polynesian Subsistence, Nutrient Flows, and Long-Term Socio-Ecosystem Dynamics: Insights from Stable Isotope Analysis of the Pacific Rat (Rattus exulans) in Island Foodwebs

**Permalink** https://escholarship.org/uc/item/9wd8764f

**Author** Swift, Jillian Amy

**Publication Date** 2016

Peer reviewed|Thesis/dissertation

Polynesian Subsistence, Nutrient Flows, and Long-Term Socio-Ecosystem Dynamics: Insights from Stable Isotope Analysis of the Pacific Rat (*Rattus exulans*) in Island Foodwebs

By

Jillian Amy Swift

A dissertation submitted in partial satisfaction of the

requirements for the degree of

Doctor of Philosophy

in

Anthropology

in the

Graduate Division

of the

University of California, Berkeley

Committee in charge:

Chancellor's Professor Emeritus Patrick V. Kirch, Chair Professor Christine Hastorf Professor Anthony D. Barnosky

Spring 2016

Polynesian Subsistence, Nutrient Flows, and Long-Term Socio-Ecosystem Dynamics: Insights from Stable Isotope Analysis of the Pacific Rat (*Rattus exulans*) in Island Foodwebs

Copyright © 2016 by Jillian Amy Swift

#### Abstract

Polynesian Subsistence, Nutrient Flows, and Long-Term Socio-Ecosystem Dynamics: Insights from Stable Isotope Analysis of the Pacific Rat (*Rattus exulans*) in Island Foodwebs

by

#### Jillian Amy Swift

#### Doctor of Philosophy in Anthropology

#### University of California, Berkeley

#### Chancellor's Professor Emeritus Patrick V. Kirch, Chair

This dissertation applies stable carbon ( $\delta^{13}$ C) and nitrogen ( $\delta^{15}$ N) isotope analysis to archaeological specimens of the human-transported Pacific rat (*Rattus exulans*) to investigate prehistoric patterns of subsistence, site use, and long-term socioecosystem dynamics on Polynesian Islands. The Pacific rat is a small commensal species characterized by low dietary selectivity and a limited home range. Its close association with past Polynesian peoples—and ubiquity in Polynesian archaeological sites—suggests that dietary change in this species can provide insight into changing island landscapes and human subsistence regimes. Pacific rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values were compared across three contrastive island socio-ecosystems: Mangareva, the Marquesas, and Tikopia.

Spatiotemporal trends in Pacific rat stable isotope ratios are related to localized changes in human activity and subsistence practices, as well as global ecosystem processes which include avifaunal extinctions, resource depression, and soil nutrient cycling. On Mangareva, temporal variations in  $\delta^{13}$ C and  $\delta^{15}$ N values were assessed from three sites: the Onemea site, Taravai Island (TAR-6), Nenega-iti Rockshelter, Agakauitai Island (AGA-3), and Kitchen Cave Rockshelter, Kamaka Island (KAM-1). Declining  $\delta^{15}$ N values through time at all three Mangarevan sites reflect archipelago-wide socio-ecosystem changes related to site activity and avifaunal population declines. At the Hane dune site, Ua Huka Island (Marquesas), shifts in rat diet provide insight into changing Marquesan settlement and subsistence practices. On the Polynesian Outlier of Tikopia, a 650-year period of stability in rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values during the Kiki Phase (800-160 BC) suggests that mechanisms for long-term socio-ecosystem sustainability have been in place on the island for over 2000 years. These results demonstrate the capacity for stable isotope analysis of the Pacific rat to provide a new, lowimpact line of evidence towards reconstructing localized patterns of site use, subsistence practices, and island ecology.

Table of	Contents
----------	----------

List of Figures and Tables	<i>ii</i>
Acknowledgements	iii
Chapter 1: Introduction	1
The Human Colonization of the Pacific	1
Transported Landscapes: Polynesian Subsistence and Ecosystem Change	2
Islands as "Model Systems"	4
Pacific Rats as Proxies for Nutrient Flows in Island Socio-Ecosystems	5
Preparation and Analysis of Bone Collagen for C and N Stable Isotope Ratios	6
Carbon an Nitrogen Stable Isotope Analysis in Oceania	9
Dissertation Outline	12
Chapter 2: Stable Isotope Analysis of Pacific Rats from Mangareva	13
Introduction	13
The Pacific Rat in Polynesian Ecosystems	15
Materials and Methods	16
Results	21
Seabird Population Decline and Nutrient Flows on Mangareva	25
Conclusion	27
Chapter 3: Coastal Subsistence and Settlement at the Hane Dune Site, Marguesas	28
Introduction	28
Subsistence and Settlement in the Marquesas Islands	29
Materials and Methods	32
Results	33
Discussion	36
Conclusions	38
Chapter 4: Pacific rat diet as proxy for human-ecosystem dynamics on Tikopia	40
Introduction	40
Long-Term Human Ecodynamics on Tikopia	41
Materials and Methods	44
Results	46
Socio-Ecosystem Dynamics and Pacific Rat Stable Isotope Ratios	49
Conclusions	52
Chapter 5: Conclusions	. 54
Mangareva, the Marquesas, and Tikopia in Comparative Perspective	54
Mangareva, the Marquesas, and Tikopia in Comparative Perspective   Future Directions for Stable Isotope Studies of the Pacific Rat	54 56

## Figures

Figure 2.1. Map of the Mangarevan archipelago and labeled study sites	14
Figure 2.2. Pacific rat $\delta^{13}$ C and $\delta^{15}$ N values from all three Mangarevan sites	23
Figure 2.3. Rat $\delta^{13}$ C and $\delta^{15}$ N values from Mangarevan sites with potential food sources	24
Figure 2.4. Avifaunal remains from all three Mangarevan sites	26
Figure 3.1. Map of the Marquesas Islands with Hane dune site labeled	30
Figure 3.2. Pacific rat $\delta^{13}$ C and $\delta^{15}$ N values from the Hane dune site	36
Figure 3.3. Rat $\delta^{13}$ C and $\delta^{15}$ N values from Hane with potential food sources	37
Figure 4.1. Map of Tikopia with study sites labeled	41
Figure 4.2. Pacific rat $\delta^{13}$ C and $\delta^{15}$ N values from Tikopia by cultural phase	49
Figure 4.3 Rat $\delta^{13}$ C and $\delta^{15}$ N values from Tikopia with potential food sources	51
Figure 5.1. Mean Pacific rat $\delta^{13}$ C and $\delta^{15}$ N values from all study sites	55

## Tables

Table 2.1. Results of bone collagen stable isotope analysis from all three Mangarevan sites	19
Table 2.2. Mean Pacific rat $\delta^{13}$ C and $\delta^{15}$ N values by layer from each Mangarevan site	22
Table 3.1. Results of bone collagen stable isotope analysis from the Hane dune site	34
Table 3.2. Mean rat bone collagen $\delta^{13}$ C and $\delta^{15}$ N by cultural phase from Hane	35
Table 4.1. Revised chronology for Tikopia's prehistoric cultural phases	42
Table 4.2. Results of bone collagen stable isotope analysis from Tikopia	47
Table 4.3. Descriptive statistics for Pacific rat $\delta^{13}$ C and $\delta^{15}$ N values by cultural phase	48

#### Acknowledgements

I am immensely grateful to a number of people for their support in making this dissertation possible. This work is the culmination of years of collaborations, friendships, and inspirations, and I apologize if I have overlooked any of those here.

First to my dissertation committee, I thank you all for sharing your knowledge, your support, and your time with me over the years. Foremost to Pat Kirch, who is an endless font of knowledge about the cultures and prehistory of the Pacific Islands, and to whom I will be forever indebted for his mentorship and encouragement. To Christine Hastorf, who has a remarkable knack for challenging me to find new ways to think about my research, and whose intellectual rigor has vastly improved this work and made me a better scientist. And to Tony Barnosky, who has taught me so much about communicating across disciplinary boundaries, and to always think about how we can harness our knowledge of the past to build a better future.

I would also like to thank my collaborators in this research. To Melanie Miller, a friend and role model, who encouraged me to pursue stable isotope analysis and is responsible for my training (though any errors in this work are entirely my own). To Eric Conte and Guillaume Molle for granting me permission to conduct destructive analysis on Pacific rat remains recovered from the Hane dune site, and for securing funding for these endeavors. To Pat Kirch once again for granting me permission to conduct destructive analysis on Pacific rat remains recovered from Tikopia and Mangareva, and for inviting me to be a part of the excavation crew on Mangareva in 2012 and 2014.

This research was funded by a National Science Foundation Doctoral Dissertation Research Improvement Grant (BCS-1452364) while on tenure as a National Science Foundation Graduate Research Fellow (DGE-1106400). Additional support came from the National Science Foundation (Grant No. CNH-1313830), the *Grand Observatoire de l'environnement et de la biodiversité terrestre et marine du Pacific Sud* (GOPS), and the UC Berkeley Stahl Fund and Lowie-Olson Fund.

I wish to thank the people of Mangareva, especially our gracious hosts Edouard and Denise Sanford on Taravai Island, and Tihoni Reasin on Kamaka. I am also grateful to Tepano Paeamara, may he rest in peace, and his wife Helene Paeamara. I will always remember their open hearts, kindness, and generosity towards us. Thank you also to the other members of the 2012 and 2014 Mangareva excavation teams, Guillaume Molle, Jennifer Kahn, Alex Baer, and Rose Guthrie.

I am grateful to the Bernice P. Bishop Museum, and especially to Mara Mulrooney and Charmaine Wong, for permitting me to access and conduct destructive analysis on Pat Kirch's excavated faunal remains from Tikopia. Thank you to Todd Dawson, Stefania Mambelli, and Paul Brooks of Berkeley's Center for Stable Isotope Biogeochemistry for their assistance in stable isotope preparation and analysis, and to Stefania Mambelli for providing comments on this research. I am also grateful to Gina Palefsky for assisting me with bone collagen extraction through the Berkeley Undergraduate Research Apprenticeship Program (URAP).

I feel incredibly fortunate to have been at UC Berkeley these past seven years, and to be surrounded by amazing scholars and people. I am especially thankful to Jun Sunseri, who took me under his wing (zooarchaeology pun intended), and I aspire to follow in his footsteps and ensure every day spent as an educator and researcher is meaningful to myself, my students, and the communities I serve. Thank you also to Diane Gifford-Gonzalez for my training in zooarchaeology. The commute to Santa Cruz was well worth it! I am grateful to Kent Lightfoot, a man of tremendous generosity and spirit, who somehow finds the time to check up on and encourage us all. Laurie Wilkie, Sabrina Agarwal, Rosemary Joyce, Junko Habu, Lisa Maher, Meg Conkey, and Ruth Tringham all made important contributions to my education and development as a scholar. I am also grateful to the unsung heroes of the Anthropology Department, Ned Garrett and Kathleen Van Sickle, and Tomeko Wyrick and Nico Tripcevich of the Berkeley Archaeological Research Facility.

I was lucky enough to enter Berkeley with a cohort of some of the most amazing people I've ever met, who have become colleagues and close friends. Annelise Morris, Melanie Miller, Bryan Cockrell, Peter Nelson, and Flavio Silva, I have learned so much from all of you over the years, and I cherish the wonderful times we have had together (with many more to come). Thanks are also due to my lab mates in the Oceanic Archaeology Laboratory, James Flexner, Alex Baer, Tom Sapienza, and Kirsten Vacca. I truly feel that the day I joined the OAL I gained three brothers. Their friendships have meant so much to me, and I look up to all of them. I am also thankful for the friendship, conversation, and mentorship of the extended and honorary OAL *'ohana*, especially Rose Guthrie, Guillaume Molle, Deia Nickelson, Jenny Kahn, Mark McCoy, Sidsel Millerstrom, and Peter Mills.

Thank you to the Berkeley graduate students and the Beer Queens of yore for creating the fantastic community that convinced me I needed to be here in the first place. I hesitate to call anyone out by name because they have all influenced my life and scholarship in one way or another, but I feel I owe particular thanks to Teresa Dujnic Bulger, Anna Browne Ribeiro, Katie Chiou, Elliot Blair, John Chenoweth, Rob Cuthrell, Sara Gonzalez, Amanda Guzman, Anna Harkey, Di Hu, Ashley Lipps, Shanti Morell-Hart, Gabriel Sanchez, Trent Trombley, and Julie Wesp. Whether they know it or not, they have all been sources of inspiration and friendship.

I am deeply indebted to two particular individuals who served as my mentors prior to coming to Berkeley. Rita Wright first inspired my passion for archaeology, taught me how to conduct archaeological research, and encouraged me to apply to graduate school. Her mentorship has meant the world to me. And to Rockford Weitz, without whom I probably would never have even gone to college. I am so grateful for his guidance and willingness to teach a puzzled high school kid how to "play the game" of academia. Thank you for getting me here, Rocky.

I owe a final word of thanks to my friends and family for their continued love, support, and patience. To my mother and father, Jim and Marci Swift, thank you for the sacrifices you have made so that I could be where I am today. Your love is the deep-set foundation that allows me to keep growing and I am so lucky to be your daughter. To my brothers, Afton and Andy, for always being there for me to look up to, and encouraging me to push myself further. Finally, to my husband Emmanuel. Thank you for always believing in me. I could not have done this without you.

#### Chapter 1

#### Introduction

This dissertation applies bone collagen stable isotope analysis to archaeologically recovered remains of the Pacific rat (*Rattus exulans*) in order to investigate nutrient flows in three contrastive Polynesian island socio-ecosystems. The close commensal association of Pacific rats with past Polynesian people and archaeological sites, combined with their limited home range and omnivorous diet, render assemblages of prehistoric rat bones useful proxies for understanding paleoenvironmental and human paleodietary change. Archaeological bone collagen of Pacific rats was analyzed for carbon ( $\delta^{13}$ C) and nitrogen ( $\delta^{15}$ N) stable isotope ratios from the islands of Mangareva (Gambier Archipelago, French Polynesia), the Hane Dune Site on Ua Huka Island (Marquesas Islands, French Polynesia), and the Polynesian Outlier of Tikopia (Solomon Islands). This chapter provides an introduction to the region, cultures, theoretical perspectives, and methodologies engaged with in this research.

#### The Human Colonization of the Pacific

The Pacific Ocean covers over 30% of the Earth's surface, and its islands scattered throughout encompass at least 40,000 years of prehistory and myriad peoples and cultures. The first people to occupy this region inhabited the islands of Near Oceania, which include Papua New Guinea and parts of Island Melanesia, including the Bismarck and Solomon archipelagoes (see Green 1991). The first inhabitants of Near Oceania arrived by sea, crossing the island zone of Wallacea into the continent of Sahul, which at the time incorporated Australia, Tasmania, and New Guinea. Other islands within Near Oceania—most of these visible from nearby islands—were settled by around 30,000 years ago (Kirch 2000a; Wickler and Spriggs 1988).

The islands of Remote Oceania, which encompass the rest of island Melanesia, Micronesia, and Polynesia, are separated by much greater distances. Voyaging to these islands required complex seafaring technology, and as a result, these islands were settled considerably later. The first people to arrive in Remote Oceania (around 3000 BP; Petchey et al. 2014) were associated with the Lapita Cultural Complex: an expansion of Austronesian-speaking peoples out of Island Southeast Asia. The Lapita culture is perhaps most famously distinguished by its dentate-stamped ceramics. In addition, the Lapita people introduced a suite of cultural practices, agricultural techniques, artifacts, and plant and animal species to Oceania (see Kirch 1997a). Evidence of Lapita arrival has been found in Near Oceania and as far east as Samoa and Tonga.

The archipelagoes of Samoa and Tonga were first settled around 2850 BP (Burley et al. 2012), and along with Futuna and 'Uvea encompass the Polynesian Homeland region. After Lapita settlement in this area, voyaging to new islands appears to have paused for nearly 2000 years, while a uniquely Polynesian culture developed. Features of this Polynesian culture include the disappearance of Lapita pottery in favor of undecorated ceramics known as Polynesian Plainware, followed by the disappearance of ceramics altogether. In addition, a distinctly Polynesian suite of basalt adzes, shell and bone ornamentation, and fishing technologies developed. These artifacts would form the basis of the non-perishable (and therefore archaeologically recoverable) distinctly Ancestral Polynesian material culture.

After the formation of this uniquely Polynesian culture in the Western Polynesian homeland region, long-distance voyaging resumed and the rest of the Polynesian triangle was

colonized (see Matisoo-Smith 2015 for a review of the most recent evidence for the chronology of settlement in Oceania). The Polynesian region, formed by connecting Hawaii, New Zealand, and Rapa Nui as vertices to form a triangle, is of particular interest to comparative studies as it is geographically distinct and can also be considered a culturally monophyletic unit: the islands of Polynesia were colonized rapidly by groups of people originating from the same Western Polynesian homeland with similar cultural beliefs, knowledge, practices, and languages. These island societies were relatively isolated from one another following colonization, facilitating the subsequent development of numerous unique Polynesian cultures from the same ancestral origin. Outside of the Polynesian triangle are roughly 18 islands known as Polynesian Outliers. The people of these islands speak Polynesian languages, and many trace their origins to islands in the Western Polynesian homeland region (Firth 1961; Kirch 1984). Uniquely, the Polynesian people were not the first to arrive to these islands, as they had previously been settled during the Austronesian expansion into Remote Oceania. As a result, the Polynesian Outliers carry a blend of characteristically Polynesian cultural practices with the plants, animals, and cultures of islands elsewhere in the Pacific.

#### Transported Landscapes: Polynesian Subsistence and Ecosystem Change

Polynesian people transported a set of familiar plant and animal species to new islands, ensuring both their survival and the continuation of Polynesian cultural practices in new island landscapes. This set of "portmanteau biota" (Crosby 1986) or canoe plants and animals (Whistler 2009) included agricultural staples such as taro (*Colocasia esculenta*), yam (*Dioscorea* spp.), and breadfruit (*Artocarpus altilis*), as well as plants with other uses such as the *ti* plant (*Cordyline fruticosa*) and the paper mulberry (*Broussonetia papyrifera*). Transported animals included the domestic pig (*Sus scrofa*), dog (*Canis lupus familiaris*), and chicken (*Gallus gallus*), and the commensal Pacific rat, as well as inadvertent 'stowaways' including geckos, land snails, weeds, and arthropods. It is unclear whether the Pacific rat should be included in this list of inadvertent stowaways or if it was introduced intentionally as a potential food source (see Matisoo-Smith et al. 1999, Matisoo-Smith and Robins 2004).

Ethnographic and historic accounts point to some consistencies in foodways and subsistence practices across Polynesian islands (see Kirch and Green 2001). Polynesian meals were centered on a starch base (such as taro, sweet potato, breadfruit, or yam), which was prepared in numerous ways including baking in an underground earth oven, fermenting, and combined with emollients to form a pudding. While staple crops could be served on their own, a "proper meal" is formed with the addition of a flesh food "relish," which encompasses both marine (e.g., fish, shellfish, turtle, marine mammals) and terrestrial (e.g., pig, dog, chicken and other avifauna) resources (Kirch and O'Day 2003).

Many Polynesian subsistence systems appear to have followed the developmental trajectory laid out by Yen (1973), which outlines an early reliance on marine resources and indigenous fauna, followed by a gradual development and intensification of agricultural regimes based around transported Polynesian crops and specialized to suit particular island environments. This developmental scenario is supported by numerous archaeofaunal assemblages from early colonization-phase sites with high proportions of avifauna, marine mammals, and an abundance of benthic and pelagic ichthyofaunal species (e.g., Anderson et al. 1994; Kirch et al. 2015; Steadman et al. 1990), as well as extensive research into the development of agricultural systems

on numerous Polynesian islands (e.g., Barber 2000; Kirch 1994; Kurashima and Kirch 2011; Ladefoged et al. 2003, 2011; Quintus et al. 2016).

Humans, their translocated species, and the development of agricultural regimes had immense and long-lasting impacts on Polynesian island ecosystems. Kirch (1983) recognized the introduction of exotic flora and fauna, fire clearance of endemic forest for agriculture, and the modification of island erosional and depositional processes as major anthropogenic influences to Polynesian landscapes. These processes and their impacts have since been demonstrated in the Cook Islands (e.g., Allen 1998; Kirch 1996; Kirch et al. 1995), Hawaii (e.g., Athens 1997; Dixon et al. 1997), Mangareva (e.g., Kirch et al. 2015), Rapa Nui (e.g., Bahn and Flenley 1992; Hunt and Lipo 2010), the Marquesas (e.g., Aswani and Allen 2009; Rolett 1992, 1998), and the Society Islands (e.g., Kahn et al. 2014, 2015; Lepofsky et al. 1996), among other cases.

Extinctions and extirpations of indigenous flora and fauna, especially avifaunal species, commonly followed human arrival to new islands. Predation by humans and other introduced species, loss of habitat from agricultural forest clearance, and competition with introduced species contributed to the loss of indigenous plants and animals. Steadman (1995) estimates an average of 10 avifaunal extinctions or extirpations per island in the Pacific. Prior to human contact, land birds on Polynesian islands had evolved in an environment without terrestrial mammalian predators, and were therefore naïve to the threat of human predation. Steadman and Kirch (1990) suggest that early human acquisition of avifauna more closely resembled a gathering, rather than hunting, activity due to a lack of predator response mechanisms. Seabird extinctions had deleterious effects on island ecosystems, restricting the transfer of marine nutrients to terrestrial soils and inhibiting seed dispersals (James 1995). An extensive body of zooarchaeological literature documents intensive human consumption and subsequent elimination of avifaunal populations in Polynesia (e.g., Holdaway 1989; Steadman 1993, 1995, 1997, 2006; Steadman and Kirch 1990; Steadman and Justice 1998; Steadman and Rolett 1996).

Transported commensal species (pig, dog, chicken, and rat) played a role in these landscape transformations as a new source of competition and predation for island flora and fauna. Although Polynesians may have attempted to transport all four of these animals to new islands, the introductions of pig, dog, and chicken were not always successful. In some cases, these species were successfully introduced but later extirpated, putatively due to trophic competition over limited resources (Kirch 2000b). Giovas (2006) determined a correlation between island size (used as a proxy for island productivity) and the persistence of pig populations into the European contact era. Bay-Petersen (1983) suggested that the elimination of pig and dog populations on small islands is not only related to terrestrial productivity, but also to the increased access to marine resources per capita on small islands, obviating the need for terrestrial protein sources.

The Pacific rat in particular has been singled out as a source of significant ecological devastation both in present and past island ecosystems (e.g., Athens et al. 2002; Harper and Bunbury 2015; Holdaway 1989; Hunt and Lipo 2007, 2010; Meyer and Butaud 2009; Mieth and Bork 2010; Porch 2008; Ruffino et al. 2015; Shiels and Drake 2015; Towns et al. 2006). Unlike the pig, dog, and chicken, the rat is widespread across the Pacific and nearly ubiquitous in Polynesia. The Pacific rat arrived on Polynesian islands devoid of natural predators (apart from humans) and without competition from other small mammals such as mice or other rat species. Left unchecked, rat populations could multiply quickly, producing several litters per year of up to 7 young according to some estimates (Kramer 1971; Moller and Craig 1987; Tamarin and Malecha 1972). Rats could have major impacts on indigenous flora and fauna: by consuming

small birds and bird eggs and native plants and plant seeds, they exacerbated the impacts of human forest clearance and hunting on these species, while also slowing their propagation. The European-introduced ship rat (*Rattus rattus*), Norwegian rat (*Rattus norvegicus*), and house mouse (*Mus musculus*) have in many cases overtaken the Pacific rat as more significant contemporary ecological menaces. However prior to these introductions, the Pacific rat appears to have been partly responsible for prehistoric species extinctions and ecosystem change.

#### Islands as "Model Systems"

This dissertation compares data from three separate Oceanic islands to engage with the concept of islands as model systems for understanding global socio-ecosystem processes. The term "socio-ecosystem" is employed by Barton et al. (2004) to address the inextricable connection of environments and landscapes to human society and cultural practices throughout history. The socio-ecosystem concept recognizes that neither are people "passive consumers or rapacious exploiters of ecosystems," nor ecosystems simply "a backdrop for human agency or a larder to fuel human economies" (Barton et al. 2004:254). In this dissertation I interpret rat stable isotope values within the context of developing human subsistence regimes, activities, and social structures. Variations in rat diet through time are hypothesized to reflect changes within linked social and natural landscapes.

The use of islands as model systems has encouraged cross-disciplinary collaboration between archaeologists and natural scientists to explore the relationship between humans and their environment. Burney (1997) proposed that tropical islands in the Pacific and elsewhere could serve as "paleoecological laboratories" for assessing the impacts of human arrival to new ecosystems. Vitousek (2002) asserted that Pacific Islands are useful as real model systems for understanding evolution and speciation, conservation biology, and cultural processes. Kirch (2007a) identified a number of factors which make Polynesian islands particularly appealing as contrastive model systems for understanding socio-ecosystem change: (1) a relatively late colonization of previously uninhabited islands by populations with agriculturally-based economies; (2) short, well-controlled chronologies (3000-1000 year duration); (3) a high degree of isolation after colonization; (4) clear, traceable transitions from low-density to high-density human populations; (5) transformations in sociopolitical complexity; and perhaps most importantly, (6) a common origin in an Ancestral Polynesian culture (see Kirch and Green 2001). Kirch has compared numerous island systems including Tikopia, Mangaia, Mangareva, and the Hawaiian Archipelago to isolate the role of variables such as island size, geologic age, isolation, and maximum elevation in long-term island socio-ecosystem development and sustainability (see Kirch 1997b, 2004, 2007a, 2007b).

Many researchers have attempted to determine the cause of Rapa Nui's perceived "collapse," as this has been upheld as a microcosm for global human-environment interaction and resource decline (e.g., Bahn and Flenley 1992; Diamond 2005; Hunt 2007; Hunt and Lipo 2010; Mieth and Bork 2010). Rolett and Diamond (2004) compared nine environmental variables (rainfall, elevation, area, volcanic ash fallout, Asian dust transport, uplifted coral *makatea* terrain, increases in latitude, age, and isolation) across Pacific Islands and found that rainfall and higher latitude (itself a measure of temperature declines) were key factors in evaluating island susceptibility to deforestation. They determined that Easter Island was particularly fragile, as it was at the extreme end of all variables considered.

Vitousek et al. (2004) used the Hawaiian archipelago as a model for agricultural intensification, soil nutrient cycling, and increasing social stratification. They suggested that rainfall, geologic age, and overall soil fertility were key factors in the development of extensive dryland agricultural systems and sociopolitical complexity in the leeward regions of the Hawaiian Islands. They argued that the general relationship between these variables were critical not only in Hawaiian prehistory, but could also be applied towards understanding agricultural intensification in other tropical environments. These ideas were developed more extensively in a volume edited by Kirch (2011a), which incorporated additional variables such as demography, spatial distribution, and a chronology of sociopolitical development. Baer et al. (2015) have begun to extend this research into the dryland agricultural system of Kaupō, Maui.

More recently, the ambitious Mo'orea IDEA project (Davies et al. 2016) has brought together an international collaborative network of researchers from the physical, biological, computer, informational, and social sciences who endeavor to model the entire socio-ecological system of Mo'orea. Their larger aim is to understand sustainable practices both within islands and coastal regions, and as a model for continental regions and global processes. The Mo'orea IDEA project is one of many recent initiatives to engage with the notion that humankind has become one of the most significant impacts on geological and ecological systems. As the concept of the Anthropocene (see Crutzen 2002; Steffen et al. 2011; Zalasiewicz et al. 2012) becomes more prevalent in contemporary scholarship, the modeling of human-environment interactions on Polynesian Islands can provide critical insights into global issues of resource management and sustainability (Vitousek and Chadwick 2013).

#### Pacific Rats as Proxies for Nutrient Flows in Island Socio-Ecosystems

Stable isotope analysis of commensal fauna has been applied to a range of anthropological topics, including paleoenvironmental reconstruction, subsistence, mobility and animal husbandry practices (e.g., Allen and Craig 2009; Craig 2009; West 2007 in the Pacific, and see reviews by Birch 2013; Zangrando et al. 2014). Previous research has applied carbon and nitrogen stable isotope ratios of archaeologically recovered dog remains as a proxy for human diet (see Guiry 2012). Although dogs were introduced to a number of Polynesian islands in prehistory (Hawai'i, New Zealand, Society Islands, Tuamotu Islands, the Marquesas, and Mangareva), they were frequently managed food sources with controlled diets. Dogs were often fed a vegetarian diet, and penned to restrict access to other food sources (Luomala 1960; Titcomb 1969). Thus, it is unlikely that reconstructed dog diet in Polynesia would provide a proper analog to human diet. A notable exception to this may be New Zealand, where dogs (kuri) were occasionally used for hunting birds and rats, could survive to old age, and had a more carnivorous diet which included human meal scraps (Clark 1997a, 1997b; Coutts and Jurisich 1973). Likewise, past Polynesian husbandry strategies of the domestic pig inhibit direct comparisons to human diet (West 2007). The commensal Pacific rat, also a mammalian omnivore, lived in similarly close association with people but does not appear to have been managed by Polynesian people and thus may provide a better proxy for human diet.

Guiry and Gaulton (2015) established critical groundwork for the application of stable isotope analysis of rats as proxies for past human diet and activity. Contemporary ecological studies support the idea that rat diets are highly generalized, and that rat dietary stable isotope ratios will reflect an averaging out of locally available resources (e.g., Caut et al. 2008; Hobson et al. 1999; Ruffino et al. 2011). However, Guiry and Gaulton caution that archaeological sites

represent complex social and natural ecosystems, which carry additional complicating factors. They suggest greater intra-annual resource variation, different predator-prey relationships (humans in particular), and interspecific competition may cause rats to adopt specialized feeding behaviors. Although past Polynesian people may not have intentionally managed rat diet, it is possible that the threat of their predation altered rat feeding behavior in habitation sites and other areas of high human density. Despite these warnings however, Guiry and Gaulton demonstrated change in rat diet through time which they were able to correlate with changing socioeconomic conditions at Ferryland, Newfoundland.

Pacific rat remains are recovered frequently from Polynesian archaeological sites, and it appears the species was successful in establishing the human household as an ecological niche. On Kapiti Island, New Zealand, Bramley (2014) observed home ranges of between 26-89 m in diameter for the Pacific rat, compared to home ranges of 218-916 m in diameter for the larger Norway rat. The Pacific rat rarely burrows, instead preferring open areas with good ground cover or disturbed areas such as villages and households (Dwyer 1978; Taylor 1975). Rats may have spent considerable time in the roof thatching of Polynesian households, opportunistically scavenging food scraps while human dwellers slept (see Oliver 2002:83). A recent study by Morand et al. (2015) in contemporary mainland Southeast Asia found that the Pacific rat was particularly specialized to household sites compared to other sympatric rat species. Uniquely, *R. exulans* was found almost exclusively in households within villages, rather than in natural habitats.

Much of our understanding of Pacific rat diet comes from contemporary ecological studies. This rat species has been observed consuming insects, snails, land crabs, lizards, turtle hatchlings, bird eggs, and even other rats (Spennemann 1997). However, many dietary studies have found Pacific rat diet to be primarily vegetal, with most protein derived from arthropods rather than terrestrial or marine resources of a higher trophic level (e.g., Bunn and Craig 1989; Shiels et al. 2013). Pacific rat diet has also been shown to vary based on seasonal resource availability. Bone collagen stable isotope data represent an averaging of diet over longer timespans and cannot detect seasonal variation in diet, however this seasonal sensitivity further suggests that rats dwelling in households will reflect the resources available within or nearby these structures.

Modern studies do not necessarily provide a robust analog to prehistoric Pacific rat diet. European arrival on Polynesian Islands introduced a suite of new flora and fauna as well as new cultural practices which led to transformations in island landscapes as well as human subsistence practices and activities. Introduced species included previously unknown predators such as the domestic cat (*Felis catus*) and competitors such as the black rat, brown rat, and house mouse. These introductions impacted Pacific rat populations and may have led to significant deviance from pre-European diet and behavior. Further, these contemporary studies have focused on rats living away from human habitation sites. The diet of rats living within contemporary Polynesian village and household contexts has not been studied. Bone collagen stable isotope analysis of Pacific rat remains from other Polynesian archaeological sites indicate at least occasional dietary inputs of high trophic level proteins (Commendador et al. 2013; Richards et al. 2009).

#### Preparation and Analysis of Bone Collagen for Carbon and Nitrogen Stable Isotope Ratios

The application of carbon and nitrogen stable isotope analysis to archaeological bone collagen has proven an effective tool in paleodietary studies (see Katzenberg 2008; Schwarcz

and Schoeninger 1991; Sealy 2001). An isotope is a molecule of a particular element which contains the same number of protons and electrons as other molecules of that element but varies in its number of neutrons. Isotopes can be either stable, meaning they do not undergo radioactive decay (e.g., <sup>12</sup>C and <sup>13</sup>C) or radioactive (e.g., <sup>14</sup>C). The addition of a neutron changes very little about a molecule, however the slight increase in atomic mass allows for small but predictable variation in the way isotopes are distributed before and after a physical or chemical reaction. In general, heavier isotopes tend to react more slowly, and form stronger chemical bonds. The difference between the stable isotope ratios of the substrates and products of a reaction is known as *fractionation*, and this process forms the basis of all stable isotopic analyses (see reviews by Hoefs 1987; Sharp 2007).

Carbon and nitrogen stable isotopes fractionate in predictable ways throughout ecosystems and foodwebs. A significant source of carbon fractionation originates in the different photosynthetic pathways used by plants (Bender 1968; 1971). C<sub>3</sub> plants tend to have  $\delta^{13}$ C values in the range of -34 to -21‰, while C<sub>4</sub> plant  $\delta^{13}$ C values fall between -15 to -8‰. Crassulacean Acid Metabolism (CAM) plants often have  $\delta^{13}$ C values that fall between the ranges of C<sub>3</sub> or C<sub>4</sub> plants. Most Polynesian economic crops are C<sub>3</sub> plants, however sugarcane (a Polynesian crop plant) and most tropical grasses are C<sub>4</sub>. While terrestrial plants derive most carbon from atmospheric CO<sub>2</sub>, carbon in marine systems is derived from a mix of sources including terrestrial detritus and dissolved bicarbonate. Thus, marine primary producers tend to have significantly less negative  $\delta^{13}$ C values compared to terrestrial plants, though these values vary depending on species, tissue, and location (Lee-Thorp 2008; Schoeninger and DeNiro 1984).

Nitrogen stable isotope ratios are useful in determining the trophic level of a species, as an enrichment of 3-5‰ occurs between producers and consumers for every "step" up the trophic ladder. An average figure of 3-3.5‰ is frequently employed in dietary studies (Bocherens and Drucker 2003; DeNiro and Epstein 1981; Minagawa and Wada 1984). Because marine foodwebs have long chains of producers and consumers, N stable isotope ratios of marine systems tend to be more enriched (greater abundances of <sup>15</sup>N) than their terrestrial counterparts. Nitrogen fixing plants, which include legumes, sugar cane, and blue-green algae draw on atmospheric N<sub>2</sub> and tend to have lower  $\delta^{15}$ N values compared to non-fixers that rely on soil N which is more enriched (Schoeninger and DeNiro 1984). N stable isotope ratios are also influenced by numerous environmental factors, including temperature, aridity, and salinity (DeNiro and Hastorf 1985).

Stable isotope abundances are measured through an isotope ratio mass spectrometer (IRMS). Samples are converted to gas form, ionized, and accelerated into a focused ion beam. This beam of charged isotopes then enters a magnetic field which causes the particles to accelerate in an arced path. As acceleration is mass-dependent, each isotope assumes a different trajectory (heavier molecules follow an arc with a larger radius than lighter molecules) allowing them to become spatially separated by mass upon exiting the magnetic field. These mass-separated ion beams are directed into a series of Faraday cups which generate an electrical current proportional to the number of molecules to impact the cup. The strength of the current from each Faraday cup is then measured to calculate the relative proportion of each isotope within the sample.

Stable isotope abundances are expressed as ratios of the heavier isotope to the lighter isotope (e.g.,  ${}^{13}C/{}^{12}C$ ). However, because the differences in isotopic abundances are so small, these ratios are not expressed as absolute values but relative to an internationally established standard, and in parts per thousand (‰). These results are expressed in  $\delta$  notation, where

$$\delta \mathbf{X} = [(\mathbf{R}_{\text{sample}} \div \mathbf{R}_{\text{standard}}) - 1] * 1000\%.$$

For C and N stable isotope ratios, this formula becomes:

$$\delta^{13}C = \left[ \left( {^{13}C_{sample}}/{^{12}C_{sample}} \div {^{13}C_{standard}}/{^{12}C_{standard}} \right) - 1 \right] * 1000\%$$
 and

$$\delta^{15}N = [({}^{15}N_{sample}/{}^{14}N_{sample} \div {}^{15}N_{standard}/{}^{14}N_{standard}) - 1] * 1000\%$$

The standard for calculating  $\delta^{13}$ C values is PeeDee Belemnite (PDB), a cretaceous marine fossil (*Belemnitella americana*) from the PeeDee formation in South Carolina. Because this standard is enriched in <sup>13</sup>C relative to most extant plants and animals,  $\delta^{13}$ C values derived from archaeological bone collagen and potential food sources tend to be negative. The standard for calculating  $\delta^{15}$ N values is an accepted standardized value for atmospheric air (Ambient Inhalable Reservoir, or AIR).  $\delta^{15}$ N values tend to be positive numbers, as AIR is depleted in <sup>15</sup>N relative to most samples.

This study utilizes C and N stable isotope ratios from bone collagen of the Pacific rat. Bone collagen, the organic portion of bone, is comprised of amino acids, some of which cannot be synthesized *de novo* and instead come from dietary proteins. As such, bone collagen is heavily weighted towards the protein component of diet (Ambrose and Norr 1993; Tieszen and Fagre 1993). Bone collagen is remodeled over time, however for small short-lived species such as rats this tissue records the average diet consumed during most of the individual's lifespan (see Tieszen et al. 1983).

There are several methods of preparing bone collagen for stable isotope analysis, and the choice of method is often dependent on time, money, and available equipment and laboratory facilities. A recent meta-analysis by Sealy et al. (2014) compared the older, simpler method (after Ambrose 1990) with the more intensive ultrafiltration method (after Brown et al. 1988; Longin 1971) and determined there were no statistically significant differences between the results produced from either method. The first method involves demineralization of bone chunks in dilute hydrochloric acid (HCl) followed by a dilute sodium hydroxide (NaOH) treatment to remove humic contaminants. This treatment produces a collagen 'pseudomorph,' which is overwhelmingly composed of bone collagen but may also contain small amounts of other proteins such as osteocalcin. The latter method is more rigorous and includes a gelatinization and ultrafiltration step to isolate bone collagen. Pestle et al. (2014) further confirmed the comparability of results across labs using different preparation methods, equipment, and calibrations. They found that major differences in preparation methods employed by different laboratories did not have a significant effect on stable isotope results. Rather, the largest differences in results were introduced through variations in instrumentation and standardization. They suggest that resulting  $\delta^{13}$ C and  $\delta^{15}$ N values from the same sample may vary between labs by up to 0.6‰ and 0.9‰ respectively (their Minimum Meaningful Difference, or MMD).

Bone collagen preservation can be assessed through several measures, the most widely accepted of which is the ratio of preserved carbon to nitrogen in the sample (atomic C:N ratio). Glycine, which comprises over 1/3 of the amino acid sequence of bone collagen, has a 2:1 ratio of carbon to nitrogen, which contributes to bone collagen's unique C:N ratio. Most other amino acids have a much higher C:N ratio, and intact collagen is estimated to have a roughly 3:1 ratio

of carbon to nitrogen. Archaeological bone collagen samples are generally accepted as intact if they fall within the range of 2.9-3.6 (Ambrose 1990; DeNiro 1985), though some laboratories use a more restricted range of 3.1-3.5 (van Klinken 1999). Secondary quality control indicators include collagen yield (by weight %), as well as wt% C and wt% N within the sample. Modern bone is comprised of approximately 20-22 wt% collagen. Minimum collagen yield cutoffs vary depending on the study region, and range from 0.5-5% (Pestle and Colvard 2012; Schwarcz and Schoeninger 1991; van Klinken 1999). The characteristic concentrations of carbon and nitrogen in intact bone collagen are roughly 35 wt% and 11-16 wt%, respectively (van Klinken 1999). Analysis of amino acid composition may also be used to assess collagen preservation, though this can be costly, time consuming, and inefficient (see DeNiro and Weiner 1988). More subjective methods of assessing collagen are by morphological integrity (if the resulting collagen retains a semblance of its original shape) and by color. Pure collagen is pale yellow to white, whereas contaminated collagen will take on a brownish hue (Koch et al. 1994).

#### **Carbon and Nitrogen Stable Isotope Analysis in Remote Oceania**

Carbon and nitrogen stable isotope analysis has recently been applied to both human and animal remains in Remote Oceania to investigate changing human diet, subsistence systems, and ecology across time and space. A study by Leach et al. (1996) enumerated several limitations to stable isotopic dietary studies in the Pacific. These included the wide variety of potential food sources in coastal systems, the similarity in values between inshore reef sources and terrestrial C<sub>4</sub> sources, and the influence of marine-derived sediments on terrestrial plant  $\delta^{15}$ N and  $\delta^{13}$ C values. Degraded bone collagen in tropical environments can serve as a further hindrance to archaeological stable isotope studies in the Pacific (see Van Klinken 1999). However, a wealth of research over the past several decades has demonstrated the effectiveness of this analytical technique in the tropical Pacific.

Some of the earliest stable isotopic analyses in the Pacific were conducted on human bone collagen from the Mariana Islands. McGovern-Wilson and Quinn (1996) analyzed bone collagen of ten individuals recovered from burials at Afetna, Saipan and found that  $\delta^{13}$ C and  $\delta^{15}$ N values corresponded to greater marine resource inputs—particularly of offshore fish—than previously indicated by the faunal record. Ambrose et al. (1997) compared the diet of several human populations from three different islands within the Marianas (Guam, Rota, Saipan) during the Latte Period (AD 1000 to historic contact). The study analyzed both the organic (bone collagen) and mineral (hydroxyapatite) components of bone and found a significant input of C<sub>4</sub> plants to the diet of Saipan individuals relative to those from Guam and Rota. This study suggested that C<sub>4</sub> plants such as seaweed and sugarcane played a more significant dietary role in some regions of the Pacific than previously thought.

Stable isotope analysis of individuals from Fiji shows considerable spatiotemporal variation in human diet. Valentin et al.'s (2006) bone collagen analysis of nine elite individuals from late prehistoric/early historic burials at the Korotuku burial mound, Cikobia, North Fiji demonstrated a predominantly vegetal diet with some contributions from high trophic level marine resources. In contrast, analysis of a canid tooth from the same burial context indicated dog diet contained significant inputs of shellfish. Based on these stable isotope data and a high frequency of dental caries, the authors suggested that the people buried at Korotuku had a highly selective diet based around frequent consumption of high-value foods including puddings and pelagic fish (see Leach 2003). In contrast, bone collagen and apatite analysis conducted by Jones

and Quinn (2009) on a population of nine individuals from Nayau, Aiwa Levu, and Aiwa Lailai indicated that Fijian diet was composed primarily of  $C_3$  plants supplemented by reef resources, rather than pelagic fish. Early occupants also appear to have consumed  $C_4$  plants, which the authors suggest may have come in the form of sea grapes. The dietary differences between these populations may be related to status differences, with elite members of society having more frequent access to high trophic level proteins. A study by Field et al. (2009) confirmed variations in Fijian diet which extended beyond age, class, and status distinctions. Stable isotope analysis of 26 individuals from Sigatoka Valley and Waya Island (2700-300 BP) were compared to assess differences in Fijian diet related to geographic variability. Human diet in the Sigatoka Valley remained predominantly terrestrial. The diet of individuals from Waya Island became more terrestrial over time, but continued to be dominated by marine resources into late prehistory. The authors suggested that smaller islands with less arable land likely retained marine-focused diets throughout prehistory.

Two studies by Valentin et al. (2010) and Kinaston et al. (2014) assessed the diet of Lapita populations (3000-2500 BP) from the Teouma Site, Efate Island, Vanuatu. Valentin et al.'s (2010) analysis of 23 individuals found significant dietary contributions from terrestrial fauna (over 50% of total protein consumption), as well as some terrestrial plants, shellfish, and inshore fish. Terrestrial food sources may have included indigenous fauna such as bats, reptiles, and birds, or introduced domestic fauna such as pigs, rats, and chickens. Kinaston et al. (2014) expanded this study with the addition of carbon, nitrogen, and sulfur (S) stable isotope data from 49 individuals and a collection of dietary baseline stable isotope data from modern and archaeological flora and fauna. This study confirmed the importance of terrestrial animals to Lapita diet, in addition to reef sources and sea turtles. While horticultural systems were likely being established at this time, they appear to have been less important to overall Lapita subsistence compared to later populations. In addition, male individuals had a larger range of  $\delta^{15}$ N and  $\delta^{13}$ C values and overall higher  $\delta^{15}$ N values than female individuals, indicating males had a more varied diet with greater access to high trophic level proteins.

Within Polynesia, an analysis by Valentin et al. (2011) of 14 individuals from Tutuila Island, American Samoa (1000-150 BP) found diet to be predominantly terrestrial with some contributions of reef fish and shellfish, along with declining marine resource consumption through time. Stantis et al. (2015) analyzed carbon, nitrogen, sulfur, and strontium (Sr) stable isotope ratios of 41 individuals recovered from the 'Atele burial site, Tongatapu Island, to assess human diet and mobility in Tonga (500-150 BP). Although the diet of the entire population was focused primarily on terrestrial plant and marine protein sources, they discovered significant dietary differences related to sex and social status. Similar to Kinaston et al.'s (2014) results from Vanuatu, females had lower  $\delta^{15}$ N values than males. Individuals recovered from a burial mound associated with commoners had less negative  $\delta^{13}$ C values compared to individuals recovered from a burial mound interpreted as more elite. The increased proportion of terrestrial sources in elite diet likely came in the form of more prestigious manifestations of terrestrial crops (such as puddings), though this type of culinary elaboration cannot be detected through stable isotope analysis (Stantis et al. 2015).

Kinaston et al. (2013) similarly used C, N, and Sr isotope analysis of 38 individuals from the Wairau Bar site on the South Island of New Zealand to assess human diet and mobility. The Wairau Bar site is one of the earliest known sites in New Zealand (late 13<sup>th</sup> century AD). Significant differences in stable isotope ratios between three distinct groups of human burials (Group 1 versus Groups 2 and 3) led Kinaston et al. (2013) to suggest the individuals in Group 1

may come from a colonizing population. Brown and Thomas (2015) later reevaluated these data and argued that burial groups could not be distinguished based on stable isotope ratios, dietary differences were best explained by sex differences, and Sr stable isotopic variation could not distinguish Group 1 as a founding population. Kinaston et al. (2015) responded to Brown and Thomas (2015), arguing that they misunderstood the purpose of their study and their claims were invalid. Kinaston et al. (2015) noted that their study was not attempting to establish burial groups using isotope data, but rather to assess patterns within *pre-existing* burial groups identified by archaeological analysis. In addition to problems with Brown and Thomas' (2015) interpretations of the isotopic data, Kinaston et al. (2015) argued that their data-driven analysis, detached from supporting archaeological and ethnographic lines of evidence, was poorly supported and inconsistent with previous interpretations of New Zealand subsistence and prehistory.

Bone collagen analysis of 99 individuals from the Namu burial ground (750-300 BP), Taumako Island (Southeastern Solomon Islands) provides stable isotopic dietary information from a Polynesian Outlier (Kinaston et al. 2013). Overall, diet on Taumako during this time consisted of terrestrial C<sub>3</sub> resources and a range of marine proteins including reef fish and shellfish as well as high trophic level, offshore species. Diet also varied according to status and sex differences within Taumako society. Based on associated grave goods in burials, the wealthiest individuals had less negative  $\delta^{13}$ C and higher  $\delta^{15}$ N values than the rest of the population, and male individuals had higher  $\delta^{15}$ N values than females (though elite females still had higher  $\delta^{15}$ N values than non-elite males and females). Elite members of Taumako society appear to have had more frequent access to high trophic level (and highly valued) proteins, including offshore fish, pig, dog, and turtle.

Several stable isotopic studies within Polynesia have also focused on the diet of terrestrial animals. Allen and Craig (2009) conducted stable isotope analysis on bone collagen from humans, pigs, and dogs recovered from Aitutaki, Southern Cook Islands. Their findings showed a transition from a largely marine-based diet to one more focused on terrestrial resources for all three species over time, as well as declining human consumption of marine carnivores. They also observed a restriction in pig  $\delta^{13}$ C and  $\delta^{15}$ N values in later prehistory which may be an indication of resource depression and tighter management of pig populations. Pig and dog were both extirpated from Aitutaki prior to European contact, presumably due to competition between humans and domesticates over limited food resources (see Kirch 2000b). Studies by Richards et al. (2009) and West (2007) from the Hanamiai Dune site, Tahuata Island provide new insight into Marquesan diet and animal husbandry. Richards et al. (2009) analyzed bone collagen from pig, dog, rat, and marine fauna as well as four human teeth not associated with human remains. The four human bone samples analyzed did not have a strongly marine diet. Rather, dietary protein was obtained largely from terrestrial mammals such as pigs, dogs, and perhaps rats. West's (2007) intensive analysis of pig husbandry in the Marquesas included bone collagen stable isotope analysis of pig remains through time at the Hanamiai site. He argued that the temporal variation seen in pig diet suggested that pigs were allowed to return to a feral state upon initial introduction to the Marquesas, and were re-domesticated once a productive agricultural system had developed.

Commendador et al. (2013) analyzed 41 human teeth along with 132 archaeologically recovered specimens of terrestrial and marine fauna to assess changing diet on Rapa Nui (Easter Island). Their results indicate a largely terrestrially-based diet with protein inputs from rats and chickens, and minimal inputs from marine sources. They observe an overall decline in  $\delta^{15}$ N values over time and suggest that this may be related to increasing reliance on agricultural

products due to terrestrial resource depression and declining animal populations. They also note that this decline could be the result of changing soil-plant baseline <sup>15</sup>N. Numerous biogeochemical factors can influence nitrogen stable isotope ratios in plants and soils, and the authors suggest potential changes in rainfall patterns, soil N availability, and decreasing inputs of <sup>15</sup>N-enriched seabird guano may have lowered baseline  $\delta^{15}N$ .

#### **Dissertation Outline**

The following chapters present analyses of carbon and nitrogen stable isotope ratios from Pacific rat bone collagen recovered from three distinct island contexts. Chapter 2 presents stable isotope data from three separate islands on the almost-atoll of Mangareva (Gambier Islands, French Polynesia). Temporal declines in  $\delta^{15}$ N values at all three of these sites indicate large-scale changes in island nutrient flows, likely related to the massive avifaunal extinctions that occurred shortly after human arrival to the archipelago. Chapter 3 examines rat stable isotope ratios through time at the Hane Dune Site on Ua Huka Island (Marquesas Islands, French Polynesia). Results from this site support the close relationship between human and rat diet in Polynesian prehistory, and indicate a later shift towards intensified agriculture and animal husbandry. Chapter 4 examines changing rat diet on the Polynesian Outlier of Tikopia, and finds evidence for a long history of island socio-ecosystem sustainability. Finally, Chapter 5 concludes by placing these three case studies in comparative perspective and suggests future directions for the application of small mammal stable isotope ecology towards understanding human subsistence and environmental change.

#### Chapter 2

# Stable isotope analysis of Pacific rat (*Rattus exulans*) from archaeological sites in Mangareva (French Polynesia): Implications for nutrient flows in island socio-ecosystems

#### **Introduction**

The analysis of diet-derived bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N from archaeologically recovered faunal remains can provide paleoecological insights through the reconstruction of ancient food webs and the trophic position of humans and other animals within those webs. Small-bodied mammals in particular have proven effective proxies for paleoclimatic reconstructions (e.g., Gehler et al. 2012; Hynek et al. 2012); the analysis of commensal archaeofaunal species can generate new evidence for investigations into long-term humanenvironment interaction (ecodynamics). Here we report bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N stable isotope data for 84 Pacific rats (*Rattus exulans*) from three archaeological sites in the Mangareva Islands (Gambier Archipelago) to investigate the implications of changing foodweb dynamics and trophic positioning for this species through time in this southeastern Polynesian archipelago.

Recent attention to bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values of commensal fauna demonstrates their utility in investigations of paleoecology, animal husbandry, and human diet (e.g., Birch 2013; Guiry 2012; Stevens et al. 2008; Zangrando et al. 2014). Stable isotope analysis of domestic pigs (Sus scrofa) and dogs (Canis familiaris) in Polynesia has shed light on the intensification of pig husbandry in the Marquesas Islands (Richards et al. 2009; West 2007) and provided indications of resource stress and competition between humans and domestic species on Aitutaki, Cook Islands (Allen and Craig 2009). The commensal association of rats with humans suggests that the reconstructed diet of rats recovered from archaeological contexts can provide a local understanding of nutrient inputs and resource availability. Guiry and Gaulton (2015) utilized bone collagen stable isotope analysis of past black rat (Rattus rattus) populations at Ferryland, Newfoundland to assess changes in human social and economic practices between the  $17^{\text{th}}$  and  $18^{\text{th}}$  centuries. While they caution against utilizing rat  $\delta^{13}$ C and  $\delta^{15}$ N as a direct proxy for human diet, their study demonstrated that archaeological rat remains present a new source of evidence for understanding past human foodways, activities, and ecodynamics. The low dietary selectivity, limited home range (Atkinson and Moller 1990), and ubiquity of the Pacific rat in Polynesian island prehistory position it as an ideal species for exploring localized change in island foodwebs and nutrient flows. The stable isotope data from Mangareva document changes in rat dietary  $\delta^{13}$ C and  $\delta^{15}$ N through time across diverse archaeological contexts and indicate large-scale transformations in island ecodynamics and human subsistence activities.

Three recently excavated sites in the Mangareva Islands provide the data for this analysis (Figure 2.1). All three sites offer stratified archaeological sequences with abundant Pacific rat remains throughout the duration of human occupation. The sites also vary in geographic location, timing and duration of human site use, as well as site function and history. The Onemea Site (TAR-6) is an open-air deposit containing the earliest evidence for human activity in the Mangareva Islands. Well-stratified deposits from two rockshelter sites, Nenega-iti (AGA-3) and Kitchen Cave (KAM-1) provide contexts for investigating long-term site ecology, human activity, and rat diet.



Figure 2.1: Map of the Mangarevan archipelago and surrounding barrier reef with labeled study sites.

#### The Pacific Rat in Polynesian Ecosystems

The Pacific Rat (*Rattus exulans*) is ubiquitous throughout Polynesia, typically appearing in archaeological sites concurrent with the earliest evidence for initial human occupation. Although it is virtually certain that rats required human agency to arrive on new islands, it remains unclear whether Polynesians intentionally transported the Pacific rat as a potential food source or passively transported them as "stowaways" on Polynesian voyaging canoes (Matisoo-Smith 1994; Tate 1935). In Mangareva, rats were reportedly not eaten, although they were known to be consumed in other Polynesian islands such as Mangaia (Kirch et al. 1995; Matisoo-Smith et al. 1999; Hiroa 1938:194). The human-mediated transport of rats to new islands has allowed rats to serve as useful proxies for understanding the human colonization of the Pacific through ancient DNA analysis (e.g., Matisoo-Smith 1994; Matisoo-Smith et al. 1999), and the early presence of *R. exulans* in Polynesian archaeological sites provides directly dateable, short-lived materials for estimating the first arrival of Polynesians to new islands (Anderson 2004; Kirch 2011a; Wilmshurst et al. 2008).

Previous archaeological research has documented the frequently extensive transformations to Polynesian island landscapes following human arrival, including the introduction of nonnative species, clearance of native forest, increased rates of erosion and sedimentation, soil nutrient depletion, and extinction and extirpation of indigenous flora and fauna (e.g., Allen 1998; Burney et al. 2001; Kirch 1983, 1996; Kirch and Hunt 1997). Attention has recently turned to the impacts of Polynesian-introduced plants and animals within island socio-ecosystems, and it is now recognized that rats in particular played a key role in the extirpations and extinctions of indigenous biota on oceanic islands (e.g., Athens et al. 2002; Holdaway 1989). Pacific rats multiply rapidly (Kramer 1971; Moller and Craig 1987; Tamarin and Malecha 1972), though in most continental regions populations are kept in check by other natural predators such as snakes, cats, or birds of prey. However, the absence of native terrestrial carnivores on Oceanic islands apparently allowed rat populations to multiply extremely rapidly.

Pacific rats are opportunistic omnivores that will actively prey upon a range of species including insects, snails, small lizards, turtle hatchlings, and bird eggs and small birds (Norman 1975; Spenneman 1997). *R. exulans* diet tends to be more carnivorous than that of the black rat (*R. rattus*) though still predominantly vegetal. Modern gut content analysis demonstrates protein contributions derived primarily from arthropods (Shiels et al. 2013). Relying on modern rat dietary studies, however, may be problematic as rat diet varies based on numerous factors including sex, age, local ecology, and seasonal resource availability (Bunn and Craig 1989). Dramatic differences between pre- and post-European contact Polynesian landscapes, historic introductions of new predators, and modern competition and niche partitioning between *R. exulans* and the European-introduced black rat, Norwegian brown rat (*R. norvegicus*), and house mouse (*Mus musculus*) have undoubtedly transformed the foraging behavior of *R. exulans*, thus making modern gut content studies poor analogs for assessing prehistoric Pacific rat diet.

Pollen cores from the 'Ewa Plain, O'ahu (Hawaiian Islands) demonstrate massive forest decline after the arrival of Pacific rat but prior to significant human activity in this region, leading Athens et al. (2002) to suggest that rat predation on indigenous vegetation—rather than forest clearance for agricultural activities—was the most likely cause for initial native forest decline and avifauanal habitat loss. In contrast, on Nihoa Island, where rats were never introduced, native palm forests persist (Hunt and Lipo 2010). The abundance of rat-gnawed *Jubaea* palm seeds discovered on Rapa Nui (Easter Island) has sparked debate over the extent to

which rat predation contributed to the deforestation that eventually led to the society's so-called 'collapse' (Diamond 2005; Hunt and Lipo 2007, 2010; Mieth and Bork 2010).

Rat impact on indigenous avifauna through direct predation and habitat destruction is evidenced in numerous studies of modern seabird populations on islands with and without rats (e.g., Brooke et al. 2010; Caut et al. 2008; Jones et al. 2008; Towns et al. 2006). Analyses of Polynesian archaeofaunal assemblages present a consistent trend of rapid disappearances of a range of land bird and seabird species shortly after human (and rat) arrival to Polynesian islands (Steadman 2006). C and N stable isotope analysis of modern Norwegian rats has connected <sup>15</sup>Nenriched rat diets to the decline of nesting island seabird populations (e.g., Hobson et al. 1999; Major et al. 2007). The diet of rats analyzed in these studies fell into several regionallydependent dietary groups, reflecting the observed heterogeneous distributions of island resources. The observed dietary adaptability makes rats particularly effective island predators, and suggests that rat diet is sensitive to local resource availability. Thus, dietary change through time in rat populations can provide insight into localized processes of socio-ecosystem change.

#### **Materials and Methods**

#### Background

Mangarevan prehistory presents a particularly dramatic case of environmental transformation following human—and rodent—arrival. Mangareva is an almost-atoll of 10 small volcanic islands encircled by a barrier reef on the easternmost boundary of French Polynesia. The encompassing barrier reef fosters a rich lagoon environment, standing in stark contrast to a depauperate terrestrial landscape. Archaeobotanical analyses document the transition from indigenous forest cover to a landscape dominated by Polynesian economic plants after human arrival (Kirch et al. 2015). Steep hill slopes with thin soils and little remaining forest cover provide minimal opportunity either for agricultural productivity or nesting seabird habitat. Agriculture is restricted to small coastal arboricultural patches and limited root crop cultivation in valley floors. Early written accounts of Mangareva describe a deforested landscape dominated by *Miscanthus* grasslands (Cooke 1935). However, archaeological excavations have uncovered evidence for a richer, more diverse Mangarevan ecosystem prior to human arrival.

Zooarchaeological analyses document steep declines in indigenous seabird populations shortly after human arrival, as well as the disappearance of three landbird species: an extinct heron and two extirpated columbids (Steadman and Justice 1998; Worthy and Tennyson 2004). Human and rat predation combined with the replacement of primary forest with coconut arboriculture (seabirds generally avoid nesting in coconut palms; Young et al. 2010) likely deleteriously impacted avifaunal communities in early prehistory. Seabird decline would have led to depletion of key soil nutrients from marine subsidies, perpetuating a cycle of terrestrial landscape degradation and resource stress.

Along with rats, Polynesians transported three other terrestrial vertebrates on longdistance colonizing voyages: the domestic pig, dog, and chicken. Remains of all three have been recovered from excavations in Mangareva, though never in abundances that would indicate a strong reliance on these animals for food; all three species disappeared from the islands prior to European contact (Green and Weisler 2004). In contrast, marine resources—especially of inshore and lagoonal species—were abundant. There is little evidence for resource depletion in the lagoon environment and no statistically significant changes in measurements of marine mollusks and fish vertebrae through time in Mangarevan archaeological sites (Conte and Kirch 2004; Kirch et al. 2015).

#### Study Sites

The Pacific rat assemblages analyzed in this study were recovered from recent excavations directed by Patrick Kirch and Eric Conte (Conte and Kirch 2004; Kirch et al. 2010, 2015).

#### The Onemea Site (TAR-6)

The Onemea site on Taravai Island provides the earliest evidence of human occupation in the Mangareva Islands. Multiple radiocarbon dates place the first human activity in Layer III at approximately AD 950 (Conte and Kirch 2004; Kirch et al. 2010). The site appears to have been mainly a campsite for hunters and fishermen. Kirch et al. (2010) characterize Layer III as a palimpsest representing several centuries of repeated, brief visits to the site. Abundant avifaunal remains with taphonomic modifications indicative of human cooking and consumption demonstrate that one purpose of these visits was seabird exploitation. More intensive use of the site is evident in Layer II, where multiple earth ovens and the presence of fishing gear indicate a series of repeated visits to the site over a period of about two hundred years (from ca. AD 1200-1400), after which time Onemea appears to have been largely abandoned. Layer I postdates human occupation of the site.

#### Nenega-Iti Rockshelter (AGA-3)

The Nenega-iti Rockshelter on Agakauitai Island was tested by Conte and Kirch in 2003 (Conte and Kirch 2004), and expanded excavations were carried out at the site in 2012 (Kirch et al. 2015). Rat bone samples were recovered from the main trench (units D9, E9, and F9), which abuts and extends out from the rockshelter wall (see Kirch et al. 2015, fig. 3). The stratigraphic sequence at AGA-3 begins with Layer IV, marking the initial occupation of the site sometime in the 13<sup>th</sup> century AD. Layer IV continues until the mid-17<sup>th</sup> century and is followed by a colluvial deposit (Layer III) that, while thick, may only represent a single depositional event. The final occupation phase, Layer II, terminates around the beginning of the 19<sup>th</sup> century. The site was not used regularly after European contact, although a historic religious medallion recovered from Layer II-1 suggests some brief post-contact visitation. Unlike Onemea, the Nenega-iti Rockshelter was likely a permanent habitation site with evidence for various human activities, including food remains, pearl shell fishhooks, branch coral abraders, bone thatching needles, and numerous basalt adzes.

#### *Kitchen Cave (KAM-1)*

Site KAM-1 on Kamaka Island was given the name Kitchen Cave by Roger Green, who first excavated the site in 1959 (Green and Weisler 2000). Kitchen Cave was re-excavated in 2012 and 2014 under the direction of Kirch, and excavated materials are presently under analysis. Rat elements were selected from both the 2012 and 2014 excavation assemblages. Kamaka is a particularly small, isolated, and resource-poor island even for Mangareva, and it is uncertain whether people continuously occupied the island throughout its prehistory. Kitchen Cave would have been a welcoming location to visitors or inhabitants as an expansive shelter with a permanent cave drip for fresh water. Recent AMS dates on short-lived botanical taxa from the 2012 and 2014 excavations place the first occupation of Kitchen Cave at around AD 1300, with a continuous sequence of occupation phases through the mid-18<sup>th</sup> century (Kirch 2015).

#### Stable Isotope Analysis

Pacific rat elements were selected for stable isotope analysis by calculating Minimum Number of Individuals (MNI) by excavation level to maximize sample sizes while limiting potential resampling of the same individual. Appendicular elements were selected whenever possible, as differential rates of bone turnover have been observed between appendicular and cranial versus axial elements for *R. norvegicus* (Wolfe and Klein 1996). However, as the Pacific rat is a short-lived species it is unlikely that bone turnover rates would significantly differentiate dietary  $\delta^{13}$ C and  $\delta^{15}$ N values between bone elements.

Specimens were sonicated with ultrafiltered water for four hours, dried, and abraded to remove surface contaminants. Samples were cut into ~1 mm chunks and bone collagen was extracted using a modified version of Ambrose (1990; and see Sealy et al. 2014), producing collagen 'pseudomorphs.' Samples over 20 mg were treated with 0.5 M HCl with fresh HCl applied every 24 hours until bone chunks lost their mineral structure and became springy to the touch. Samples weighing less than 20 mg were treated with a more dilute solution of 0.25 M HCl to prevent excessive sample loss. All samples were treated with a 0.1 M NaOH solution for 24 hours to remove humic contaminants. Samples were rinsed and freeze dried for 48 hours.

Dry samples from sites TAR-6 and AGA-3 were analyzed simultaneously for carbon, nitrogen, and sulfur contents (% dry weight) and carbon (C), nitrogen (N), and sulfur (S) stable isotope ratios. KAM-1 samples were analyzed simultaneously for C and N contents (% dry weight) and C and N stable isotope ratios only. Analyses were conducted via elemental analyzer/continuous flow isotope ratio mass spectrometry using a CHNOS Elemental Analyzer (vario ISOTOPE cube, Elementar, Hanau, Germany) coupled with an IsoPrime 100 mass spectrometer (Isoprime Ltd., Cheadle, UK). All isotope analyses were conducted at the Center for Stable Isotope Biogeochemistry at the University of California, Berkeley. Long-term external precision for C, N, and S isotope analyses is 0.1‰, 0.15‰, and 0.5‰ respectively.

Stable isotope data recovered from bone collagen was assessed for various measures of preservation (see Table 2.1 for the results of the analysis and quality control indicators). Modern bone contains approximately 22% collagen by weight, and Pestle and Colvard (2012) recommend a minimum of 0.5% collagen preservation for archaeological bone samples recovered from tropical environments. More stringent requirements applied in other regions range from 1-5% collagen (e.g., Ambrose 1990; Schwarcz and Schoeninger 1991; van Klinken 1999) and the majority of samples met this requirement, with all samples falling above 0.5% collagen preserved. Four samples fell outside the recommended C/N ratio range of 2.9-3.6 (Ambrose 1990; DeNiro and Weiner 1988; van Klinken 1999) and were eliminated from study. Nine samples dissolved during demineralization and were not analyzed.

Samples from TAR-6 and AGA-3 were analyzed for stable sulfur isotope data ( $\delta^{34}$ S), however all resulting values hovered around ~20‰, the same  $\delta^{34}$ S value as ocean water (Chukhrov et al. 1980; Peterson and Fry 1987). Although  $\delta^{13}$ C and  $\delta^{15}$ N values similarly suggest a marine component to many diets (see below),  $\delta^{34}$ S values on small islands such as those of Mangareva are likely heavily influenced by the sea spray effect (Richards et al. 2001). Sulfur stable isotope data will not be discussed further.

Table 2.1: Results of *Rattus exulans* bone collagen carbon and nitrogen stable isotope analysis and quality control indicators from The Onemea Site (TAR-6), Nenega-Iti Rockshelter (AGA-3), and Kitchen Cave (KAM-1).

Sample	Excavation	Lavor	Flomont	Sido	$\delta^{13}C$	δ <sup>15</sup> N (%a)	% C	% N	C:N Ratio
INU. KAM 1 1	Dint B3	Layer	Element	D	(/00)	(/00)	/0 C	15 51	3 24
KAM12	D3	I	Femur	D	-19.5	9.0 14 1	45.07 26.51	0.2	3.24
$KAM_{-1-3}$	B3	I II	Femur	к I	-14.3	14.1	33.25	9.2 11.08	3.31
KAM-1-4	B3	11 11	Femur	I	-14.5	15.4	26.51	9.40	3.24
KAM15	D3	11 11	Femur	L I	-13.7	13.4	20.51	9.40 11.67	3.29
KAM17	D3	11	Femur	L I	-12.0	13.9	52. <del>4</del> 5	14.58	3.24
KAM18	D3		Femur	L I	-12.2	13.2	30.07	14.58	3.54
$K \Lambda M_{-1} Q$	B3		Femur	I	-12.1	12.0	30.16	10.20	3.42
$KAM_{-1} 10$	B3		Femur	I	-14.6	13.7	12 52	14.81	3.35
KAM-1-11	B3		Tibia	L I	-14.8	15.7	42.52	15.06	3.35
$K_{AM-1-12}$	B3		Tibia	T	-17.0	16.2	43.60	15.00	3.32
KAM-1-13	B3		Femur	I	-1/.0	15.2	42 10	14.57	3.30
KAM-1-14	B3		Mandible	I	-14.9	14.3	43 79	15.44	3 31
KAM-1-16	B3		Humerus	I	-13.1	14.5	42.97	14.86	3.31
KAM-1-17	B3	IV1	Femur	R	-19.0	14.1	34.82	11.80	3.47
KAM-1-18	B3	IV1	Humerus	R	-19.0	19.4	43 13	14.94	3.42
KAM-1-19	B3	IV1	Humerus	R	-15.7	11.1	43.15	15.01	3.37
KAM-1-21	B3	IV1 IV1	Innominate	R	-17.2	16.5	43.83	15.01	3 41
KAM-1-21	B3	IV1 IV1	Tibia	R	-17.2	17.3	43.57	14.91	3 41
KAM-1-23	B3	IV2	Femur	L	-16.1	14.8	27 37	9.25	3 45
KAM-1-24	B3	IV2	Femur	L	-19.1	16.1	42.92	14.83	3.38
KAM-1-25	B2	IV2	Innominate	Ē	-15.6	17.5	45.88	16.89	3.17
KAM-1-26	B4	IV2	Femur	Ē	-16.6	18.4	37.66	13.57	3.24
KAM-1-32	C3	IV2	Femur	R	-19.4	16.0	42.74	14.91	3.34
KAM-1-33	C3	IV2	Femur	R	-12.2	14.9	46.08	16.70	3.22
KAM-1-31	B4	V	Humerus	L	-18.9	20.1	45.85	16.86	3.17
KAM-1-34	C3	V	Femur	R	-16.2	16.6	48.14	17.09	3.29
KAM-1-35	C3	V	Femur	L	-14.1	15.1	37.17	12.93	3.37
KAM-1-28	B4	VI	Femur	L	-15.9	19.7	47.07	19.66	3.25
KAM-1-29	B4	VI	Femur	L	-15.5	16.9	37.10	12.88	3.35
KAM-1-30	B4	VI	Humerus	L	-13.7	14.6	49.19	17.08	3.36
AGA-3-2	E9	II1	Mandible	R	-15.4	11.4	46.24	15.63	3.45
AGA-3-3	E9	II2	Tibia	R	-16.7	13.2	45.42	15.12	3.50
AGA-3-4	E9	II2	Humerus	L	-16.8	13.3	43.39	14.02	3.61
AGA-3-5	E9	II2	Humerus	L	-17.4	14.2	47.26	16.85	3.27
AGA-3-6	E9	II2	Humerus	L	-13.8	12.4	43.65	14.45	3.52
AGA-3-7	E9	II2	Humerus	L	-15.7	13.8	42.40	14.79	3.34
AGA-3-8	E9	II3	Mandible	L	-18.2	12.7	39.70	14.14	3.28
AGA-3-9	E9	II3	Mandible	L	-16.4	14.5	42.42	15.01	3.30

AGA-3-10	E9		II3	Mandible	L	-13.9	14.9	41.06	14.02	3.42
AGA-3-11	E9		II3	Femur	R	-18.6	14.2	44.09	14.88	3.46
AGA-3-12	E9		IV3	Innominate	R	-19.6	15.0	40.33	14.14	3.33
AGA-3-13	E9		IV3	Innominate	R	-18.9	10.5	35.96	11.89	3.53
AGA-3-14	E9		IV3	Innominate	R	-15.5	12.9	43.08	15.32	3.28
AGA-3-15	E9		IV3	Innominate	R	-17.5	12.2	42.59	15.41	3.23
AGA-3-16	E9		IV4	Ulna	L	-18.6	12.0	43.81	15.20	3.36
AGA-3-17	E9		IV5	Tibia	L	-14.4	14.0	40.25	14.46	3.25
AGA-3-18	E9		IV5	Tibia	L	-18.5	12.1	38.92	13.10	3.46
AGA-3-19	E9		IV5	Tibia	L	-15.0	12.0	41.81	14.46	3.37
AGA-3-20	E9		IV5	Tibia	L	-17.4	14.6	41.94	14.48	3.38
AGA-3-21	E9		IV5	Tibia	L	-19.2	12.2	40.41	13.42	3.51
AGA-3-22	F9		II1	Femur	R	-11.1	8.7	38.34	12.98	3.45
AGA-3-23	F9		II2	Femur	R	-10.2	7.6	30.11	10.47	3.36
AGA-3-24	F9		II2	Femur	R	-17.9	13.4	43.75	14.23	3.59
AGA-3-25	F9		II2	Femur	L	-17.0	13.6	28.57	9.95	3.35
AGA-3-26	F9		II2	Femur	L	-12.6	10.8	39.02	13.82	3.29
AGA-3-27	F9		II2	Femur	L	-16.3	12.0	33.25	11.66	3.33
AGA-3-28	F9		II2	Femur	L	-18.1	13.8	37.11	11.89	3.64
AGA-3-29	F9		II3	Tibia	R	-16.9	15.1	39.99	14.40	3.24
AGA-3-30	F9		II3	Tibia	R	-18.4	14.5	39.71	14.01	3.31
AGA-3-31	F9		III	Femur	L	-17.9	13.8	20.58	6.93	3.46
AGA-3-32	F9		III	Femur	L	-15.8	12.0	34.14	11.96	3.33
AGA-3-33	F9		III	Femur	L	-18.3	12.7	34.31	11.74	3.41
AGA-3-34	F9		IV1	Femur	L	-19.9	17.0	34.47	11.57	3.48
AGA-3-35	F9		IV1	Femur	L	-17.4	13.2	39.44	14.13	3.26
AGA-3-36	F9		IV1	Femur	L	-18.8	14.5	37.00	13.24	3.26
AGA-3-37	F9		IV1	Femur	L	-17.7	14.4	33.62	11.91	3.29
AGA-3-38	F9		IV2	Humerus	R	-16.9	12.4	37.31	13.12	3.32
TAR-6-1		5	Ι	Mandible	L	-17.9	8.6	46.34	17.23	3.14
TAR-6-2		5	IIA	Tibia	R	-19.2	9.6	48.54	17.78	3.19
TAR-6-3		5	IIA	Tibia	R	-19.6	13.1	47.27	16.49	3.34
TAR-6-4		5	IIA	Femur	R	-19.4	12.8	47.52	16.85	3.29
TAR-6-5		5	IIB	Mandible	L	-19.1	11.6	46.75	17.06	3.20
TAR-6-6		5	IIB	Vertebra	-	-19.6	12.5	44.89	16.00	3.27
TAR-6-7		6	Ι	Mandible	R	-19.6	11.0	46.38	16.25	3.33
TAR-6-8		6	Ι	Mandible	R	-16.2	5.2	47.07	16.84	3.26
TAR-6-9		6	IIA	Tibia	L	-18.7	12.8	48.29	17.31	3.25
TAD ( 10		(	TT 4	Cranial	р	12.4	12.4	17 16	17 15	2.22
1AK-0-10		6		r ragment	к	-13.4	12.4	47.46	1/.13	5.25
1AK-0-11		6 6	пв IIB	numerus	L R	-10.1	14.1	47.56	10.84	3.30
1AK-0-13		11	Ι	1 1b1a	L	-18.9	12.3	45.22	10.10	3.26
1AK-6-16			-	Femur	_	-19.3	6.5	48.38	17.25	3.27

TAR-6-17	11	IIA	Mandible	R	-18.8	8.5	48.99	18.04	3.17
TAR-6-18	11	IIA	Humerus	L	-18.6	8.5	48.65	17.84	3.18
TAR-6-19	11	IIA	Humerus	R	-14.0	12.1	46.37	16.02	3.38

#### **Results**

In small, short-lived mammals, the organic portion of bone (bone collagen) records the chemical signatures of the average diet consumed during the whole lifetime of the animal. Bone collagen is made up of amino acids and is heavily weighted towards the protein component of diet (Ambrose and Norr 1993; Tieszen and Fagre 1993). Carbon stable isotope ratios ( $\delta^{13}$ C) are primarily useful in distinguishing the dietary contributions of C<sub>3</sub> and C<sub>4</sub> plants or marine dietary inputs (DeNiro and Epstein 1978; Schoeninger and DeNiro 1984). Most Polynesian-introduced economic plants are terrestrial C<sub>3</sub> plants, and hence it is largely assumed that C<sub>4</sub> plants are not a significant factor in islander diet, though Jones and Quinn (2009) document some prehistoric human consumption of C<sub>4</sub> plants on Fiji in the form of sea grapes. Although negligible to human diet, sugar cane, tropical grasses and grass seeds are often C<sub>4</sub> plants and may have contributed to rat diet either through direct consumption or consumption of C<sub>4</sub> terrestrial animals. Rat bone collagen  $\delta^{13}$ C values have been reported as ~3% more positive than diet, though this may vary with dietary change (Lee-Thorp et al. 1989). Nitrogen stable isotope values increase as an individual rises through a local food chain. Along each "step" in the food chain, an associated increase in  $\delta^{15}$ N by approximately +3.5‰ is expected (Bocherens and Drucker 2003; DeNiro and Epstein 1981: Minagawa and Wada 1984). Marine food webs are more complex with many levels of consumers and therefore marine carnivores-particularly seabirds and marine mammals—have higher  $\delta^{15}$ N values than terrestrial counterparts.

Results of rat bone collagen stable isotope analysis from all sites are plotted in Figure 2.2, while Table 2.2 presents mean  $\delta^{13}$ C and  $\delta^{15}$ N values for each archaeological site by stratigraphic layer. Individual rat  $\delta^{13}$ C and  $\delta^{15}$ N values vary significantly demonstrating a diverse and adaptable diet. An archipelago-wide trend of declining  $\delta^{15}$ N values emerges with implications for major transformations in nutrient flows within human-centered foodwebs in Mangareva.

Table 2.2. Weah Kanas examines bone conagen o Wand o C values by layer at each site.								
	Excavation	Mean δ <sup>13</sup> C	Mean δ <sup>15</sup> N					
	Layer							
Onemea (TAR-6)	IIB	$-18.4 \pm 1.6\%$	$12.6 \pm 1.1\%$					
	IIA	$-18.0 \pm 2.3\%$	$10.8 \pm 2.3\%$					
	Ι	$-18.3 \pm 1.5\%$	$7.8 \pm 2.5\%$					
Nenega-iti (AGA-3)	IV	$-17.7 \pm 1.7\%$	$13.3 \pm 1.6\%$					
	III	$-17.3 \pm 1.4\%$	$12.9\pm0.9\%$					
	II3	$-17.1 \pm 1.8\%$	$14.3\pm0.9\%$					
	II2	$-15.7 \pm 2.5\%$	$12.5 \pm 1.9\%$					
	II1	$-13.2 \pm 3.0\%$	$10.1 \pm 1.9\%$					
Kitchen Cave (KAM-1)	VI	$-15.0 \pm 2.5\%$	$17.1 \pm 2.5\%$					
	V	$-16.4 \pm 2.4\%$	$17.3 \pm 2.6\%$					
	IV	$-16.9 \pm 2.1\%$	$16.2 \pm 2.3\%$					
	III	$-14.7 \pm 1.7\%$	$14.3 \pm 1.4\%$					
	II	$-14.3 \pm 1.5\%$	$14.7 \pm 1.0\%$					
	Ι	$-17.0 \pm 3.6\%$	$11.8 \pm 3.2\%$					

Table 2.2: Mean *Rattus exulans* bone collagen  $\delta^{15}N$  and  $\delta^{13}C$  values by layer at each site.

#### The Onemea Site (TAR-6)

A total of 16 rat skeletal elements from the main trench at Onemea (Test pits 5, 6, and 11; Kirch et al. 2010) provided bone collagen stable isotope data. Limited sample size resulted in only two samples analyzed from Layer III, both of which dissolved during demineralization. Collagen preservation is a persistent problem in tropical and subtropical environments due to frequent cycles of precipitation and evaporation (see Pestle and Colvard 2012). The Onemea site is more exposed than the two rockshelter sites, and Layer III is comprised of fine-grained Aeolian-transported, calcareous sands with few organics apart from recovered faunal material, conditions that are detrimental to collagen preservation. Absent representation from Layer III, Layer II was subdivided into two components of roughly equal depth to provide chronological resolution. However, these are arbitrary divisions within a continuous cultural layer, so the resulting temporal divisions should be interpreted with some caution. Sample TAR-6-13 was recovered from a pit hearth feature associated with Layer IIB. The sample shows no signs of extreme heating and its isotope ratios are consistent with other samples recovered from the layer (DeNiro et al. 1985).

Rat  $\delta^{13}$ C and  $\delta^{15}$ N values reflect mixed diets composed of C<sub>3</sub> plants with marine or C<sub>4</sub> protein contributions. Three individuals from Layer II have significant marine dietary inputs, indicated by more positive C and N stable isotope ratios. Food sources that may have contributed to enriched  $\delta^{15}$ N values at Onemea include high trophic level reef and marine fish or chickens (*Gallus gallus*) with C<sub>4</sub> diets (Figure 2.3). Through time, rat  $\delta^{13}$ C values remain relatively consistent, while mean  $\delta^{15}$ N values decline through a span of one to two trophic levels from 12.6  $\pm$  1.1‰ in Layer IIB to 10.8  $\pm$  2.3‰ in Layer IIA and finally 7.8  $\pm$  2.5‰ in Layer I (Bocherens and Drucker 2003; Minagawa and Wada 1984). These changes begin in Layer IIA, though small sample sizes limit the certainty of this trend.



Figure 2.2: Distributions of *Rattus exulans* bone collagen  $\delta^{15}N$  and  $\delta^{13}C$  by layer at all three sites.

This shift was likely influenced by the cessation of frequent site use after Layer II. While Mangarevan people were actively hunting and fishing at Onemea, rats would have had opportunities to scavenge high trophic level protein sources including marine fish and seabirds from meal scraps. Without people to provide easy access to high trophic level sources after the site was abandoned, rats likely shifted to wild foraging for plant matter and more easily obtainable terrestrial protein sources, such as arthropods. Rat diet from Layer I may also be influenced by dramatic ecosystem changes brought about by European arrival. The introduction of new predators and competition from the black rat, brown rat, and house mouse may have significantly affected diet and foraging behavior (Shiels et al. 2013). The rats at Onemea have the lowest overall  $\delta^{15}$ N values of all three study sites. This may be related to Onemea's primary

function as a fishing camp with nonpermanent habitation. Less consistent access to human food scraps would create an averaging out of high and low trophic level proteins over the rat's lifetime, resulting in comparatively lower  $\delta^{15}N$  values.



Figure 2.3: Distributions of rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values in parts per mil (‰) for all three case study sites with baseline  $\delta^{13}$ C and  $\delta^{15}$ N values of potential dietary contributions, compiled from previously published material (Allen and Craig 2009; Beavan Athfield et al. 2008; Brooke et al. 2010; Jones and Quinn 2009; Richards et al. 2009). Values have not been adjusted for trophic fractionation.

#### Nenega-Iti Rockshelter (AGA-3)

A total of 37 rat bones were included for analysis from Nenega-iti Rockshelter. Two samples have C:N ratios slightly higher than 3.6, however their stable isotope ratios are in line with other rats from the same context and all other quality control indicators suggest that these collagen samples are intact. Results from Nenega-iti show minimal change in  $\delta^{13}$ C and  $\delta^{15}$ N values over time. Bone collagen samples from the earliest layer (Layer IV) are slightly more enriched in <sup>15</sup>N (13.3 ± 1.6‰) with more negative  $\delta^{13}$ C values (-17.7 ± 1.7‰), and Layer II rat bone collagen  $\delta^{15}$ N values are slightly lower (12.8 ± 2.0‰) with less negative  $\delta^{13}$ C values (-15.9 ± 2.5‰). Layer III rats were recovered from a nearly sterile depositional event with minimal evidence of associated human activity, however their values are similar to those from rats found in Layer IV and Layer II-3. Within Layer II, a trend emerges of  $\delta^{15}$ N values becoming more depleted (from 14.3 ± 0.9‰ in Layer II-3 to 10.1 ± 1.9‰ in Layer II-1) and  $\delta^{13}$ C values becoming more depleted (from -17.1 ± 1.8‰ in Layer II-3 to -13.2 ± 3.0‰ in Layer II-1) through time. Higher and less variable  $\delta^{15}$ N values compared to Onemea likely reflects a sequence of

more permanent human habitation which granted rats frequent access to high trophic level food scraps such as fish, pig, and dog.

#### Kitchen Cave (KAM-1)

Rats at Kitchen Cave present the most elevated  $\delta^{15}$ N values of all three sites analyzed (mean  $\delta^{15}$ N at KAM-1 is  $15.4 \pm 2.4\%$  compared to  $13.0 \pm 1.8\%$  at AGA-3 and  $10.7 \pm 2.6\%$  at TAR-6). The range of  $\delta^{13}$ C values is similar to those from TAR-6 and AGA-3. As with the previous two sites,  $\delta^{15}$ N values are highest in the earliest phase of human occupation (mean  $\delta^{15}$ N in Layer VI is  $17.1 \pm 2.5\%$ ) and decline through time ( $11.8 \pm 3.2\%$  in Layer I). The earliest excavation layers from Kitchen Cave show consistently higher N and variable C stable isotope ratios. A shift towards declining  $\delta^{15}$ N values and less negative  $\delta^{13}$ C values begins in Layer III. Seabirds and marine mammals, which tend to have elevated  $\delta^{15}$ N values due to their high position in marine food webs, are the only food groups sufficiently enriched in <sup>15</sup>N to account for the highest  $\delta^{15}$ N values in Layers IV, V, and VI.

#### Seabird Population Decline and Nutrient Flows on Mangareva

We hypothesize that the archipelago-wide decline in rat bone collagen  $\delta^{15}$ N values in Mangareva is related to dramatic reductions in seabird populations that occurred within two to three centuries following Polynesian arrival (Steadman and Justice 1998; Worthy and Tennyson 2004). Depletion of <sup>15</sup>N in rat bone collagen occurs shortly after a significant decrease is observed in recovered avifaunal remains from all three sites (see Figure 2.4). Diminishment of avifaunal populations could have impacted rat bone collagen  $\delta^{15}$ N in two non-mutually-exclusive ways: directly, through the removal of a <sup>15</sup>N-enriched protein source from rat diet, and indirectly through the removal of <sup>15</sup>N-enriched seabird guano deposits from terrestrial soils.

Early rat  $\delta^{13}$ C and  $\delta^{15}$ N values, particularly from KAM-1, suggest protein inputs from high trophic level marine sources. Reported bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values for seabirds range from approximately -17 to -12‰ and 16 to 18‰ respectively (Brooke et al. 2010; Richards et al. 2009), and for marine mammals -10.3 ± 0.5‰ and 17.6 ± 2.2‰, respectively (Richards et al. 2009). Analyses of zooarchaeological assemblages from Mangarevan archaeological sites provide little evidence for pelagic fishing or hunting of marine mammals. Thus, the most readily available food source to provide high trophic level protein contributions to rat diet would have been seabirds. Throughout Mangareva's archaeological history, people largely procured marine foods from the archipelago's rich lagoon environment, which shows minimal indication of resource depletion through time (Kirch et al. 2015; Weisler and Green 2013). There is no indication that rat diet became more inclusive of terrestrial C<sub>3</sub> inputs over time; it is likely that as seabird communities diminished these dietary inputs were replaced by lagoon resources that would have been obtained via scavenging human food scraps (AGA-3, KAM-1), or by wild foraging for terrestrial C<sub>4</sub> resources (TAR-6).



Figure 2.4: Stratigraphic distributions of avifaunal remains by Number of Identified Specimens (NISP) at all three sites.

Allen and Craig (2009) note similar temporal declines in  $\delta^{15}$ N values for human, pig, and dog bone collagen from Aitutaki, Cook Islands. They attribute this trend to marine resource depression beginning sometime around the 14<sup>th</sup> or 15<sup>th</sup> century, and the faunal record from Aitutaki demonstrates declines in fishing and decreased consumption of marine carnivores. Mangarevan rat stable isotope ratios may document a similar trend, however instead of a depleted marine ecosystem, declining seabird populations and terrestrial resource depression drove the changes observed in rat diet. It appears that Mangarevan rat (and by extension human) diet became more reliant on lagoon resources as seabirds and terrestrial resources diminished. Indeed, decreasing seabird populations may have helped maintain lagoon system productivity in the face of increasing human predation through the elimination of significant avian predation (Wainwright et al. 1998).

Recent studies correlate seabird guano fertilization to greater soil productivity as well as higher baseline N availability and  $\delta^{15}$ N values in island ecosystems (e.g., Anderson and Polis 1999; Briggs et al. 2012; Fukami et al. 2006; Mizota and Naikatini 2007; Szpak 2014). Small

island ecosystems in particular are often heavily reliant on marine-derived nutrient subsidies, and the removal of guano inputs on Mangareva would have impacted agricultural productivity as well as lowered baseline  $\delta^{15}$ N values. Modern artificial fertilizers tend to have  $\delta^{15}$ N values of around 0‰, while plants fertilized with animal manure are generally more enriched in <sup>15</sup>N. Due to the unique process by which seabirds metabolize and excrete nitrogen, seabird guano deposits can have  $\delta^{15}$ N values of around 27‰ (Szpak et al. 2012). Observed  $\delta^{15}$ N values between primary producers fertilized with versus without seabird guano differ by up to +11‰ (Wainwright et al. 1998). Removal of guano from an ecosystem would therefore lower  $\delta^{15}$ N values across primary producers and this effect would resonate through entire island foodwebs, creating an artificial appearance of species trophic decline with no direct change to diet.

Diminishment of seabird populations likely occurred at different rates and intensities on individual islands within Mangareva. Though seabirds are rarely observed on Taravai (TAR-6) or Agakauitai (AGA-3) Islands today, communities of nesting or roosting frigate birds and terns persist on more isolated Kamaka Island (KAM-1). Due to Kamaka's comparatively high degree of isolation and present-day seabird species richness and abundance, it and two other nearby, unoccupied islets in the southern end of the archipelago (Manui and Makaroa) have been proposed as refuges for remaining seabird populations (Waugh et al. 2013). These factors likely also account for the persistently higher  $\delta^{15}$ N values in rat bone collagen analyzed from KAM-1.

#### **Conclusion**

Our analysis indicates that rat dietary  $\delta^{13}$ C and  $\delta^{15}$ N values were influenced both directly and indirectly by the decimation of seabird populations in the Mangarevan island ecosystem. At all three study sites, the elimination of seabirds after Polynesian arrival is reflected in a decline through time in rat bone collagen  $\delta^{15}$ N values, as well as new dietary regimes. At Nenega-iti and Kitchen Cave, less negative  $\delta^{13}$ C values suggest that avifaunal protein was replaced with marine resources from Mangareva's rich lagoon environment. As rats could not have acquired these resources on their own, they likely came in the form of human meal scraps. At Onemea, rats would have come into contact with humans less frequently, especially after site abandonment. As seabird populations declined and Mangarevan people abandoned Onemea, rat diet transitioned to foraging for terrestrial C<sub>3</sub> and C<sub>4</sub> resources.

Soil-plant dynamics interact with baseline ecosystem <sup>15</sup>N enrichment in numerous complex ways, and recent studies have only begun to highlight the implications these factors have for stable isotopic reconstructions of ancient diets (see Szpak 2014). Our results suggest that seabird guano input to soil nutrient cycling is a significant variable that can greatly impact N stable isotope ratios in small island ecosystems. Future stable isotope analyses of archaeological materials in these regions must carefully consider the significance of ecosystem dynamics in reconstructing past diet. Additional research into the complexities of shifting baseline  $\delta^{13}$ C and  $\delta^{15}$ N values and their distributions throughout foodwebs is critical for proper contextualization and interpretation of stable isotope ratios.

#### Chapter 3

#### Coastal subsistence and settlement at the Hane Dune Site, Ua Huka (Marquesas Islands): New insights from Pacific rat (*Rattus exulans*) stable isotope analysis

#### **Introduction**

The Hane dune site on Ua Huka Island in the Marquesas figures prominently in longstanding discussions over the chronology of East Polynesian settlement and prehistoric Marquesan subsistence. Pioneering excavations by Yosihiko Sinoto (Sinoto 1966; Sinoto and Kellum 1965) demonstrated the site's long archaeological sequence and early radiocarbon dates likely associated with the first phase of human occupation on the archipelago. Although its chronology and prehistory has long been debated by archaeologists, the site itself has remained virtually untouched since Sinoto's intensive excavations in 1964-5. Here we revisit the Hane dune site to analyze materials from renewed excavations in 2009 (see Conte and Molle 2014) and apply a new technique towards investigating subsistence and ecology in the Marquesas. We present bone collagen stable isotope  $\delta^{13}$ C and  $\delta^{15}$ N data from 33 Pacific rat (*Rattus exulans*) specimens recovered from three phases of site use at the Hane dune site to investigate changing site use and subsistence practices in the Hane valley.

Guiry and Gaulton (2015) have reviewed the potential applications and limitations of utilizing stable isotope analysis of rat remains as a human behavioral proxy. Their analysis of black rats (Rattus rattus) from Ferryland, Newfoundland indicates commensal rats can provide localized insights into changing site use, foodways, and activity patterns. The Pacific rat was selected for this study due to its omnivorous diet, small home range, and wide distribution throughout the Pacific (Roberts 1991; Williams 1973). This commensal species is often found associated with human activity in the earliest layers of Polynesian archaeological deposits. AMS radiocarbon dates of Pacific rat bone collagen have been used as a proxy for the arrival of human settlers to new islands, and have contributed to the development of the commensal model of colonization in the Pacific (Anderson 2004; Kirch 2011b; Matisoo-Smith 1994, 2002, 2009; Matisoo-Smith et al. 1999; Wilmshurst et al. 2008). It remains unclear whether the Pacific rat was introduced to the Polynesian islands intentionally or as inevitable stowaways on voyaging canoes. Rats were reportedly considered a food source on some islands, including Mangaia, Niue, and Easter Island (Matisoo-Smith et al. 1999). On Mangaia, rats were elevated to the level of a delicacy, and the phrase "sweet as a rat" was employed to favorably describe foods (Gill 1894:315; Kirch et al. 1995). In the Marquesas, rats were evidently not a valued food source. However, Handy notes that it was rumored in Taiohae, Nuku Hiva, that the impoverished upland-dwellers with limited access to marine resources were driven to eating rats either braised or mashed raw with breadfruit paste (Handy 1923:199).

Pacific rats are omnivores with a diet that varies seasonally and between regions (Bunn and Craig 1989; Shiels et al. 2013). Although modern observational and gut-content studies have shown that rat diet is primarily vegetal, rats have been observed to predate a range of species including insects, turtle hatchlings, small birds and bird eggs (Norman 1975; Duron et al. 2015; Spenneman 1997). However, modern studies of Pacific rat diet are not necessarily reliable analogs for the past, as Polynesian landscapes have undergone massive transformations since European contact and the introduction of new predators and competition has likely further impacted Pacific rat diet. Pacific rat dietary selectivity may prohibit the utility of this species as a
*direct* proxy for human diet; however, temporal changes in rat diet provide useful insight into transformations in local site ecology and patterns of human activity and resource use.

# Subsistence and Settlement in the Marquesas Islands

### The Hane Dune Site and the Marquesan Cultural Sequence

The first archaeological investigation of the Marquesas Islands was undertaken by Robert Suggs (1961), who conducted extensive survey and excavation on the island of Nuku Hiva. Suggs established a cultural sequence for the region which—though his absolute chronology has proven problematic-stands as an influential outline of settlement and prehistory for the entire Marquesan archipelago (Sinoto 1970; Rolett 1998). More recently, Allen (2004) presented an updated chronology for this sequence, maintaining the names and overall themes of Suggs' original culture historical periods: The Settlement Period (AD 900-1100) sees the first arrival of Polynesians to the islands, and is characterized by small, less permanent habitation in the most favorable (largely coastal) areas of the islands. Subsistence was focused on experimentation with agricultural systems and the exploitation of easily depleted indigenous prey such as naïve avifauna, turtles, and marine mammals. The Developmental Period (AD 1100-1300) shows a decline in native fauna along with more permanent habitation and increasing reliance on agricultural products. In the Expansion Period (AD 1300-1600), populations grow and begin to move into valley interiors, and the reliance on agricultural and domestic products intensifies. Finally, the Classic Period (AD 1600-1970) is marked by increasing conflict between settlements in neighboring valleys, accompanied by movement of habitation sites further inland and the elaboration of monumental architecture.

The Hane dune site, an expansive dune complex located in Hane Bay on the southern end of Ua Huka Island (Figure 3.1), was first excavated by Yosihiko Sinoto in 1964-5 (Sinoto 1966). Sinoto's investigations uncovered a continuous stratified deposit dating back to the earliest phase of human occupation in the archipelago; consequently, the site has played a key role in understanding Marquesan colonization and prehistory (Allen 2004; Conte and Molle 2014; Sinoto 1970). Sinoto compared the Hane assemblage with the artifacts used in Suggs' cultural sequence and suggested that the Hane site likely contained deposits dating to even earlier than those found at Nuku Hiva. Following debate over the accuracy of the radiocarbon chronology established by Sinoto (see Anderson et al. 1994; Kirch 1986; Rolett 1998; Spriggs and Anderson 1993), the Hane dune site was reexcavated by G. Molle and E. Conte in 2009. Reevaluation of the site established a new chronological and cultural sequence with three main phases (see Conte and Molle 2014; Molle 2011). The earliest phase (Phase I) marks the initial settlement of the dune at around 900 CE. Phase II establishes long-term sustained human occupation and the development of the site into a coastal hamlet around the beginning of the 13<sup>th</sup> century. After this, the site complex appears to have been abandoned and occupation shifted to the valley interior. Phase III shows a return to the site along with a drastic change in site use: the appearance of human burials and probable funerary structures in the 15<sup>th</sup> and 16<sup>th</sup> centuries suggests that the site functioned as a cemetery, rather than as a residential complex, in later prehistory.



Figure 3.1: Map of the Marquesas Islands with the Hane dune site labeled, and inset map of French Polynesia.

#### Subsistence trends in the Marquesas

Ethnographic and historic accounts (e.g., Dening 1974; Handy 1923) describe the Marquesan subsistence economy as one dominated by breadfruit arboriculture and storage. As in other Polynesian societies, the Marquesan meal consisted of a starch base (often breadfruit paste or *popoi*), which could be eaten on its own or complemented by a flesh food "relish" to form a complete meal (Kirch and O'Day 2003). The islands were plagued by frequent periods of drought and to stave off famine, fermented breadfruit (*ma*) was stored underground in large fermentation pits. The intensification of pig husbandry in the Marquesas may also have been a response to unpredictable harvests, as a form of storage 'on the hoof' (see Dye 2014; Vayda et al. 1961; West 2007). However, high value foods were often under the control of the island's elite, who maintained full orchards of breadfruit trees and the largest *ma* pits. They managed the distribution of communal *ma* in times of drought, as well as the distribution of pig meat at feasts and large fish catches (Dye 1990; Handy 1923:168).

It is difficult to extricate pre-contact changes in subsistence practices at the Hane dune site from those of the entire archipelago, as studies from Hane have helped form the basis of much of our understanding of past Marquesan subsistence. Kirch's (1973) pioneering analysis of floral and faunal assemblages from Ha'atuatua, Nuku Hiva Island, the Hane dune site, and several other sites on Ua Huka Island was the first to demonstrate a temporal shift from marineoriented hunting and gathering activities to terrestrially-focused subsistence through agriculture and animal husbandry. According to Kirch's analysis, at Hane, seabirds and other marine resources comprised a significant portion of the Settlement Period faunal assemblage. During the Developmental Period, focus began to shift towards an agricultural economy centered on breadfruit and other Polynesian-introduced crops. This terrestrial focus intensified during the Expansion and Classic Periods, with reduced fishing and increasing investment in pig husbandry. Dye's (1990) analysis focusing on marine resources from the same sites further confirms the later preference of pig and shellfish (inland and nearshore prey) over offshore marine resources.

Sweeney et al. (1993) questioned these studies by Kirch and Dye, suggesting that aggregating data from multiple sites, utilizing bone weight and meat weight to quantify resource use, and evaluating these data through the use of closed arrays (percentage comparisons) biased interpretations of the relative importance of terrestrial versus marine resource contributions. Using rank order abundances and focusing solely on the Hane dune site assemblage, Sweeney et al. found minimal change in the subsistence economy apart from a drop-off in avifaunal remains over time. Anderson et al.'s (1994) reanalysis of the Hane sequence likewise supported the decline of avifauna and turtles, but did not indicate an increase in pig or shellfish remains through time. They argued that there was little evidence for a shift from marine to terrestrial resource use at the site. Dye (1996), however, pointed out that in limiting themselves to the Hane dune site assemblage, Sweeney et al.'s (1993) sample size was significantly reduced, leading to sampling bias. Dye further argued that the closed arrays used in his and Kirch's (1973) analyses speak to the relative importance of food groups through time which, though this may not establish absolute increases in terrestrial mammal remains, demonstrates real trends in the waning importance of marine exploitation and intensification of agriculture and pig husbandry. Davidson et al. (1999) endeavored to reanalyze the fish remains from Hane, newly identifying tuna (Scombridae) remains in the assemblage through the presence of caudal vertebrae. These demonstrated a statistically significant decline in tuna fishing through time, though they emphasized that tuna fishing remained a comparatively important activity even in late prehistory.

Research from elsewhere in the Marquesas supports a shift from frequent marine exploitation to intensified arboriculture and pig husbandry. At the Hanamiai dune site, Tahuata Island, Rolett (1992, 1998) documented a decline in avifaunal species, increasing emphasis on arboriculture and pig husbandry, and a movement from off-shore fishing to inshore marine resource exploitation, a similar pattern to that seen at Hane by Kirch (1973) and Dye (1990). Stable isotope analysis of pig remains from Hanamiai suggests pig husbandry began around AD 1200 and intensified through time, demonstrated by restricted pig dietary variability in later prehistory (Richards et al. 2009; West 2007). The diets of rat, pig, and dog at Hanamiai all indicated greater inputs of marine protein during the early phases of occupation at the site, followed by increasing focus on terrestrial resources and the disappearance of avifauna (Richards et al. 2009; Steadman and Rolett 1996). At Anapua Rockshelter, Ua Pou Island, Leach et al. (1997) likewise found that although tuna remains were more prevalent than in most other Polynesian sites, offshore fishing—and perhaps all fishing practices—appeared to decline over time. Employing C and N stable isotope analysis of the Pacific rat as a new line of evidence towards understanding changes in site use and resource distribution can aid in resolving these debates over long-term subsistence patterns in the Marquesas Islands.

# **Materials and Methods**

Bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N stable isotope analysis has proven effective in reconstructing the diets of individual humans and animal species (Katzenberg 2008; Sealy 2001). Bone collagen is synthesized from dietary amino acids and is thus weighted towards the protein component of diet (Ambrose and Norr 1993; Tieszen and Fagre 1993). The abundance of C and N stable isotopes in bone collagen are expressed in  $\delta$  notation as ratios relative to established standards and measured in parts per mil (‰), where:  $\delta = [(Rsample/Rstandard) - 1] * 1000$ . R represents the ratio of stable isotopes, here either  ${}^{13}C/{}^{12}C$  or  ${}^{15}N/{}^{14}N$ . The standards for C and N stable isotope analysis are Vienna Pee Dee Belemnite (VPDB) and atmospheric nitrogen (also known as Ambient Inhalable Reservoir—AIR), respectively.

The photosynthetic pathways of C<sub>3</sub>, C<sub>4</sub>, and CAM plants lead to differential stable carbon isotopic fractionation, with C<sub>3</sub> plants having  $\delta^{13}$ C values in the range of -34 to -21‰, and C<sub>4</sub> plants in the range of -15 to -8‰. CAM plants can resemble either C<sub>3</sub> or C<sub>4</sub> plants, and often have  $\delta^{13}$ C values falling between those of C<sub>3</sub> and C<sub>4</sub> plants (Bender 1971). This differential fractionation is carried up the food chain resulting in more negative  $\delta^{13}C$  values in C<sub>3</sub> consumers than in C<sub>4</sub> consumers. Marine primary producers take in carbon from a mix of sources, though generally have carbon stable isotope ratios resembling those of C<sub>4</sub> plants. Thus, carbon stable isotope ratios can also aid in distinguishing terrestrial versus marine inputs (Schoeninger and DeNiro 1984).  $\delta^{13}$ C values in rat bone collagen have been reported as roughly 3‰ more positive than diet, however this fractionation may be influenced by variations in diet (Lee-Thorp et al. 1989).  $\delta^{15}$ N values further aid in assessing relative marine and terrestrial dietary contributions, as well as trophic position. A stepwise <sup>15</sup>N-enrichment between trophic levels results in consumers having  $\delta^{15}$ N values roughly 3.5% higher than their food sources (Bocherens and Drucker 2003; DeNiro and Epstein 1981; Minagawa and Wada 1984). Marine foodwebs tend to be more complex with many levels of consumers, so these systems are generally more enriched in <sup>15</sup>N, with higher  $\delta^{15}$ N values, than their terrestrial counterparts.

The rat bone elements analyzed in this study were selected from the faunal assemblage of the 2009 reexcavations of the Hane dune site (see Conte and Molle 2014). Elements were

selected for stable isotope analysis by calculating Minimum Number of Individuals (MNI) by layer within each 1 x 1 m excavation unit in order to maximize sample sizes while minimizing the chance of resampling from the same individual. Effort was made to select appendicular over cranial or axial elements, as these three categories have previously demonstrated differential rates of turnover in brown rats (*Rattus norvegicus*; Wolfe and Klein 1996). However, some axial elements were selected due to small sample sizes. It is assumed that bone collagen turnover rates will not significantly impact dietary  $\delta^{13}$ C and  $\delta^{15}$ N values between elements, as *Rattus exulans* is a relatively short-lived and small-bodied species, surviving around 1 year in the wild (Atkinson and Towns 2001). Although bone collagen turnover occurs more rapidly in small-bodied animals, bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values reflect a long-term average of diet and likely represent the majority of the rat's short lifespan.

All bone elements were sonicated with ultrafiltered water for four hours, dried, and abraded to remove adhering surface contaminants. Samples were then cut into approximate 1 mm chunks and collagen was extracted following a modified version of Ambrose (1990; and see Sealy et al. 2014). Samples weighing over 25 mg were treated with 0.5 M HCl with fresh acid applied every 24 hours until bone chunks demineralized. Samples weighing under 25 mg were treated with a more dilute 0.25 M HCl solution in order to slow reaction time and prevent excessive sample loss. After demineralization, all samples were treated the same. Samples were then rinsed with ultrafiltered water and treated with 0.1 M NaOH for 24 hours to remove humic contaminants. Following treatment, samples were rinsed again and freeze dried for 48 hours. Dry samples were analyzed simultaneously for C and N contents (% dry weight) and C and N stable isotope ratios. Analyses were conducted via elemental analyzer/continuous flow isotope ratio mass spectrometry using a CHNOS Elemental Analyzer (vario ISOTOPE cube, Elementar, Hanau, Germany) coupled with an IsoPrime 100 mass spectrometer (Isoprime LTd., Cheadle, UK). Analyses were conducted at the Center for Stable Isotope Biogeochemistry at the University of California, Berkeley. Long-term external precision for C and N isotope analysis is 0.1‰ (C) and 0.15‰ (N).

#### **Results**

The results of *R. exulans* bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N stable isotope analysis are presented in Table 3.1, along with quality control indicators. The primary means of assessing collagen preservation is the ratio of carbon to nitrogen in the sample, or the C/N ratio. The C/N ratio should resemble that of modern bone collagen, and samples are likely to be contaminated if they fall outside the range of 2.9-3.6 (Ambrose 1990; DeNiro and Weiner 1988; van Klinken 1999). One sample did not meet this criterion and was eliminated from analysis. Modern bone contains approximately 22% collagen by weight, and the recommended amount of collagen preservation in archaeological samples recovered from most environments ranges from 1-5% (Ambrose 1990; Schwarcz and Schoeninger 1991; van Klinken 1999). However, in tropical environments where bones are exposed to high temperatures, frequent rains, and microbial activity, this is more difficult to attain. Pestle and Colvard (2012) demonstrate that 0.5% collagen preservation in bones recovered from tropical and subtropical environments can still be viable. Other secondary indicators of collagen preservation include the concentration of carbon and nitrogen in bone collagen. Intact bone collagen contains roughly 35% carbon by weight, and should contain around 11-16% nitrogen by weight (Van Klinken 1999).

Sample	Sauana	Lavan	Cultural Phase	Flomont	Side	s <sup>13</sup> C	s15N	9/ C	0/ N	C·N	9/ Call
Number	Square	Layer	Phase	Element	Side	0 C	0 N	% C	% N	C:N	% Coll
HAN-I	E8	A	11	Tili	ĸ	-18.5	ð.1	44.90	10.31	3.22 2.10	15./1
HAN-2	E10	B	1112	1101a	L	-18.0	11.2	46.48	17.01	3.19	4.26
HAN-3	E/	B/C	1112 1111	Innominate	ĸ	-18.6	11.2	46.48	1/.01	3.19	11.56
HAN-4	E7	C			L	-17.5	11.2	45.41	16.51	3.21	12.40
HAN-5	E7	C		Tibia	L	-19.0	13.1	46.81	16.92	3.23	11.90
HAN-6	E7	С	1111	Femur	R	-18.4	11.1	46.19	16.80	3.21	9.89
HAN-7	E7	С	III1	Femur	R	-17.6	11.9	45.63	16.39	3.25	9.06
HAN-8	E8	С	III1	Humerus	L	-19.0	12.3	46.16	16.70	3.22	13.95
HAN-9	E8	С	III1	Humerus	L	-17.1	9.9	44.17	16.08	3.20	3.79
HAN-10	E7	E	II	Femur	R	-18.2	13.4	45.02	15.40	3.41	3.81
HAN-11	E7	E/F	II	Femur	R	-12.8	12.0	49.95	16.66	3.50	6.39
HAN-12	E8	F	II	Ulna	R	-17.2	9.1	46.64	16.46	3.31	6.73
HAN-13	E9	F	II	Femur	R	-17.5	15.2	33.03	11.68	3.30	1.55
HAN-14	E7	G	II	Femur	L	-14.2	17.1	47.07	16.85	3.26	4.10
HAN-15	E8	G	II	Tibia	L	-16.1	15.9	41.68	14.47	3.36	2.73
HAN-16	E8	G	II	Tibia	L	-16.6	19.7	46.42	16.16	3.35	4.29
HAN-17	E9	G	II	Mandible	R	-14.5	16.4	46.45	16.54	3.28	3.78
HAN-18	E9	G	II	Mandible	R	-19.0	11.2	43.12	15.44	3.26	4.29
HAN-19	E9	G	II	Mandible	R	-16.3	18.1	48.88	17.77	3.21	9.95
HAN-20	E7	Н	Ι	Humerus	L	-16.0	17.3	42.95	14.20	3.53	4.76
HAN-21	E7	Н	Ι	Humerus	L	-16.1	15.7	47.45	15.55	3.56	8.52
HAN-22	E7	Н	Ι	Humerus	L	-14.1	15.2	42.77	15.13	3.30	7.39
HAN-23	E7	Н	Ι	Humerus	R	-17.4	13.7	44.42	15.55	3.33	12.40
HAN-24	E9	Н	Ι	Femur	R	-15.7	14.5	43.43	15.49	3.27	6.97
HAN-26	E9	Н	Ι	Femur	L	-17.3	13.8	42.07	13.85	3.54	4.67
HAN-27	E8	Ι	Ι	Humerus	L	-20.8	8.1	45.28	15.73	3.36	1.31
HAN-28	E8	Ι	Ι	Humerus	L	-16.4	12.6	43.14	15.36	3.28	8.41
HAN-29	E9	Ι	Ι	Humerus	R	-12.6	16.2	43.95	15.35	3.34	3.24
HAN-30	E9	J	Ι	Femur	L	-13.9	15.2	35.89	11.94	3.51	3.67
HAN-31	E9	J	Ι	Innominate	R	-14.1	14.8	35.94	12.56	3.34	8.22
HAN-32	E10	J	Ι	Femur	R	-13.1	15.3	42.44	15.03	3.30	7.34
HAN-33	E10	J	Ι	Mandible	L	-16.3	15.5	43.33	15.67	3.23	8.93
HAN-34	E10	J	Ι	Mandible	R	-13.8	16.7	44.12	15.25	3.37	3.04

# Table 3.1: Results of carbon and nitrogen stable isotope analysis of *Rattus exulans* bone collagen, along with quality control indicators.

Rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values have been plotted by the cultural phases for the Hane dune site established by Conte and Molle (2014) in Figure 3.2. A final phase, Phase IV, was established for this study to demarcate the top aeolian sand layer deposited after the site was no longer under permanent use (designated Layer A by Conte and Molle). Table 3.2 provides

mean  $\delta^{13}$ C and  $\delta^{15}$ N values for each cultural phase. Rat diet is similar in Phases I and II, though in Phase II  $\delta^{15}$ N values vary more widely (14.8 ± 3.3‰ compared to 14.6 ± 2.2‰ in Phase I), while  $\delta^{13}$ C values become slightly more negative (-16.2 ± 1.9‰ compared to -15.5 ± 2.2‰ in Phase I). It is likely that due to more permanent human habitation in Phase II, rats had more frequent access to high trophic level marine sources as well as more variety in available food scraps. In both Phases I and II, rat diet appears to be a mixture of C<sub>3</sub> and C<sub>4</sub> sources with significant inputs from high trophic level marine sources. Rat diet becomes more terrestrially focused in Phase III, with  $\delta^{13}$ C becoming more negative (-18.1 ± 0.8‰) and  $\delta^{15}$ N values declining by roughly one trophic level (11.5 ± 1.0‰; Bocherens and Drucker 2003; Minagawa and Wada 1984). Only a single sample represents rat diet after site abandonment (Phase IV), but this continues the trend towards more terrestrial and lower trophic level inputs. However, a more precise temporal provenience cannot be established for this element, and dietary change here may be largely a byproduct of the large-scale transformations to Marquesan ecosystems following European arrival.

Phase	n	Mean δ <sup>13</sup> C	Mean δ <sup>15</sup> N
Ι	14	$-15.5 \pm 2.2\%$	$14.6\pm2.2\%$
II	10	$-16.2 \pm 1.9\%$	$14.8\pm3.3\%$
III	8	$-18.1 \pm 0.8\%$	$11.5\pm1.0\%$
IV	1	-18.3‰	8.1‰

Table 3.2: Mean rat bone collagen  $\delta^{13}C$  and  $\delta^{15}N$  values by cultural phase from the Hane dune site.

Richards et al. (2009) report  $\delta^{13}$ C and  $\delta^{15}$ N values of several marine species from the Marguesas Islands. These data suggest that rats in Phases I and II were likely consuming marine carnivores such as snappers, groupers, jackfish, and tuna. For example, reported  $\delta^{13}$ C and  $\delta^{15}$ N values for two species of tuna (Gymnosarda unicolor and Katsuwonus pelamis) are  $-11.5 \pm 2.5\%$ and  $12.3 \pm 2.7\%$ , respectively. Rats with the most elevated  $\delta^{15}$ N may also have been consuming seabirds, whose reported bone collagen  $\delta^{13}$ C values range from -17 to -12.2‰ and  $\delta^{15}$ N values from 15-18.4‰ (Brooke et al. 2010; Richards et al. 2009). Surprisingly, reported  $\delta^{15}$ N values for two species of parrotfish (Scarus sp. and Calotomus sp.)-a predominantly herbivorous reef fish—are much higher than those of most other fish analyzed at  $18.8 \pm 0.4\%$  (see Richards et al. 2009 Appendix for  $\delta^{13}$ C and  $\delta^{15}$ N values of additional marine species). This variation speaks to the complexity of marine foodwebs as well as variations in diet and ecology within fish families. Shifts in both  $\delta^{13}$ C and  $\delta^{15}$ N values in Phase III indicate rat diet became more oriented towards terrestrial and C<sub>3</sub> resources, though there may still be some inputs from nearshore fish and shellfish. Elevated  $\delta^{15}$ N values in some rats may also be associated with increased consumption of terrestrial domesticates, particularly chickens (Gallus gallus) and pigs (Sus scrofa). Reported  $\delta^{13}$ C values for chickens recovered from Pacific Islands range widely (-21 to -15‰), depending on region and resource availability.  $\delta^{15}$ N values for chickens (10.4 ± 1.5‰) tend to fall within the lower end of reported values for inshore fish (Jones and Quinn 2009; Storey et al. 2010). Dietary and osteological evidence from Marguesan archaeological contexts suggests that pigs were more intensively managed in later prehistory, and their diets varied from terrestrial- to marine-based depending on the husbandry strategies employed (Richards et al. 2001; West 2007).



Figure 3.2:  $\delta^{13}$ C and  $\delta^{15}$ N values for Pacific rat (*Rattus exulans*) bone collagen by cultural phase at the Hane dune site.

#### **Discussion**

Rat dietary change at the Hane dune site as expressed through bone collagen C and N stable isotope ratios corresponds with the interpretations of Hane subsistence proposed by Kirch (1973) and Dye (1990). Rat diet shifted from having access to high trophic level marine proteins in Phases I and II to a more terrestrially-focused diet in Phase III, with less frequent access to marine sources and protein inputs from a lower trophic level. Figure 3.3 plots rat dietary  $\delta^{13}$ C and  $\delta^{15}$ N values from all three cultural phases in relation to potential food sources. Early marine proteins may have come from pelagic and bottom-dwelling carnivorous fish or seabirds. These resources were likely available in the form of leftover meal scraps for rats who occupied human households as their primary ecological niche, though rats may also have hunted for seabird hatchlings and eggs (Norman 1975; Spenneman 1997). These high trophic level marine resources would have become scarce in Phase III as seabird populations were greatly diminished, and as people shifted their residences from the coast to inland locations.

The change in activity at the Hane dune site from permanent habitation site to largely ceremonial complex between Phases II and III likely also impacted food consumption practices, limiting the resources available to commensal scavengers such as rats. Without permanent human occupation at the site, rats may have been forced to rely more frequently on wild foraging for terrestrial resources. Marine contributions to rat diet in Phase III seem to have been reduced though not entirely eliminated. Even after the cessation of site use in the early 17<sup>th</sup> century. Conte and Molle (2014) suggest that Hane Bay would have been a desirable fishing locale, and likely continued to be visited by groups of fishermen either living further inland or in another location within the coastal area. However, the frequency and nature of fishing practices may have changed: marine dietary inputs in Phase III appear to come from low trophic level fish and shellfish, likely exploited from nearshore habitats. Analysis of fishhook assemblages from the Marquesas further confirm this movement away from offshore fishing, as more diverse fishing kits in early prehistory were replaced with assemblages predominantly of the simple, jabbing hook type (Sinoto 1966; Suggs 1961; Rolett 1998). Dye (1990) suggests these changes resulted from intensified social stratification and warfare, processes which gave elites more control over canoes and access to prime fishing locations, and which made offshore excursions a more dangerous venture. Hook-and-line fishing with simple jabbing hooks, along with other fishing

methods which leave no trace in the archaeological literature such as netting, trapping, and poisoning, would have been ideal for exploitation of the nearshore environment.

Davidson et al.'s (1999) reanalysis of the ichthyofaunal remains recovered from Sinoto's excavations of the Hane dune site suggest that although pelagic tuna fishing declined through time, it remained an important aspect of fishing practices at Hane into late prehistory. This insight provokes an exploration into the different types of evidence gleaned from the faunal versus stable isotopic indices. As Allen and Craig (2009) note, these two archives each come with their own set of potential biases but when combined can better inform interpretations of past subsistence. Zooarchaeological assemblages are subject to taphonomic and recovery processes which may favor certain species or elements, as well as cultural depositional processes which accumulate the activities of numerous groups over large spans of time. In contrast, dietary information from stable isotope analysis is tied to particular individuals—introducing bias through dietary preference—and represents a shorter period of time. In this case, the analysis of Pacific rat elements represents individual diet for around one year (Atkinson and Towns 2001). Davidson et al. found an MNI of 14 for late-period Scombridae, compared to 94 in the early period (Davidson et al. 1999; Table 7). This small number—the product of a few fishing excursions—is not likely to significantly influence rat stable isotope  $\delta^{13}$ C and  $\delta^{15}$ N values.



Figure 3.3:  $\delta^{13}$ C and  $\delta^{15}$ N values for Pacific rat (*Rattus exulans*) bone collagen from all three cultural phases plotted with background values for potential food sources. Dotted circle highlights dietary shift in Phase III. Values have not been adjusted for dietary fractionation. Background values estimated from Allen and Craig 2009; Brooke et al. 2010; Kinaston et al. 2014; Richards et al. 2009; Storey et al. 2010.

It is also interesting to note that the majority of Scombridae remains from Hane identified by Davidson et al. (1999) are caudal vertebral elements. Handy (1923:167-168) describes the process of dividing a successful catch onshore, while afterwards families returned to their homes further inland to enjoy the spoils. Occasionally—particularly after important deep-sea expeditions—the chief would take the catch further into the valley to his "feast place" or a ceremonial plaza (*tohua*) to divide the meal. The head of the fish, as the fattiest, oiliest and often the portion most valued by Polynesians (Kirch and O'Day 2003; O'Day 2004) was likely the domain of the chief and consumed where the elites resided. This activity would have become more prevalent through prehistory as chiefs gained more power and control over fishing practices. Further, it suggests that a large portion of fish catches would have been consumed and disposed of at inland sites, inaccessible to coast-dwelling rats. These depositional factors may result in an underrepresentation of the actual amount of offshore fish consumed in the later Hane faunal assemblages. However, an increase in the overall abundance of shellfish and inshore fish remains in later prehistory still points to significant changes in the Marquesan fishing industry on Ua Huka Island (Kirch 1973; Dye 1990).

In aggregate, stable isotope data from multiple *R. exulans* individuals provides insight into large-scale trends through time at a particular location. The Pacific rat's limited home range suggests that although these data may not speak to island-wide trends, they can provide a new line of evidence towards interpreting the subsistence record at the Hane dune site. Differences between human and rat dietary preferences may limit isotopically reconstructed rat diet's use as a direct proxy for human diet; however, significant shifts in  $\delta^{13}$ C and  $\delta^{15}$ N values between Phases I/II and Phase III demonstrate a clear transition in the site's occupational history and intensity of use. This provides a new window into understanding localized patterns of activity and subsistence practices in the coastal region of Hane.

# **Conclusions**

As one of the most studied sites in the Marguesas, settlement and subsistence practices at Hane figure prominently in present understandings of the archipelago's long-term prehistory. The results presented here conform to initial interpretations of Hane subsistence as primarily marine-oriented in its earliest phases, later transitioning to a terrestrial economy of breadfruit arboriculture and intensive pig husbandry (Kirch 1973; Dye 1990). Pacific rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values indicate a reduction in both quantity and quality of marine inputs available at the dune complex by Phase III. Although Hane likely continued to act as a prime fishing location for the inhabitants of the valley, marine resource exploitation represented in the Hane coastal area became focused on nearshore resources such as shellfish and low trophic level inshore fish. Catches from the less frequent offshore fishing excursions were likely brought inland to be reallocated by the chief and consumed. The revised chronological sequence of the Hane dune site (Conte and Molle 2014) indicates this transition occurred around the 14<sup>th</sup> century following an initial abandonment of the dune for several decades. Broadly, this would place these developments in the early Expansion Period (Allen 2004), consistent with the settlement pattern and subsistence trends outlined by Suggs (1961) for Nuku Hiva. Investigations of other sites in the Marquesas (e.g., Allen 2004; Rolett 1998) suggest that similar subsistence and settlement trends occurred throughout the archipelago, though with local variation in timing and process.

The results presented here would not be properly contextualized without previous intensive analyses of Hane's archaeofaunal record. A detailed analysis of the rich assemblage of marine remains from the 2009 excavations is currently in progress and will bring further insight into long-term changes in marine subsistence at Hane. This study provides a new line of evidence into human subsistence and activity at the Hane dune site, suggesting that cultural depositional practices—particularly in later prehistory—may introduce bias both the faunal and isotopic subsistence archives. Stable isotope analysis of small commensal species can provide new insight into local site activity and ecology, as well as complement archaeofaunal and ethnographic analyses to create greater resolution on past subsistence activity.

#### Chapter 4

# Pacific rat (*Rattus exulans*) diet as proxy for human-ecosystem dynamics on Tikopia: A stable isotopic record of long-term stability in a Polynesian Outlier

### **Introduction**

The island of Tikopia is often presented as an example of a long-term sustainable island society with a resilient agricultural ecosystem (e.g., Diamond 2005; Kirch 2007a; Mertz 2002). Tikopia is a Polynesian "Outlier," one of about 18 islands located geographically outside the Polynesian triangle region but whose inhabitants speak Polynesian languages and trace their ancestry to Polynesia (see Fig 4.1). The arrival of Polynesian triangle by voyagers from a West Polynesian homeland (the Samoa, Tonga, Futuna, 'Uvea region). However, Tikopia's archaeological history extends back much further than the arrival of Polynesians, to the initial Austronesian expansion into Remote Oceania. These first people to settle Tikopia arrived roughly 2800 years ago (Kirch and Yen 1982; Kirch and Swift in submission).

Although it is a small island (4.6 km<sup>2</sup>), Tikopia has proven capable of supporting a relatively dense population. The island owes its stability in part to its geologically young age (ca. 90,000 years) and nutrient-rich soils. A relatively large portion of the island is made up of cultivable land. Its success is also due to the generations of human inhabitants who actively managed Tikopia's landscape and developed a complex, multi-story sustainable agricultural system which combines arboriculture with field cropping. However, population control also played a key role in maintaining the island's long-term resilience: the Tikopia ensured their population never exceeded the island's carrying capacity through several growth-limiting mechanisms, which included infanticide, enforced celibacy, prevention of contraception, abortion, sea voyaging, and warfare (Borrie et al. 1957; Firth 1936:173-4).

The concept of islands as "model systems" for global ecosystem functioning has grown in popularity as humans are increasingly recognized as a force for planet-wide change (e.g., Davies et al. 2016; Vitousek 2002, 2004; Warren et al. 2015). The Polynesian islands in particular are suited to such analyses due to their relatively short chronologies, agriculturallybased economies, and common origin from a West Polynesian homeland (see Kirch 2007a). As a Polynesian Outlier, Tikopia's long archaeological history and blend of Melanesian and Polynesian cultural influences presents a contrastive study to islands within the Polynesian triangle. Tikopia has figured into discussions of long-term socio-ecosystem dynamics in Polynesia through comparison with other islands including Mangaia, Hawaii, and Mangareva (Kirch 1983; 1997b; 2007a).

The following research presents bone collagen stable isotope data from Pacific rat (*Rattus exulans*) remains recovered from archaeological sites on Tikopia to explore long-term island socio-ecosystem dynamics through the lens of a small, commensal omnivore. Stable isotope analysis of small-bodied mammals has proven effective in paleontological contexts for providing localized, high-resolution paleoenvironmental data (e.g., Gehler et al. 2012; Grimes et al. 2008; Hynek et al. 2012). In archaeological contexts, the diet of commensal rat species is also sensitive to changes in human behavior. These dietary changes in rat populations through time offer insights into shifts in local ecosystems as well as human activities, including changing site use and subsistence practices (see Guiry and Gaulton 2015). On Tikopia, these data can be used to

explore pre-Polynesian and Polynesian landscape transformations and subsistence economies in what has been called a "model of the sustainable microcosm" (Kirch 1997b:35).



Figure 4.1: Map of Tikopia with study sites labeled, and inset map showing Tikopia and the divide between near and remote Oceania.

#### Long-Term Human Ecodynamics on Tikopia

Kirch and Yen's (1982) pioneering excavations on Tikopia established a nearly threethousand year sequence of occupation, with evidence for an ever-evolving island ecosystem in the hands of purposeful land managers. The Tikopian archaeological sequence can be divided into four main cultural phases: the Kiki, Sinapupu, Tuakamali, and Historic Phases. Recently, Kirch and Swift (in submission) revisited the chronological sequence established by Kirch and Yen's original radiocarbon dates (Kirch and Yen 1982:Table 50), and augmented these with 13 new AMS radiocarbon dates obtained on samples from the original Kirch and Yen excavations. The results of Bayesian analysis incorporating both original and new AMS radiocarbon dates have further refined the chronological sequence for Tikopia (see Table 4.1).

Tikopian cultural phases were established by Kirch and Yen (1982) based on changes in ceramic and other artifact assemblages, however these phases are also accompanied by significant transitions in island geomorphology, the faunal record, and other artifact assemblages. The beginning of the Kiki Phase, marking the earliest arrival of people on Tikopia, occurred around 2745 BP (795 cal BC) based on new AMS dates from the TK-4 and TK-36 sites (Kirch and Swift, in submission). Kiki Phase ceramics were largely locally-made, sand-tempered

earthenware, some of which were decorated with distinctly Lapita-style dentate-stamped motifs (see Dickinson 1998; Kirch 1997b; Mead et al. 1975). The transition to the Sinapupu Phase (160 BC to AD 1110) is distinguished by cessation of local pottery manufacture in favor of a smaller quantity of imported, volcanic-sand tempered and well-fired ceramics. These ceramics are not of the Lapita style, but rather bear incised and appliquéd designs in the style of the later Mangaasi ceramic series identified in Vanuatu (see Bedford 2006; Garanger 1972). The final phase prior to European contact, the Tuakamali Phase (AD 1110-1780), is characterized by the cessation of ceramic use altogether as well as the appearance of distinctly Polynesian-style material culture, including pearl shell trolling lures, basalt adzes, and bone and shell ornaments. Kirch and Yen (1982) interpret the beginning of the Tuakamali phase as marking the arrival of voyagers from the West Polynesian homeland, the direct ancestors of Tikopia's contemporary social groups. Tikopian oral traditions (Firth 1961) likewise trace the origins of the island's several lineages to specific islands in West Polynesia.

 Table 4.1: Revised chronology and distinguishing features of Tikopia's prehistoric cultural phases.

Phase	Duration	Key Features
Kiki	800 to 160 BC	Locally-produced, Lapita-style ceramics; heavy exploitation of
		marine resources, avifauna, fruit bats; slash-and-burn agriculture
Sinapupu	160 BC to AD 1110	Imported Mangaasi-style ceramics; population growth;
		intensification of agricultural production and pig husbandry; fire
		use declines with the beginning of tree cropping
Tuakamali	AD 1110 to 1780	Arrival of Polynesian people; formation of Ravenga Tombolo and
		Lake Te Roto; cessation of agricultural fire use; development of
		multi-story arboricultural cropping system; late extirpation of pig

Tikopian geomorphology and land use underwent significant changes during this roughly 2500 year span, due to a combination of natural processes (such as sea level change) combined with active landscape management practices by the Tikopia. Mid-Holocene sea level declines led to shoreline progradation, while clearance and land use on hillsides accelerated erosion and generated an influx of nutrient-rich sediment to lowland areas. These processes had a twofold effect on Tikopian subsistence through time: an increase in the total cultivable land area, and restriction in the total area of accessible reef flat. Early Tikopian agriculture focused on shifting cultivation and intensive use of fire, indicated by extensive charcoal in the deeper strata at Rakisu (Kirch and Yen 1982:154). By the late Sinapupu and Tuakamali Phases, fire use had been abandoned in favor of developing a complex arboricultural system of Polynesian socio-economic crops which mimicked natural forest cover. This new arboricultural system restabilized hillside slopes, slowing the deposition of new sediment onto coastal flats. However, another significant geomorphologic change occurred during the Tuakamali Phase: the formation of the Ravenga tombolo on the eastern side of the island, which created a large lake (Te Roto) in the center of the island. The tombolo developed naturally over the course of several centuries, though the Tikopia actively prevent its erosion by constructing sea walls and infilling the lake margin (Kirch and Yen 1982:11).

Faunal evidence corroborates these changes in Tikopian subsistence and island ecology. Early Kiki Phase assemblages are typical of colonization-phase sites throughout Remote Oceania, with an early reliance on wild resources including marine fauna, native avifauna, and fruit bats. Evidence for depression in these resources through time exists in the form of decreasing mollusk size and abundance in post-Kiki Phase deposits as well as the extinction of at least six avifaunal species (Kirch 1983: Table 47; Steadman et al. 1990). As these resources diminished through the Kiki Phase, the Tikopia began to experiment with agricultural systems, and increasingly relied on cultivated resources over wild hunting and foraging. By the Sinapupu Phase, an agricultural economy focused on shifting cultivation and animal husbandry (pigs and chickens) had developed. Although the Tikopia continued to exploit marine resources from the surrounding reef, progradation of the island's beach ridges had eliminated nearly half of the formerly available reef area (Kirch 1983; Kirch and Yen 1982:107).

Considerable changes occurred in the Tikopian subsistence system during the Tuakamali Phase. The beginning of this phase saw the new arrival of voyagers from the West Polynesian homeland. The use of fire in land clearance ceased almost entirely while the complex arboricultural system observed ethnohistorically developed. Multistory orchard gardening of Polynesian tree crops (e.g., coconut, breadfruit, Polynesian chestnut) and intensively mulched field crops (yams, taro) formed the backbone of a largely sustainable agroeconomy (Kirch and Yen 1982:38-39). This new agricultural system mimicked the structure of the natural forests it replaced, providing new habitat for nesting avifauna (Steadman et al. 1990). The formation of the Te Roto Lake created new opportunities for cultivating communities of lake fish (milkfish, mackerel) and hosting duck (*toroa*) populations on the lake surface. Pig husbandry continued in the Tuakamali Phase, however at some point prior to Western contact, the decision was made to eliminate pigs from the island. This decision was purportedly due to pigs becoming an overwhelming pest to orchard crops (Firth 1959:34). Kirch (2000b) argued this decision ultimately reflects trophic competition between pigs and humans over limited resources, a phenomenon seen on several other small islands.

Unlike islands within the Polynesian triangle, Tikopia lies to the west of the "exulansonly" line (Roberts 1991) of rat distribution, which separates most islands in Remote Oceania by virtue of having only a single rat species (R. exulans) prior to European contact. A second rat species, the large spiny rat (Rattus praetor) was introduced prehistorically to Tikopia. It was not found in the earliest archaeological deposits of the island (site TK-4), but seems to appear several hundred years after initial colonization, in the Late Kiki Phase. Flannery et al. (1988) suggested that *R. praetor* arrived on Tikopia through continued contact with other islands in the Solomon archipelago. Relatively little is known about the large spiny rat, which is likely native to New Guinea and is presently found there and on several other islands in Near Oceania. However, it is now extinct on Tikopia (White et al. 2000). Unlike the Pacific rat, which creates nests and husking stations, the spiny rat is a strong burrower and—while still an opportunistic feeder appears to have been less commensally associated with people (Taylor et al. 1982). Spiny rat remains are easily distinguishable from the Pacific rat due to their larger size, and indeed R. exulans remains greatly outnumber those of R. praetor in all sites from Kirch and Yen's excavations (1982:280). Although the two rat species may have been sympatric, it does appear that the Pacific rat was more successful in colonizing the Tikopian household as an ecological niche.

### **Materials and Methods**

Several characteristics of the Pacific rat position it as an ideal proxy species for local cultural and environmental change on Tikopia. The rat is widely distributed across the Pacific, and was apparently transported to Polynesian islands with some of the earliest human voyagers to settle new islands. Because of this, Pacific rats have also been used as a proxy for human colonization and migration in Polynesia (the 'commensal model' for the settlement of the Pacific; Matisoo-Smith 2009) and the chronology of initial human arrival to new islands (Anderson 2004; Kirch 2011b; Wilmshurst et al. 2008). Pacific rats are omnivores with low dietary selectivity and a limited home range (26-89 m in diameter; Atkinson and Moller 1990; Bramley 2014), which suggests their diet can reflect the full range of resources available at a particular site. Although modern Pacific rat diet is predominantly vegetal, this species has been observed predating a large range of animal species for protein, including arthropods, turtle hatchlings and eggs, small birds and bird eggs, and even other rats (Campbell et al. 1984; Norman 1975; Spenneman 1997). This wide range of potential dietary inputs, as well as the Pacific rat's close association with human activities and food sources, suggest that rat diet will change through time along with variations in human subsistence, resource use, and landscape management practices.

# Study Sites

Pacific rat remains were analyzed from three separate sites on Tikopia (TK-4, TK-35, and TK-36; see Fig 1) excavated by Kirch and Yen in 1977-78, which contain deposits from all three prehistoric cultural phases: Kiki, Sinapupu, and Tuakamali. These sites are all situated on the Rotoaia Paleodune, an arc-shaped series of former beach ridges in the center of the western lowland region of the island (Faea District). Although the dune is presently 200+ meters inland from the modern coastline, these sites were located near the original shoreline at the time of their prehistoric occupation. Today, the Rotaia region is intensively cultivated with nutrient-rich organic sediment that accumulated during centuries of prehistoric occupation and midden deposition in the area.

Site TK-4 is a large  $(4500+ m^2)$  settlement site containing the earliest evidence for human occupation on Tikopia. Pacific rat remains were analyzed from the intact cultural Layer II, which represents the Kiki Phase. This layer can be further stratigraphically subdivided into earlier (C2) and later (C1) subphases to provide further chronological resolution within this 650-year time period (see Kirch and Yen 1982:99). Sites TK-35 and TK-36 are both included within the Sinapupu beach ridge sequence, which contains rich deposits from the Kiki, Sinapupu, and Tuakamali phases (see Kirch and Yen 1982:89, Fig 37).

#### Stable Isotope Analysis

Carbon and nitrogen stable isotope ratios have proven useful in reconstructing the diet of archaeologically recovered human and faunal remains (see Katzenberg 2008; Sealy 2001). Bone collagen is synthesized from dietary amino acids and thus is weighted towards the protein component of diet (Ambrose and Norr 1993; Tieszen and Fagre 1993). Stable isotope abundances are expressed in delta ( $\delta$ ) notation as ratios between two isotopes (e.g.,  ${}^{13}C/{}^{12}C$ ) relative to an established standard, and expressed in parts per thousand (‰) where:  $\delta$  =

[(Rsample/Rstandard) - 1] \* 1000. The established standards for C and N stable isotope analysis are calibrated to Pee Dee Belemnite (PDB) and atmospheric nitrogen (Ambient Inhalable Reservoir, or AIR), respectively.

Carbon stable isotope ratios are primarily useful in distinguishing between  $C_3$  and  $C_4$  plant consumption, as well as between terrestrial  $C_3$  and marine resources. Differential discrimination of <sup>13</sup>C between Calvin cycle ( $C_3$ ), Hatch-Slack cycle ( $C_4$ ), and Crassulacean Acid Metabolism (CAM) photosynthetic pathways result in distinct ranges of  $\delta^{13}C$  values.  $\delta^{13}C$  values of  $C_3$  plants tend to fall between -34 to -21‰, and  $C_4$  plants between -15 to -8‰. CAM plants tend to have  $\delta^{13}C$  values falling between the ranges of  $C_3$  and  $C_4$  plants, though may also resemble  $C_4$  plants. Carbon intake in marine systems comes from a mix of sources including terrestrial detritus and dissolved bicarbonate, and in general produce  $\delta^{13}C$  values resembling those of  $C_4$  plants.

Nitrogen stable isotope ratios are primarily useful in determining trophic position, and can also aid in distinguishing terrestrial versus marine resource contributions.  $\delta^{15}N$  values can be influenced by numerous factors, including environmental conditions, physiology, and diet; however in general, an approximate 3.5‰ increase in  $\delta^{15}N$  values accompanies every "step" up the trophic ladder (Bocherens and Drucker 2003; DeNiro and Epstein 1981; Minagawa and Wada 1984). Because marine systems are complex, containing many levels of producers and consumers, marine resources tend to be more enriched in <sup>15</sup>N than their terrestrial counterparts.

The collections from Kirch and Yen's 1977-78 archaeological excavations are curated at the Bernice Pauahi Bishop Museum; the specimens used in this analysis were acquired through a destructive analysis loan. A total of 87 rat femora were selected for analysis, taking care to select only left or right sides for each excavation level to minimize the possibility of resampling from the same individual. Between levels, femora were compared for potential matching pairs to further reduce the chances of double sampling.

Bone elements were sonicated for four hours in ultrafiltered water, dried, and abraded to remove adhering surface contaminants. Specimens were then crushed to ~1 mm chunks using an agate mortar and pestle. Collagen extraction proceeded following a modified version of Ambrose (1990; and see meta-analyses by Pestle et al. 2014; Sealy et al. 2014). Samples weighing over 20 mg were treated with 0.5 M HCl, and samples under 20 mg with 0.25 M HCl, with fresh acid applied every 24 hours until demineralized. Following demineralization, all samples were rinsed with ultrafiltered water and treated with 0.1 M NaOH for 12 hours. Samples were then rinsed and freeze dried for 48 hours, and weighed into tin caps. Dry samples were analyzed simultaneously for C and N contents (% dry weight) and C and N stable isotope ratios using a CHNOS Elemental Analyzer (vario ISOTOPE cube, Elementar, Hanau, Germany) and IsoPrime 100 mass spectrometer (Isoprime LTd., Cheadle, UK). Analyses were conducted at the Center for Stable Isotope Biogeochemistry at the University of California, Berkeley, with long-term external precision of 0.1‰ (C) and 0.15‰ (N).

The primary means of evaluating bone collagen is the ratio of preserved carbon to nitrogen (C/N Ratio) in the sample. Samples with C/N ratios falling outside the range of 2.9-3.6 may be contaminated and were eliminated from analysis (Ambrose 1990; DeNiro and Weiner 1988; van Klinken 1999). Secondary quality indicators include the quantity of bone collagen, as well as the weight percent carbon and nitrogen preserved in the sample. Modern bone is composed of approximately 22% collagen by weight, and archaeological samples recovered from tropical and subtropical contexts should contain at least 0.5% collagen (Pestle and Colvard

2012). C and N concentrations within samples should approximate that of modern bone collagen: roughly 35% carbon and 11-16% nitrogen by weight (van Klinken 1999).

Out of 87 specimens, 9 samples dissolved during demineralization and could not be analyzed. Of the 78 samples that were analyzed for C and N stable isotope ratios, only 28 produced C/N ratios within the recommended range of 2.9-3.6. In most cases, eliminated samples contained low % wt nitrogen, resulting in C/N ratios well above 3.6. Although the small number of viable samples from the Tikopia assemblage limits interpretation to some degree, these results are not atypical of assemblages recovered from tropical and subtropical environments. A metaanalysis by Pestle and Colvard (2012) determined that samples with at least 0.5% preserved collagen, as well as proportional loss of %C and %N in degraded collagen samples (C/N ratio between 2.9-3.6) in tropical contexts possess sufficiently unaltered collagen to produce accurate  $\delta^{13}$ C and  $\delta^{15}$ N values. Collagen degradation occurred differentially between temporal phases, with 14 of 37 (38%) samples lost in the Kiki Phase, 30 of 35 (86%) lost in the Sinapupu Phase, and 14 of 15 (93%) lost in the Tuakamali Phase. There was no correlation between collagen degradation and sample age (contra Pestle and Colvard 2012). Rather, local site conditions such as soil acidity and water percolation appear to have had the strongest impact on collagen preservation. Laboratory, time, and cost constraints prohibited an analysis of amino acid composition of bone collagen samples, however in future studies this could aid in determining whether preferential amino acid loss occurred during in situ bone collagen degradation (see Tuross 2002).

Radiocarbon dating of select Rattus exulans samples from Tikopia provides further reassurance regarding the accuracy of  $\delta^{13}$ C and  $\delta^{15}$ N values of bone collagen samples which meet minimum quality control requirements. A total of 11 rat bone samples were split, with half analyzed for bone collagen C and N stable isotope ratios by the author and half submitted to the W.M. Keck Carbon Cycle Accelerator Mass Spectrometry Laboratory at the University of California, Irvine for radiocarbon dating and independent C and N stable isotope analysis. Of these 11 samples, four contained sufficient quantities of bone collagen for AMS radiocarbon analysis, and two of these four were large enough for both <sup>14</sup>C and C & N stable isotope analysis (TKR-12, TKR-41). Samples TKR-12 and TKR-41 differed in  $\delta^{13}$ C values between labs by 0.06‰ and 0.12‰ respectively, while  $\delta^{15}$ N values demonstrated a slightly larger variance of 0.5% and 0.01% (respectively). These values are within the range of expected inter-laboratory variation based on differences such as sample preparation, instrumentation, and calibration (see Pestle et al. 2014). AMS dating results aligned well with the expected date ranges for these samples based on their stratigraphic contexts, suggesting that the remaining collagen preserved in these samples did not undergo differential deterioration. Although this sample is admittedly small, it provides reassurance that specimens meeting minimum quality control standards within this assemblage do produce viable bone collagen.

#### **Results**

Table 4.2 presents  $\delta^{13}$ C and  $\delta^{15}$ N values and quality control indicators for the 28 samples with sufficiently preserved bone collagen for further analysis. Kiki Phase rat elements were recovered from site TK-4 (early Kiki) and site TK-36 (early and late Kiki). All Sinapupu Phase rats with well-preserved bone collagen were recovered from site TK-35. The single Tuakamali Phase element was recovered from site TK-36. The loss of 68% of the total assemblage resulted in adequate representation for the Kiki Phase—such that this phase can be divided into early and

Sample			Level/									
ID	Site	Square	Layer	Element	Side	% C	% N	C:N	%Coll	δ13C	δ15N	Phase
TKR-3	TK-4	O18III	II	Femur	L	35.81	12.56	3.33	1.73	-15.3	15.9	Kiki (C2)
TKR-7	TK-4	P18IV	II	Femur	L	40.97	13.93	3.43	4.66	-19.9	11.9	Kiki (C2)
TKR-9	TK-4	P18IV	II	Femur	L	40.78	14.09	3.38	5.42	-18.8	12.6	Kiki (C2)
TKR-10	TK-4	R16II	II	Femur	L	39.22	13.20	3.47	12.79	-15.3	12.3	Kiki (C2)
TKR-11	TK-4	R16II	II	Femur	L	23.17	8.23	3.29	9.39	-14.8	12.0	Kiki (C2)
TKR-12	TK-4	R16II	II	Femur	L	18.11	5.96	3.54	13.57	-17.1	13.4	Kiki (C2)
TKR-14	TK-4	R16II	II	Femur	L	43.51	15.35	3.31	3.40	-17.2	13.2	Kiki (C2)
TKR-15	TK-4	S15II	II	Femur	L	38.05	12.69	3.50	1.12	-14.1	15.1	Kiki (C2)
TKR-18	TK-4	U17II	II	Femur	L	27.56	9.23	3.49	8.95	-18.9	14.3	Kiki (C2)
TKR-19	TK-4	U17II	II	Femur	L	39.36	13.42	3.42	3.16	-17.4	16.1	Kiki (C2)
TKR-20	TK-4	U18IV	II	Femur	L	32.56	10.76	3.53	3.82	-20.2	12.8	Kiki (C2)
TKR-23	TK-4	U18IV	II	Femur	L	37.36	12.79	3.41	3.87	-15.7	12.5	Kiki (C2)
TKR-26	TK-4	W13	II	Femur	L	44.94	15.02	3.49	2.52	-19.4	12.4	Kiki (C2)
TKR-87	TK-4	R16III	II	Femur	L	21.06	7.20	3.48	25.39	-15.0	13.3	Kiki (C2)
TKR-36	TK-35	A3	2	Femur	R	43.76	14.81	3.45	1.16	-20.2	10.9	Sinapupu
TKR-38	TK-35	C2	2	Femur	R	29.86	10.37	3.36	3.36	-18.6	8.1	Sinapupu
TKR-40	TK-35	A2	3	Femur	R	21.76	7.29	3.48	3.88	-19.3	9.2	Sinapupu
TKR-41	TK-35	A2	3	Femur	R	23.82	7.72	3.60	5.84	-19.3	9.8	Sinapupu
TKR-53	TK-36	A4	I (lower)	Femur	R	43.52	15.37	3.30	17.27	-23.0	8.7	Tuakamali
TKR-55	TK-36	A1	II	Femur	L	33.40	11.06	3.52	1.12	-20.4	11.7	Kiki (C1)
TKR-56	TK-36	A1	II	Femur	L	41.44	14.23	3.40	1.21	-18.9	13.9	Kiki (C1)
TKR-58	TK-36	A4	II	Femur	L	31.49	10.37	3.54	2.16	-18.5	12.9	Kiki (C1)
TKR-59	TK-36	A4	II	Femur	L	30.03	9.77	3.59	6.72	-16.3	13.1	Kiki (C1)
TKR-60	TK-36	A4	II	Femur	L	40.65	13.20	3.59	1.28	-17.0	13.0	Kiki (C1)
TKR-61	TK-36	A4	II	Femur	L	43.87	15.00	3.41	1.33	-17.9	11.7	Kiki (C1)
TKR-62	TK-36	A4	II	Femur	L	45.33	14.97	3.53	2.01	-18.9	11.3	Kiki (C1)
TKR-63	TK-36	B1	II	Femur	L	31.89	10.42	3.57	2.36	-15.9	13.0	Kiki (C1)
TKR-68	TK-36	A3	III (lower)	Femur	L	29.00	9.68	3.50	7.19	-17.7	15.0	Kiki (C2)

Table 4.2: Pacific rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values with quality control indicators for the Kiki, Sinapupu, and Tuakamali Phases on Tikopia.

late subphases—a small number of Sinapupu Phase samples, and only a single individual as representative of the Tuakamali Phase. Although the data set is inadequate to fully represent changing rat diet throughout the entire Tikopian sequence, it does provide insight into the Kiki and Sinapupu Phases. Most notably, these data point to a long span of relative stability in rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values, followed by an apparent decline in  $\delta^{15}$ N and more negative  $\delta^{13}$ C values in later prehistory (Figure 4.2).

Mean  $\delta^{13}$ C and  $\delta^{15}$ N values change little between the early and late Kiki Phases, with differences of +0.9‰ and -0.9‰ respectively (see Table 4.3 for descriptive statistics). Although fewer samples were successfully analyzed from the Sinapupu Phase, these individuals have  $\delta^{13}$ C and  $\delta^{15}$ N values that are statistically distinct from Kiki Phase rats, particularly with respect to nitrogen stable isotope ratios, which are depleted in <sup>15</sup>N relative to Kiki Phase rats.  $\delta^{13}$ C values from Sinapupu Phase rats fall in line with Kiki Phase rats, but are restricted to the more negative values within this range (-19.3 ± 0.6‰). The single rat bone element from the Tuakamali Phase, while obviously not constituting a representative sample, suggests a continuation of the trend towards more negative  $\delta^{13}$ C and lower  $\delta^{15}$ N values.

The Kiki Phase deposits contain a record of relatively stable Pacific rat diet for around 650 years following the initial human settlement of Tikopia. Although these data suffer from sample size issues, the available sample of Sinapupu and Tuakamali Phase rats suggests shifts in rat diet occurred between phases. The data for these chronological trends come from three separate sites (TK-4, TK-35, and TK-36), which are less than 400 m distant from each other on the Rotoaia paleodune. Thus, regional variation can be ruled out as the cause for change in rat diet between cultural phases. Rather, rat dietary shifts are presumably related to changing socioecosystem dynamics, particularly with respect to agricultural regimes and subsistence practices.

		δ <sup>13</sup> C				$\delta^{15}N$				
Phase	n	Mean	SD	Min	Max	Mean	SD	Min	Max	
Early Kiki	15	-17.1	2.0	-20.2	-14.1	13.5	1.4	11.9	16.1	
(C2)										
Late Kiki	8	-18.0	1.5	-20.4	-15.9	12.6	0.9	11.3	13.9	
(C1)										
Sinapupu	4	-19.3	0.6	-20.2	-18.6	9.5	1.2	8.1	10.9	
Tuakamali	1	-23.0	n/a	-23.0	-23.0	8.7	n/a	8.7	8.7	

Table 4.3: Descriptive statistics for Pacific rat bone collagen  $\delta^{13}C$  and  $\delta^{15}N$  values by cultural phase on Tikopia.



Figure 4.2: Pacific rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values by cultural phase.

# Socio-Ecosystem Dynamics and Pacific Rat Stable Isotope Ratios

Figure 4.3 plots rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values by cultural phase, as well as estimated values for potential food sources. These estimates were derived from previously published C and N stable isotope ratios from archaeological studies in other islands of the Pacific

(Allen and Craig 2009; Brooke et al. 2010; Kinaston et al. 2014; Richards et al. 2009; Storey et al. 2010). Diet, physiology, and environmental factors can influence diet-to-tissue fractionation. In general, rat bone collagen  $\delta^{13}$ C values have been reported as roughly 3‰ more positive than diet (Lee-Thorp et al. 1989), while  $\delta^{15}$ N values are approximately 3.5‰ higher than diet (Bocherens and Drucker 2003; DeNiro and Epstein 1981; Minagawa and Wada 1984).

Tikopia's rich reef environment and intensive agricultural system are evident in rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values as early as the Kiki Phase. Early Kiki Phase rats appear to have consumed a mix of C3 terrestrial resources as well as reef fish, some offshore fish, and potentially some seabirds for a small number of individuals. In the late Kiki Phase, these high trophic level marine resources such as avifauna and offshore fish are not at all apparent in diet. Late Kiki Phase diet was more focused on terrestrial resources, with small inputs from reef sources. This is coincident with declines in avifaunal remains as well as inshore, benthic, and pelagic fish between the early and late Kiki Phases (Kirch and Yen 1982; tables 40 and 43). By the Sinapupu Phase, rat diet was largely terrestrial C<sub>3</sub> focused. Although there appears to have been greater dietary restriction in the Sinapupu Phase, this may be a result of sampling. Kirch and Yen (1982) note an abrupt transition between the Kiki and Sinapupu phases, in both ceramic assemblages and site activity. Settlement began to expand and shifted seaward as the shoreline prograded. This settlement expansion is suggestive of an increasing population. A significant increase in pig remains indicates intensification of the terrestrial production system during this time, and associated declines in marine faunal abundances further confirm this movement towards greater reliance on terrestrial resources. Pig remains continue to appear in abundance throughout Tikopia's faunal record until the late Tuakamali Phase. Stable isotopic data from the single rat from this phase, with  $\delta^{13}$ C and  $\delta^{15}$ N values indicative of a C<sub>3</sub> terrestrial diet, is consistent with such continued investment in terrestrial agricultural production.

Bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values reflect an averaging out of dietary inputs throughout most of an individual rat's lifetime. However, several non-dietary factors may also influence C and N stable isotope ratios in bone collagen and other tissues, including physiology and local environmental variables such as temperature and annual precipitation (Amundson et al. 2003). In particular, declines in avifaunal populations after initial human arrival and changing fire use in agricultural systems could have significant impacts on baseline N availability and  $\delta^{15}$ N values. These baseline impacts to soil N would have flowed through entire foodwebs, potentially resulting in variable N stable isotope ratios without any direct dietary change. Additionally, the introduction of the spiny rat in the late Kiki Phase may have influenced Pacific rat diet and feeding behavior.

The decline of land and seabird populations—often to the point of species extinctions or extirpations—is a common pattern following initial human arrival on new islands. Hunting, predation by introduced animals (especially rats; see Brooke et al. 2010; Caut et al. 2008; Jones et al. 2008), and habitat destruction all contributed to diminished seabird populations (Athens et al. 2002; Steadman 1995). On Tikopia, avifauna faced similar population declines as well as the extinction or extirpation of 2 land bird and 4 seabird species (Steadman et al. 1990). However, Kirch (2007a) notes that prehistoric human impact to Tikopia's avifaunal communities was far less severe than on other islands such as Mangaia, Mangareva, and the Marquesas. The orchard garden system which developed and intensified through time on Tikopia would have provided ample habitat for nesting seabird communities, particularly in later prehistory.



Figure 4.3: Pacific rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values from Tikopia by phase, with estimated values for potential food sources. Possible dietary sources were estimated from previously published carbon and nitrogen stable isotope ratios from elsewhere in the Pacific (Allen and Craig 2009; Brooke et al. 2010; Kinaston et al. 2014; Richards et al. 2009; Storey et al. 2010).

Nesting seabirds in particular can provide a critical soil nutrient resource in guano, which is high in N content and enriched in <sup>15</sup>N (Anderson and Polis 1999; Briggs et al. 2012; Fukami et al. 2006; Mizota and Naikatini 2007). Szpak et al. (2012) have shown that plants fertilized with seabird guano have significantly higher  $\delta^{15}$ N values than unfertilized plants. This <sup>15</sup>N enrichment was shown to vary based on the quantity of guano inputs, with heavily fertilized plants having the highest  $\delta^{15}$ N values. However, this study also demonstrated that overfertilization with seabird guano can be detrimental, creating high concentrations of ammonium which can inhibit plant growth. Changing abundances of nesting seabird communities on Tikopia over time might have caused soil  $\delta^{15}$ N values—and therefore rat bone collagen  $\delta^{15}$ N values—to fluctuate. Fertilization with other animal dung can likewise lead to <sup>15</sup>N-enriched soils, though to a lesser degree than heavy bird guano inputs. On Tikopia, the presence of pigs until late prehistory may have provided an additional source of fertilizer. It is unclear whether the Tikopia used pig feces as fertilizer in prehistory, however this practice is known to increase  $\delta^{15}$ N values in plants by as much as 5.6‰ (see Szpak et al. 2012; Table 5).

The use of fire for forest clearance and agricultural production can likewise influence baseline  $\delta^{15}N$  values. As Szpak (2014:7) notes, the impact of fire on  $^{15}N$  enrichment is inconsistent across studies, however in general fire use and the accumulation of charred material in soils tends to temporarily increase  $\delta^{15}N$  values by up to +8.6‰. Continuous application of fire to the same region would continue to maintain  $^{15}N$ -enriched soils. However, upon cessation of fire use, soil and foliar  $\delta^{15}N$  values begin to decline and can return to their pre-fire levels in a

matter of a single decade (LeDuc et al. 2013). Fire use for forest clearance and shifting cultivation during the earlier time periods on Tikopia is well-documented (Kirch and Yen 1982:154). Extensive slash-and-burn agricultural practices were established in the Kiki Phase and continued for several centuries. This practice began to diminish through the Sinapupu Phase as the complex arboricultural system used today was in development. In the Tuakamali Phase and at present, fire is rarely used in agricultural practice on the island. These changes in fire use surely impacted  $\delta^{15}$ N values throughout Tikopia's terrestrial ecosystem. However, more research on the complexities of nitrogen stable isotope biogeochemistry in archaeological foodwebs must be conducted to establish the degree to which these practices impacted Tikopia and other tropical island systems (see review by Szpak 2014).

The arrival of the large spiny rat in the late Kiki Phase may also have affected Pacific rat C and N stable isotope ratios. Competition with a larger-bodied, sympatric rat species may have limited Pacific rat population growth as well as changed its diet and feeding behavior (see Grant 1972). Such changes have been observed between the Pacific rat and the Norway rat (Rattus norvegicus) and black rat (Rattus rattus) in other Pacific island regions. In general, the Pacific rat prefers grasslands and open areas with ample ground cover, and in the absence of competition from other species it will expand its distribution into other niches including households and gardens (Dwyer 1978; Taylor 1975). The Pacific rat can exist sympatrically alongside both Norway and black rats, however this tends to result in restricted access to food, lower quality diets, and dwindling Pacific rat populations (Harper and Veitch 2006; Russell and Clout 2005). Greater overall abundances of the Pacific rat in Tikopian archaeological sites throughout prehistory suggest that the Pacific rat had a closer commensal relationship with humans than the spiny rat. Although larger-bodied rats tend to have more competitive success, the faunal record would indicate that the Pacific rat was more successful in establishing human-occupied zones as a primary niche. In this case the Pacific rat's size may have served it well, as this would have made it easier to hide from human site occupants. Further, Russell et al. (2015) have shown that even in sites where the Pacific rat faces competition from a larger rat species (R. rattus), Pacific rat trophic position does not decline and  $\delta^{13}$ C and  $\delta^{15}$ N values from liver and muscle tissue do not change significantly. Thus, it appears unlikely that the presence of the large spiny rat would have had a strong influence on Pacific rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values.

# **Conclusion**

The archaeological record of Polynesian islands represents an accumulation of hundreds of years of experimentation and adaptation to new island landscapes. On Tikopia, multiple lines of evidence point to a record of long-term sustainability on a small, relatively isolated island. Young, nutrient-rich agricultural soils allowed for an intensification of terrestrial production and pig husbandry for centuries on an island that is uncharacteristically small to support such sustained use and high population density (Bay-Petersen 1983; Giovas 2006; Kirch 2007a). However, Tikopia's long-term sustainability came at a cost: active population regulations which ensured that the carrying capacity of the island was never exceeded. Bone collagen stable isotope analysis of the Pacific rat presents a record of 600-plus years of relative stability during the Kiki Phase, almost the same length as entire prehistoric sequences on many islands within the Polynesian triangle. Rat diet on Tikopia changed in later prehistory, showing similar trends to other Polynesian islands with declining  $\delta^{15}$ N values and less negative  $\delta^{13}$ C values in the Sinapupu and Tuakamali Phases. These changes happened gradually, and only after several

hundred years of permanent human occupation. Although Tikopian subsistence changed substantially throughout prehistory, these data indicate that the Tikopia were knowledgeable, active managers of their island landscape and resources for over 2000 years. The intensive, arboriculture-oriented Polynesian Outlier society documented by Firth (1936, 1939) is only the most recent iteration of these conscious management practices.

Stable isotope analysis of small commensal animals such as the Pacific rat can provide a new window into past subsistence practices and island socio-ecosystem dynamics. Current research has begun to highlight the significant impact of biogeochemical processes on interpretations of archaeological stable isotope data. In particular, past human activities provide numerous pathways for significant modifications to island landscapes and baseline <sup>15</sup>N abundance (see review by Szpak 2014). Further difficulty lies in attempting to compare  $\delta^{13}$ C and  $\delta^{15}$ N values across regions and other island systems. Without extensive knowledge of ecosystem processes and location-specific  $\delta^{13}$ C and  $\delta^{15}$ N values of potential food sources for all locations under study, cross-regional comparisons remain problematic. As these processes become better understood, proxy data from Pacific rat stable isotope analysis will continue to develop as a powerful tool towards understanding long-term human-environment interactions and sustainable practices.

### **Chapter 5**

### Conclusions: Mangareva, the Marquesas, and Tikopia in Comparative Perspective

This dissertation has demonstrated variation in Pacific rat  $\delta^{13}$ C and  $\delta^{15}$ N values through time across three separate island socio-ecosystems. By placing these three island case studies in comparative perspective, patterns emerge which provide insight into past human subsistence systems and anthropogenic landscape transformation. Figure 5.1 presents mean Pacific rat bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values through time for all three island case studies. It is tempting to make direct comparisons of C and N stable isotope ratios between these three island groups, however various factors including island latitude, environment, rainfall, foodweb dynamics, and prehistoric human activity can create region-specific variation in  $\delta^{13}$ C and  $\delta^{15}$ N values which inhibit the effectiveness of external absolute value comparisons. What can be more easily compared, however, are the overall trends in Pacific rat stable isotope ratios at each site, and what these tell us about island socio-ecosystem dynamics.

All rat bones analyzed in this dissertation were recovered from archaeological contexts, primarily habitation sites. The exceptions are rats from the Onemea site (Mangareva), which was an intensively used hunting camp in early Mangarevan prehistory, and from the late Hane specimens. The Hane dune site appears to have transitioned from a habitation site to a ceremonial, funerary site towards the end of its occupation. The rats from all three prehistoric phases on Tikopia, as well as the two non-habitation contexts, contain some of the most negative  $\delta^{13}$ C values and fall in the lower range of  $\delta^{15}$ N values, suggesting a more terrestrially-focused diet. The high  $\delta^{15}$ N and less negative  $\delta^{13}$ C values of rats recovered from Mangarevan and Marquesan habitation contexts indicate significant inputs of marine proteins. Rats would not have been able to acquire marine resources on their own, and thus it appears these rats were receiving a large portion of their dietary protein intake in the form of human meal scraps. This further supports the notion that changes to rat diet at habitation sites through time should reflect transformations in human subsistence practices.

Pacific rat  $\delta^{13}$ C values exhibit wide variation, becoming either less negative (Mangareva) or more negative (Tikopia, Hane) through time. In the case of Mangareva, a rich lagoon environment with an exceedingly depauperate terrestrial landscape, people (and rats) maintained a strong reliance on marine resources, even into late prehistory. These results align with previous interpretations by Field et al. (2009) from the Fiji Islands, which suggested that smaller islands with limited agricultural productivity would retain more marine-focused diets into late prehistory compared to larger islands even within the same archipelago. Intensified agricultural production may indeed have been the preferred outcome in such scenarios when possible, according to linguistic evidence and ethnographic accounts which point to the importance ascribed to staple crops in everyday meals relative to other foods (Kirch and Green 2001:144). In contrast, rat bones recovered from the Hane dune site in the much larger Marquesas Archipelago show more negative  $\delta^{13}$ C values over time. This change suggests an increasing reliance on the intensified breadfruit arboriculture and pig husbandry regimes which continued into the historic phase. Despite Tikopia's small size, the island's relatively large zones of arable land and geologically young, nutrient-rich soils supported an intensive multi-story orchard gardening system for several centuries. Tikopia was also able to support pig husbandry for an uncharacteristically long duration compared to other islands of its size.



Figure 5.1: Mean (+1SD) *Rattus exulans* bone collagen  $\delta^{13}$ C and  $\delta^{15}$ N values for the Kiki, Sinapupu, and Tuakamali Phases and for early, middle, and late Mangareva and Marquesas sites, with bracketed ranges for  $\delta^{15}$ N values at each site.

One consistent trend seen across all three island systems is a decline in  $\delta^{15}$ N values through time by as much as 7.5‰ (a value equivalent to roughly 2 trophic levels; Bocherens and Drucker 2003; Minagawa and Wada 1984). Several possible explanations for this widespread <sup>15</sup>N depletion on Polynesian islands have been posited in this dissertation. Potential influencing factors include resource depression, avifaunal extinctions and decreasing seabird guano inputs to terrestrial systems, cessation of fire use as an agricultural strategy, and human dietary change (increasing investment in terrestrial resources coupled with declining high trophic level marine fauna). These factors apparently had varying degrees of impact on each system, though what is most remarkable is the difference in time scales over which <sup>15</sup>N depletion took place. On Mangareva, this process took place over the course of roughly 600 years, from AD 1200-1800 (see Kirch et al. 2015). Similarly, the prehistoric cultural sequence at the Hane dune site spans a little over 600 years, from AD 900 to the 1500s (Conte and Molle 2014). In contrast, an equivalent decline in  $\delta^{15}$ N values on Tikopia took place through the Kiki and Sinapupu Phases, a span of roughly 1900 years. Notably,  $\delta^{15}$ N values remain relatively consistent throughout the Kiki Phase which in itself is approximately 650 years long (see Kirch and Yen 1982 and forthcoming publication by Kirch and Swift).

Resource depression is a well-documented phenomenon in all three island systems after human arrival, whether in the form of avifaunal extinctions (Anderson et al. 1994; Steadman and Justice 1998; Steadman and Kirch 1990; Worthy and Tennyson 2004), soil nutrient depletion (Kirch et al. 2015), or declining marine mollusk size (an indication of pressure on marine resources; Kirch and Yen 1982). It appears likely that resource stress, the one constant process across all three island case studies, is responsible for such dramatic declines in Pacific rat  $\delta^{15}$ N values and changes to island baseline <sup>15</sup>N enrichment. Allen and Craig (2009) found evidence for similar stress on Aitutaki (Cook Islands) in changing stable isotope ratios of domestic pig populations over time: Later pigs had a more restricted diet as well as overall lower bone collagen  $\delta^{15}$ N values. If, as this dissertation and other studies suggest, <sup>15</sup>N depletion does correlate with island resource depression, then the consistent range of  $\delta^{15}$ N values in Pacific rat bone collagen throughout the Kiki Phase on Tikopia speaks to a long history of island socioecosystem sustainability. Further, these results suggest that such mechanisms for long-term sustainability were in place long before the Tuakamali (Polynesian) Phase.

# Future Directions for Stable Isotope Studies of the Pacific Rat

The decline in Pacific rat bone collagen  $\delta^{15}$ N values over time appears to be at least partially related to resource depression, diminishing avifaunal populations, and—as a result island-wide <sup>15</sup>N depletion. However, the degree to which these processes drove changes to baseline N stable isotope ratios is uncertain. The incorporation of other stable isotopic data, such as analyzing archaeologically-recovered specimens of short-lived plant species for C and N stable isotope ratios, could provide a second line of evidence for changing island baseline  $\delta^{15}$ N. As archaeological plant specimens from Polynesian archaeological sites would likely come in the form of charcoal, this research would also require further investigation into the impacts of carbonization on  $\delta^{15}$ N values of Polynesian flora (see Bogaard et al. 2007; DeNiro and Hastorf 1985; Vaiglova et al. 2014).

Modern ecological studies have provided new insights into Pacific rat diet and feeding behavior, however they represent imperfect analogs to prehistoric Pacific rat diet. Modern rats are generally recovered from 'natural' habitats, located away from human households. Research comparing the diets of rats recovered away from villages versus those found in households would provide an indication of how human presence and new food sources impact rat diet. In addition, most ecological studies employ gut content or stable isotope analysis of high-turnover tissue such as hair, blood, muscle, and liver. Future research incorporating these modern tissues with bone collagen would provide long-term resolution of rat diet throughout its entire lifespan, as well as build new connections between archaeological bone collagen and the sample substrates frequently employed in contemporary ecological studies.

Comparative analysis of islands as model systems provides a fruitful avenue for isolating the mechanisms that contribute to long-term island socio-ecosystem sustainability. A number of researchers have contributed to the collection of floral and faunal  $\delta^{13}$ C and  $\delta^{15}$ N values specific to the Oceanic region (e.g., Allen and Craig 2009; Beavan Athfield et al. 2008; Field et al. 2009; Jones and Quinn 2009; Kinaston et al. 2014; Richards et al. 2009; Storey et al. 2010; Yoshinaga et al. 1991). However, region-specific differences in island ecology inhibit direct comparisons of stable isotope ratios from multiple islands. Processes such as fertilization, fire ecology, and species extinctions and extirpations can have significant impacts on  $\delta^{13}$ C and  $\delta^{15}$ N values across both space and time. Such changes to island baseline stable isotope ratios carry implications both for the study of rats as paleoecological proxies and for stable isotopic analyses of archaeological human remains from Pacific contexts. The results of this dissertation suggest that not only is it necessary to establish a highly localized collection of baseline  $\delta^{13}$ C and  $\delta^{15}$ N values for potential food sources, but to provide a temporal calibration for these values as well. Further collection and analysis of floral and faunal specimens tied to particular spatiotemporal contexts will continue to generate the baseline stable isotope data necessary for cross-regional comparisons.

While this research has revealed great potential for the Pacific rat to serve as a proxy for nutrient flows in island socio-ecosystems, disentangling the myriad influences on rat stable isotope ratios requires multiple additional lines of archaeological and paleoecological evidence. Pacific rat dietary changes are tied to changing Polynesian subsistence systems, as rats recovered from household contexts appear to reflect the range of resources available to past human inhabitants. As a short-lived species with a limited home range, rat stable isotope ratios ( $\delta^{15}$ N values in particular) can additionally serve as a barometer for local ecosystem change and resource depression. In many regions of the Pacific, it is difficult or inappropriate to destructively analyze human remains. Further, stable isotope analysis of humans and Pacific rats lend different insights into patterns of prehistoric human food consumption: stable isotope analysis of human populations provides long-term *dietary* data tied to particular *individuals*, while stable isotope analysis of Pacific rat populations provides long-term *subsistence* data tied to particular *sites*. The Pacific rat is commonly recovered in abundance from Polynesian archaeological sites. Stable isotope analysis of this small commensal species provides a new, low-impact method for reconstructing past Polynesian subsistence and paleoenvironments.

# References

## Allen, M.S.

1998 Holocene sea-level change on Aitutaki, Cook Islands: Landscape change and human response. *Journal of Coastal Research* 14(1):10–22.

### Allen, M.S.

2004 Revisiting and revising Marquesan culture history: New archaeological investigations at Anaho Bay, Nuku Hiva Island. *The Journal of the Polynesian Society* 113(2):143-96.

### Allen, M.S. and J.A. Craig

2009 Dynamics of Polynesian subsistence: Insights from archaeofauna and stable isotope studies, Aitutaki, Southern Cook Islands. *Pacific Science* 63:477-506.

### Ambrose, S.H.

1990 Preparation and characterization of bone and tooth collagen for isotope analysis. *Journal of Archaeological Science* 17:431-451.

# Ambrose, S.H. and L. Norr

1993 Experimental evidence for the relationship of the carbon isotope ratios of whole diet and dietary protein to those of bone collagen and carbonate. In *Prehistoric Human Bone: Archaeology at the Molecular Level*, edited by Lambert, J.B. and Grupe, G., pp. 1-38. Springer, Berlin.

Ambrose, S.H., B.M. Butler, R.L. Hunter-Anderson, and H.W. Krueger

1997 Stable isotopic analysis of human diet in the Marianas Archipelago, Western Pacific. *American Journal of Physical Anthropology* 104:343-361.

Amundson, R., A.T. Austin, E.A.G. Schuur, K. Yoo, V. Matzek, C. Kendall, A. Uebersax, D. Brenner, and W.T. Baisden

2003 Global patterns of the isotopic composition of soil and plant nitrogen. *Global Biogeochemical Cycles* 17(1):1031-1041.

### Anderson, A.J.

2004 The age disconformity in AMS radiocarbon results on *Rattus exulans* bone. *New Zealand Journal of Archaeology* 24:135-147.

# Anderson, A., Leach, H., Smith, I. and Walter, R.

1994 Reconsideration of the Marquesan sequence in East Polynesian prehistory with particular reference to Hane (MUH1). *Archaeology in Oceania* 29:29-52.

### Anderson, W.B. and G.A. Polis

1999 Nutrient fluxes from water to land: Seabirds affect plant nutrient status on gulf of California islands. *Oecologia* 118(3):324-332.

Aswani, S., and M.S. Allen

2009 A Marquesan coral reef (French Polynesia) in historical context: An integrated socioecological approach. *Aquatic Conservation: Marine and Freshwater Ecosystesms* 19:614-625.

### Athens, J.S.

1997 Hawaiian native lowland vegetation in prehistory. In *Historical Ecology in the Pacific Islands: Prehistoric Environmental and Landscape Change*, edited by P.V. Kirch and T.L. Hunt, pp. 248-270. Yale University Press, New Haven.

## Athens, J.S., H.D. Tuggle, J.V. Ward, and D.J. Welch

2002 Avifaunal extinctions, vegetation change, and Polynesian impacts in Prehistoric Hawaii. *Archaeology in Oceania* 37(2):57-78.

### Atkinson, I.A.E. and H. Moller

1990 Kiore. In *The Handbook of New Zealand Mammals*, edited by C.M. King, pp. 175-192. Oxford University Press, Auckland.

# Atkinson, I.A.E. and D.R. Towns

2001 Advances in New Zealand mammalogy 1990-2000: Pacific rat. *Journal of the Royal Society of New Zealand* 31(1):99-109.

# Baer, A., O. Chadwick, and P.V. Kirch

2015 Soil nutrients and intensive dryland agricultural production in Kaupō, Maui, Hawaiian Islands. *Journal of Archaeological Science: Reports* 3:429-436.

# Bahn, P.G. and J.R. Flenley

1992 Easter Island, Earth Island. Thames & Hudson, London.

# Barber, I.G.

2000 Wet or dry? An evaluation of extensive archaeological ditch systems from far northern New Zealand. In *Proceedings of the Fifth International Conference on Easter Island and the Pacific*, edited by C.M. Stevenson, G. Lee, and F.J. Morin, pp. 41-50. Easter Island Foundation, California.

# Barton, C.M., J. Bernabeu, J.E. Aura, O. Garcia, S. Schmich, and L. Molina 2004 Long-term socioecology and contingent landscapes. *Journal of Archaeological Method and Theory* 11(30):253-295.

#### Bay-Petersen, J.

1983 Competition for resources: The role of pig and dog in the Polynesian agricultural economy. *Journal de la Société des Océanistes* 77(29):121-129.

# Beavan Athfield, N., R.C. Green, J. Craig, B. McFadgen, and S. Bickler

2008 Influence of marine sources on <sup>14</sup>C ages: Isotopic data from Watom Island, Papua New Guinea inhumations and pig teeth in light of new dietary standards. *Journal of the Royal Society of New Zealand* 38(1):1-23.

# Bedford, S.

2006 *Pieces of the Vanuatu Puzzle: Archaeology of the North, South, and Centre.* ANU Press, Canberra.

# Bender, M.B.

1968 Mass spectrometric studies of carbon 13 variations in corn and other grasses. *Radiocarbon* 10(2):468-473.

# Bender, M.B.

1971 Variations in the <sup>13</sup>C/<sup>12</sup>C ratios of plants in relation to the pathway of photosynthetic carbon dioxide fixation. *Phytochemistry* 10(6):1239-44.

# Birch, S.E.P.

2013 Stable isotopes in zooarchaeology: An introduction. *Archaeological and Anthropological Sciences* 5(2):81-83.

# Bocherens, H. and D. Drucker

2003 Trophic level isotopic enrichment of carbon and nitrogen in bone collagen: Case studies from recent and ancient terrestrial ecosystems. *International Journal of Osteoarchaeology* 13:46-53.

# Bogaard, A., T.H.E. Heaton, P. Poulton, and I. Merbach

2007 The impact of manuring on nitrogen isotope ratios in cereals: Archaeological implications for reconstruction of diet and crop management practices. *Journal of Archaeological Science* 34(3):335-343.

# Borrie, W.D., R. Firth, and J. Spillius

1957 The population of Tikopia, 1929 and 1952. *Population Studies* 10(3):229-52.

# Bramley, G.N.

- 2014 Home ranges and interactions of kiore (*Rattus exulans*) and Norway rats (*R. norvegicus*) on Kapiti Island, New Zealand. New Zealand Journal of Ecology 38(2):328-334.
- Briggs, A.A., H.S. Young, D.J. McCauley, S.A. Hathaway, R. Dirzo, and R.N. Fisher
  2012 Effects of spatial subsidies and habitat structure on the foraging ecology and size of geckos. *PLoS ONE* 7(8): e41364. doi:10.1371/journal.pone.0041364.
- Brooke, M. de L., T.C. O'Connell, D. Wingate, J. Madeiros, G.M. Hilton, and N. Ratcliffe
  2010 Potential for rat predation to cause decline of the globally threatened Henderson petrel *Pterodroma atrata*: Evidence from the field, stable isotopes, and population modelling. *Endangered Species Research* 11:47-59.

Brown, A.A. and T. Thomas

2015 The first New Zealanders? An alternative interpretation of stable isotope data from Wairau Bar, New Zealand. *PLoS ONE* 10(10): e0135214. doi:10.1371/journal.pone.0135214.

#### Brown, T.A., D.E. Nelson, J.S. Vogel, and J.R. Southon

1988 Improved collagen extraction by modified Longin method. *Radiocarbon* 30(2):171-177.

# Bunn, T.J. and J.L. Craig

1989 Population cycles of *Rattus exulans*: Population changes, diet, and food availability. *New Zealand Journal of Zoology* 16, 409-418.

# Burley, D., M.I. Weisler, and J.-X. Zhao

2012 High precision U/Th dating of first Polynesian settlement. *PLos ONE* 7(11): e48769. doi:10.1371/journal.pone.0048769.

# Burney, D.A.

1997 Tropical islands as paleoecological laboratories: Gauging the consequences of human arrival. *Human Ecology* 25(3):437-57.

#### Burney, D.A., H.F. James, and L.P. Burney

2001 Fossil evidence for a diverse biota from Kauai and its transformation since human arrival. *Ecological Monographs* 71(4):615-641.

# Campbell, D.J., H. Moller, G.W. Ramsay, and J.C. Watt

1984 Observations on foods of kiore (*Rattus exulans*) found in husking stations on northern offshore islands of New Zealand. *New Zealand Journal of Ecology* 7:131-8.

### Caut, S., E. Angulo, and F. Courchamp

2008 Dietary shift of an invasive predator: Rats, seabirds, and sea turtles. *Journal of Applied Ecology* 45(2):428-437.

Chukhrov, F.V., L.P. Ermilova, V.S. Chuikov, and L.P. Noslik1980 The isotopic composition of plant sulfur. *Organic Chemistry* 2:69-75.

### Clark, G.J.

1997a Osteology of the Kuri Maori: The prehistoric dog of New Zealand. *Journal of Archaeological Science* 24:113-126.

# Clark, G.J.

1997b Maori subsistence change: Zooarchaeological evidence from the prehistoric dog of New Zealand. *Asian Perspectives* 36(2):200-219.

Commendador, A.S., J.V. Dudgeon, B.P. Finney, B.T. Fuller, and K.S. Esh

2013 A stable isotope ( $\delta^{13}$ C and  $\delta^{15}$ N) perspective on human diet on Rapa Nui (Easter Island) ca. AD 1400-1900. *American Journal of Physical Anthropology* 152:173-185.

# Conte, E. and P.V. Kirch (eds.)

2004 Archaeological Investigations in the Mangareva Islands, French Polynesia. Archaeological Research Facility Contribution 62, Berkeley.

# Conte, E. and G. Molle

2014 Reinvestigating a key site for Polynesian prehistory: New results from the Hane dune site, Ua Huka (Marquesas). *Archaeology in Oceania* 49:121-36.

# Cooke, C.M. Jr.

1935 Report of C. Montague Cooke, Jr. malacologist and leader of the 1934 Mangarevan Expedition. In *Report of the Director for 1934*. Bernice P. Bishop Museum Bulletin 133, Honolulu.

### Coutts, P. and M. Jurisich

1973 Canine passengers in Maori canoes. World Archaeology 5:72-85.

### Craig, J.A.

2009 Stable isotope analysis of prehistoric human and commensal diet on Aitutaki, Southern Cook Islands. Ph.D. Dissertation, Department of Anthropology, University of Auckland.

### Crosby, A.

1986 *Ecological Imperialism: The Biological Expansion of Europe, A.D. 900-1900.* Cambridge University Press, Cambridge.

#### Crutzen, P.J.

2002 Geology of mankind. *Nature* 415:23.

### Davidson, J.M., K. Fraser, B.F. Leach, and Y.H. Sinoto

1999 Prehistoric fishing at Hane, Ua Huka, Marquesas Islands, French Polynesia. *New Zealand Journal of Archaeology* 21:25-28.

Davies, N., D. Field, D. Gavaghan, S.J. Holbrook, S. Planes, M. Troyer, M. Bonsall, J. Claudet, G. Roderick, R.J. Schmitt, L.A. Zettler, V. Berteaux, H.C. Bossin, C. Cabasse, A. Collin, J. Deck, T. Dell, J. Dunne, R. Gates, M. Harfoot, J.L. Hench, M. Hopuare, P. Kirch, G. Kotoulas, A. Kosenkov, A. Kusenko, J.J. Leichter, H. Lenihan, A. Magoulas, N. Martinez, C. Meyer, B. Stoll, B. Swalla, D.M. Tartakovsky, H.T. Murphy, S. Turyshev, F. Valdvinos, R. Williams, S. Wood and IDEA Consortium

2016 Simulating social-ecological systems: The Island Digital Ecosystem Avatars (IDEA) consortium. *GigaScience* 5(14) doi: 10.1186/s13742-016-0118-5.

### Dening, G.

1974 *The Marquesan Journal of Edward Robarts, 1797-1824.* University of Hawaii Press, Honolulu.

# DeNiro, M.J.

1985 Postmortem preservation and alteration of *in vivo* bone collagen isotope ratios in relation to paleodietary reconstruction. *Nature* 317:806-809.

# DeNiro, M.J. and S. Epstein

1978 Influence of diet on the distribution of carbon isotopes in animals. *Geochimica et Cosmochimica Acta* 42:495-506.

# DeNiro, M.J. and S. Epstein

1981 Influence of diet on the distribution of nitrogen isotopes in animals. *Geochimica et Cosmochimica Acta* 45:341-351.

# DeNiro, M.J. and C. Hastorf

1985 Alteration of <sup>15</sup>N/<sup>14</sup>N and <sup>13</sup>C/<sup>12</sup>C ratios of plant matter during the initial stages of diagenesis: Studies using archaeological specimens from Peru. *Geochimica et Cosmochimica Acta* 49:97-115.

# DeNiro, M.J. and S. Weiner

1988 Chemical, enzymatic and spectroscopic characterization of "collagen" and other organic fractions from prehistoric bones. *Geochimica et Cosmochimica Acta* 52:2197-2206.

# DeNiro, M.J., M.J. Schoeninger, and C.A. Hastorf

1985 Effect of heating on the stable carbon and nitrogen isotope ratios of bone collagen. *Journal of Archaeological Science* 12:1-7.

# Diamond, J.

2005 Collapse: How Societies Choose to Fail or Succeed. Viking Penguin, New York.

# Dickinson, W.R.

1998 Prehistoric temper provinces of prehistoric pottery in Oceania. *Records of the Australian Museum* 50:263-76.

## Dixon, B., D. Soldo, and C.C. Christensen

- 1997 Radiocarbon dating land snails and Polynesian land use on the island of Kaua'i, Hawai'i. *Hawaiian Archaeology* 6:52–62.
- Duron, Q., E. Bourguet, I.O. Garcia, S. Scussel, M. Boissenin, F. Breschia, and E. Vidal 2015 Complex interactions between long standing introduced rats and native plants in a New Caledonian rainforest. Poster for the 27<sup>th</sup> International Congress for Conservation Biology, Montpelier.

# Dwyer, P.D.

1978 A study of *Rattus exulans* (Peale) (Rodentia: Muridae) in the New Guinea Highlands. *Australian Wildlife Research* 5:221-48.

Dye, T.S.

1990 The causes and consequences of a decline in the prehistoric Marquesan fishing industry. In *Pacific Production Systems: Approaches to Economic Prehistory*, edited by D.E. Yen and J.M.J. Mummery, pp. 70-84. Australian National University, Canberra.

# Dye, T.S.

1996 Assemblage definition, analytic methods, and sources of variability in the interpretation of Marquesan subsistence change. *Asian Perspectives* 35:73-88.

### Dye, T.S.

2014 Wealth in old Hawaii: Good-year economics and the rise of pristine states. *Archaeology in Oceania* 49(2):59-85.

### Field, J.S., E.E. Cochrane, and D.M. Greenlee

2009 Dietary change in Fijian prehistory: Isotopic analyses of human and animal skeletal material. *Journal of Archaeological Science* 36:1547-1556.

# Firth, R.

1936 We the Tikopia: A Sociological Study of Kinship in Primitive Polynesia. Allen and Unwin, London.

# Firth, R.

1939 Primitive Polynesian Economy. Routledge & Sons Ltd., London.

# Firth, R.

1959 Social Change in Tikopia. Macmillan Co., New York.

### Firth, R.

1961 *History and Traditions of Tikopia*. Polynesian Society Memoir 33. The Polynesian Society, Wellington.

#### Flannery, T.F., P.V. Kirch, J. Specht, and M. Spriggs

1988 Holocene mammal faunas from archaeological sites in island Melanesia. *Archaeology in Oceania* 23(3):89-94.

# Fukami, T., D.A. Wardle, P.J. Bellingham, C.P. Mulder, D.R. Towns, G.W. Yeates, K.I. Bonner, M.S. Durrett, M.N. Grant-Hoffman, and W.M. Williamson

2006 Above- and below-ground impacts of introduced predators in seabird-dominated island ecosystems. *Ecology Letters* 9:1299-1307.

#### Garanger, J.

1972 *Archéologie des Nouvelles-Hébrides*. Publications de la Société des Océanistes 30. Paris.

Gehler, A., T. Tütken, and A. Pack
2012 Oxygen and carbon isotope variations in a modern rodent community – implications for paleoenvironmental reconstructions. *PLoS ONE* 7(11): e49531. doi:10.1371/journal.pone.0049531.

# Gill, W.W.

1894 *Life in the Southern Isles; Or, Scenes and Incidents in the South Pacific and New Guinea.* The Religious Tract Society, London.

# Giovas, C.M.

2006 No pig atoll: Island biogeography and the extirpation of a Polynesian domesticate. *Asian Perspectives* 45(1):69-95.

# Grant, P.R.

1972 Interspecific competition among rodents. *Annual Review of Ecology and Systematics* 3:79-106.

# Green, R.C.

1991 Near and remote Oceania – disestablishing "Melanesia" in culture history. In *Man* and a Half: Essays in Pacific Anthropology and Ethnobiology in Honour of Ralph Bulmer, edited by A. Pawley, pp. 491-502. The Polynesian Society, Auckland.

# Green, R.C. and M.I. Weisler

2000 Archaeology: Interpretations using New Data and 40 Year Old Excavations to Establish a Sequence from 1200 to 1900 AD. Studies in Prehistoric Anthropology 19, University of Otago, Dunedin.

## Green, R.C. and M.I. Weisler

2004 Prehistoric introduction and extinction of animals in Mangareva, Southeast Polynesia. *Archaeology in Oceania* 39:34-42.

## Grimes, S.T., M.E. Collinson, J.J. Hooker, and D.P. Mattey

2008 Is small beautiful? A review of the advantages and limitations of using small mammal teeth and the direct laser fluorination analysis technique in the isotope reconstruction of past continental climate change. *Paleogeography, Paleoclimatology, Paleoecology* 266:39-50.

# Guiry, E.J.

2012 Dogs as analogs in stable isotope-based human paleodietary reconstructions: A review and considerations for future use. *Journal of Archaeological Method and Theory* 19:351-376.

# Guiry, E.J. and B.C. Gaulton

2015 Inferring human behaviors from isotopic analyses of rat diet: A critical review and historical application. *Journal of Archaeological Method and Theory* doi: 10.1007/s 10816-015-9248-9.

#### Handy, E.S.C.

1923 *The Native Culture in the Marquesas*. Bernice P. Bishop Museum Bulletin 9, Honolulu.

# Harper, G.A. and N. Bunbury

2015 Invasive rats on tropical islands: Their population biology and impacts on native species. *Global Ecology and Conservation* 3:607-627.

#### Harper, G. and D. Veitch

2006 Population ecology of Norway rats (*Rattus norvegicus*) and interference competition with Pacific rats (*R. exulans*) on Raoul Island, New Zealand. *Wildlife Research* 33(7):539-48.

# Hiroa, T.R.

1938 *Ethnology and Archaeology of Mangareva*. Bernice P. Bishop Museum Bulletin 157, Honolulu.

# Hobson, K.A., M.C. Drever, and G.W. Kaiser

1999 Norway rats as predators of burrow-nesting seabirds: Insights from stable isotope analyses. *The Journal of Wildlife Management* 63(1):14-25.

# Hoefs, J.

1987 Stable Isotope Geochemistry. Springer-Verlag, New York.

#### Holdaway, R.N.

1989 New Zealand's pre-human avifauna and its vulnerability. *New Zealand Journal of Ecology* 12:11-25.

#### Hunt, T.L.

2007 Rethinking Easter Island's ecological catastrophe. *Journal of Archaeological Science* 34:485-502.

## Hunt, T.L. and C. Lipo

2007 Chronology, deforestation, and "collapse": Evidence vs. faith in Rapa Nui Prehistory. *Rapa Nui Journal* 21(2), 85-97.

## Hunt, T.L. and C. Lipo

- 2010 Ecological catastrophe, collapse, and the myth of "ecocide" on Rapa Nui. In *Questioning Collapse*, edited by P.A. McAnany and N. Yoffee, pp. 21-44. Cambridge University Press, Cambridge.
- Hynek, S.A., B. Passey, J.L. Prado, F.H. Brown, T.E. Cerling, and J. Quade
   2012 Small mammal carbon isotope ecology across the Miocene-Pliocene boundary, northwestern Argentina. *Earth and Planetary Science Letters* 321-322:177-188.

James, H.F.

1995 Prehistoric extinctions and ecological changes on Oceanic Islands. In *Islands: Biological Diversity and Ecosystem Function*, edited by P.M. Vitousek, L.L. Loope, and H. Adsersen, pp. 88-102. Springer, Heidelberg.

Jones, H.P., B.R. Tershy, E.S. Zavaleta, D.A. Croll, B.S. Keitt, M.A. Finkelstein, and G.R. Howald

2008 Severity of the effects of invasive rats on seabirds: A global review. *Conservation Biology* 22(1):16-26.

Jones, S. and R.L. Quinn

2009 Prehistoric Fijian diet and subsistence: Integration of faunal, ethnographic, and stable isotopic evidence from the Lau Island Group. *Journal of Archaeological Science* 36:2742-2754.

Kahn, J.G., E. Dotte-Sarout, G. Molle, and E. Conte

2015 Mid- to Late Prehistoric Landscape Change, Settlement Histories, and Agricultural Practices on Maupiti, Society Islands (Central Eastern Polynesia). *The Journal of Island and Coastal Archaeology* 10(3):363-91.

Kahn, J.G., C. Nickelsen, J. Stevenson, N. Porch, E. Dotte-Sarout, C.C. Christensen, L. May, J.S. Athens, and P.V. Kirch

2014 Mid- to late Holocene landscape change and anthropogenic transformations on Mo'orea, Society Islands: A multi-proxy approach *The Holocene* 25:333-47.

Katzenberg, M.A.

2008 Stable isotope analysis: A tool for studying past diet, demography, and life history. In *Biological Anthropology of the Human Skeleton*, edited by M.A. Katzenberg and S. Saunders, pp. 413-41. Wiley-Liss, Hoboken.

Kinaston, R.L., R.K. Walter, C. Jacomb, E. Brooks, N. Tayles, S.E. Halcrow, C. Stirling, M. Reid, A.R. Gray, J. Spinks, B. Shaw, R. Fyfe, and H.R. Buckley

2013 The first New Zealanders: Patterns of diet and mobility revealed through isotope analysis. *PLoS ONE* 8(5): e64580. doi:10.1371/journal.pone.0064580.

Kinaston, R.L., H.R. Buckley, and A. Gray

- 2013 Diet and social status on Taumako, a Polynesian Outlier in the Southeastern Solomon Islands. *American Journal of Physical Anthropology* 151:589-603.
- Kinaston, R., H. Buckley, F. Valentin, S. Bedford, M. Spriggs, S. Hawkins, and E. Herrscher
  2014 Lapita diet in remote Oceania: New stable isotope evidence from the 3000-year-old
  Teouma Site, Efate Island, Vanuatu. *PLoS ONE* 9(3): e90376.
  doi:10.1371/journal.pone.0090376.

Kinaston, R.L., B. Shaw, A.R. Gray, R.K. Walter, C. Jacomb, E. Brooks, S.E. Halcrow, and H.R. Buckley

2015 Critical review of Brown and Thomas "The first New Zealanders? An alternative interpretation of the stable isotope data from Wairau Bar." *PloS ONE* 10(10): e0137616. doi:10.1371/journal.pone.0137616.

# Kirch, P.V.

1973 Prehistoric subsistence patterns in the northern Marquesas Islands, French Polynesia. *Archaeology and Physical Anthropology in Oceania* 8:24-40.

#### Kirch, P.V.

1983 Man's role in modifying tropical and subtropical Polynesian ecosystems. *Archaeology in Oceania* 18(1):26-31.

#### Kirch, P.V.

1984 The Polynesian Outliers: Continuity, change, and replacement. *The Journal of Pacific History* 19(4):224-238.

#### Kirch, P.V.

1986 Rethinking East Polynesian prehistory. Journal of the Polynesian Society 95:9-40.

# Kirch, P.V.

1994 The Wet and the Dry. University of Chicago Press, Chicago.

#### Kirch, P.V.

1996 Late Holocene human-induced modifications to a Central Polynesian island ecosystem. *Proceedings of the National Academy of Sciences of the United States of America* 93(11):5296-5300.

# Kirch, P.V.

1997a The Lapita Peoples: Ancestors of the Oceanic World. Blackwell, Oxford.

#### Kirch, P.V.

1997b Microcosmic histories: Island perspectives on "global" change. *American Anthropologist* 99(1):30-42.

## Kirch, P.V.

2000a On the Road of the Winds: An Archaeological History of the Pacific Islands before European Contact. University of California Press, Berkeley, California.

## Kirch, P.V.

2000b Pigs, humans, and trophic competition on small Oceanic islands. In *Australian Archaeologist: Collected Papers in Honour of Jim Allen*, edited by A.J. Anderson and T. Murray, pp. 427-439. Coombs Academic Publishing, Canberra.

#### Kirch, P.V.

2004 Oceanic islands: Microcosms of "global change." In *The Archaeology of Global Change: The Impact of Humans on their Environment*, edited by C.L. Redman, S.R. James, P. Fish, and J.D. Rogers, pp. 13-27. Smithsonian, Washington, DC.

# Kirch, P.V.

2007a Three islands and an archipelago: Reciprocal interactions between humans and island ecosystems in Polynesia. *Earth and Environmental Science Transactions of the Royal Society of Edinburgh* 98:85-99.

# Kirch, P.V.

2007b Hawaii as a model system for human ecodynamics. *American Anthropologist* 109(1):8-26.

# Kirch, P.V.

2011a Roots of Conflict: Soils, Agriculture, and Sociopolitical Complexity in Ancient Hawaii. School for Advanced Research Press, New Mexico.

# Kirch, P.V.

2011b When did the Polynesians settle Hawai'i? A review of 150 years of scholarly inquiry and a tentative answer. *Hawaiian Archaeology* 12:1-26.

# Kirch, P.V.

2015 Report on archaeological excavations at "Kitchen Cave" (Site 190-04-KAM-1), Kamaka Island, Gambier Archipelago, French Polynesia, 2014. Report prepared for le Service de la Culture et du Patrimoine, Polynésie Française, January 2015.

# Kirch, P.V. and R.C. Green

2001 Hawaiki, Ancestral Polynesia: An Essay in Historical Anthropology. Cambridge University Press, Cambridge.

#### Kirch, P.V. and T.L. Hunt (eds.)

1997 *Historical Ecology in the Pacific Islands: Prehistoric Environmental and Landscape Change.* Yale University Press, New Haven.

# Kirch, P.V. and S.J. O'Day

2003 New archaeological insights into food and status: A case study from pre-contact Hawaii. *World Archaeology* 34:484-97.

#### Kirch, P.V. and D.E. Yen

1982 *Tikopia: The Prehistory and Ecology of a Polynesian Outlier*. Bernice P. Bishop Museum Bulletin 238, Honolulu.

# Kirch, P.V., D.W. Steadman, V.L. Butler, J. Hather, and M.I. Weisler 1995 Prehistory and human ecology in eastern Polynesia: excavations at Tangatatau Rockshelter, Mangaia, Cook Islands. *Archaeology in Oceania* 30(2):47–65.

Kirch, P.V., E. Conte, W. Sharp, and C. Nickelsen

2010 The Onemea Site (Taravai Island, Mangareva) and the human colonization of Southeastern Polynesia. *Archaeology in Oceania* 45:66-79.

Kirch, P.V., G. Molle, C. Nickelsen, P. Mills, E. Dotte-Sarout, J. Swift, A. Wolfe and M. Horrocks

2015 Human ecodynamics in the Mangareva Islands: A stratified sequence from Nenega-iti Rock Shelter (Site AGA-3, Agakauitai Island). *Archaeology in Oceania* 50(1):23-42.

Koch, P.L., M.L. Fogel, and N. Tuross

1994 Tracing the diets of fossil animals using stable isotopes. In *Stable Isotopes in Ecology and Environmental Science*, edited by K. Lajtha and R.H. Michener, pp. 63-92. Blackwell Scientific Publications, New York.

#### Kramer, R.J.

1971 Hawaiian Land Mammals. Charles E. Tuttle Co., Tokyo.

Kurashima, N. and P.V. Kirch

2011 Geospatial modeling of pre-contact Hawaiian production systems on Molokai Island, Hawaiian Islands. *Journal of Archaeological Science* 38:3662-3674.

# Ladefoged, T.N., M.W. Graves, and M.D. McCoy

2003 Archaeological evidence for agricultural development in Kohala, Island of Hawaii. *Journal of Archaeological Science* 30:923-940.

Ladefoged, T.N., M.D. McCoy, G.P. Asner, P.V. Kirch, C.O. Puleston, A.O. Chadwick, and P.M. Vitousek

2011 Agricultural potential and actualized development in Hawaii: An airborne LiDAR survey of the leeward Kohala field system (Hawaii Island). *Journal of Archaeological Science* 38:3605-3619.

#### Leach, H.M.

2003 Did East Polynesians have a concept of luxury foods? *World Archaeology* 34:442-457.

Leach, B.F., C.J. Quinn, and G.L. Lyon

1996 A stochastic approach to the reconstruction of prehistoric human diet in the Pacific region from bone isotope signatures. *Tuhinga: Records of the Museum of New Zealand Te Papa Tongarewa* 8:154.

LeDuc, S.D., D.E. Rothstein, Z. Yermakov, and S.E. Spaulding

2013 Jack pine foliar  $\delta^{15}$ N indicates shifts in plant nitrogen acquisition after severe wildfire and through forest stand development. *Plant Soil* 373:955-65.

#### Lee-Thorp, J.A.

2008 On isotopes and old bones. Archaeometry 50(6):925-50.

Lee-Thorp, J.A., J.C. Sealy, and N.J. van der Merwe

1989 Stable carbon isotope ratio differences between bone collagen and bone apatite, and their relationship to diet. *Journal of Archaeological Science* 16:585-599.

Lepofsky, D., P.V. Kirch, and K.P. Lertzman

1996 Stratigraphic and paleobotanical evidence for prehistoric human-induced environmental disturbance on Mo'orea, French Polynesia. *Pacific Science* 50(3):253-273.

## Longin, R.

1971 New method of collagen extraction for radiocarbon dating. *Nature* 230:241-242.

## Luomala, K.

1960 The native dog in the Polynesian system of values. In *Culture in History: Essays in Honor of Paul Radin*, edited by S. Dimond, pp. 190-240. Columbia University Press, New York.

Major, H.L., I.L. Jones, M.R. Charette, and A.W. Diamond

2007 Variations in the diet of introduced Norway rats (*Rattus norvegicus*) inferred using stable isotope analysis. *Journal of Zoology* 271:463-468.

# Matisoo-Smith, E.

1994 The human colonization of Polynesia. A novel approach: Genetic analyses of the Polynesian rat (*Rattus exulans*). *Journal of the Polynesian Society* 103(1):75-87.

# Matisoo-Smith, E.

2002 Something old, something new: Do genetic studies of contemporary populations reliably represent prehistoric populations in the Pacific? *Human Biology* 74(3):489-496.

## Matisoo-Smith, E.

2009 The commensal model for human settlement of the Pacific 10 years on – what can we say and where to now? *Journal of Island and Coastal Archaeology* 4(2):151-63.

## Matisoo-Smith, E.

2015 Ancient DNA and the human settlement of the Pacific: A Review. *Journal of Human Evolution* 79:93-104.

# Matisoo-Smith, E. and J.H. Robins

2004 Origins and dispersals of Pacific peoples: Evidence from mtDNA phylogenies of the Pacific rat. *PNAS* 101(24):9167-9172.

# Matisoo-Smith, E., J.S. Allen, R.M. Roberts, G.J. Irwin, and D.M. Lambert

1999 Rodents of the sunrise: Mitochondrial DNA phylogenies of Polynesian *Rattus exulans* and the settlement of Polynesia, pp. 259-76 in *The Pacific from 5000 to 2000 BP: Colonization and Transformations*, edited by J.C. Gallipaud and I Lilley, pp. 259-76. IRD, Paris. McGovern-Wilson, R. and C. Quinn

1996 Stable isotope analysis of ten individuals from Afetna, Saipan, Northern Mariana Islands. *Journal of Archaeological Science* 23:59-65.

Mead, S.M., L. Birks, H. Birks, and E. Shaw

1975 *The Lapita Pottery Style of Fiji and its Associations*. The Polynesian Society, Wellington.

Mertz, O., T.B. Bruun, B. Fog, K. Rasmussen, and J. Agergaard

2002 Sustainable land use in Tikopia: food production and consumption in an isolated agricultural system. *Singapore Journal of Tropical Geography* 31:10-26.

Meyer, J.Y. and J.F. Butaud

2009 The impacts of rats on the endangered native flora of French Polynesia (Pacific Islands): Drivers of plant extinction or *coup de grâce* species? *Biological Invasions* 11:1569-1585.

Mieth, A. and H.R. Bork

2010 Humans, climate, or introduced rats – which is to blame for the woodland destruction on prehistoric Rapa Nui (Easter Island)? *Journal of Archaeological Science* 37, 417-426.

#### Minagawa, M. and E. Wada

1984 Stepwise enrichment of <sup>15</sup>N along food chains: Further evidence and the relation between  $\delta^{15}$ N and animal age. *Geochimica et Cosmochimica Acta* 48(5):1135-1140.

## Mizota, C., and A. Naikatini

2007 Nitrogen isotope composition of inorganic soil nitrogen and associated vegetation under a sea bird colony on the Hatana Islands, Rotuma Group, Fiji. *Geochemical Journal* 49:297-301.

# Molle, G.

2011 Ua Huka, une île dans l'histoire. Histoire pré- et post-européenne d'une société Marquisienne. Ph.D. Dissertation, Université de la Polynésie Française, Puna'auia (French Polynesia).

#### Moller, H. and J.L. Craig

1987 The population ecology of *Rattus exulans* on Tiritiri Matangi Island, and a model of comparative population dynamics in New Zealand. *New Zealand Journal of Zoology* 14(3):305-326.

Morand, S., F. Bordes, K. Blasdell, S. Pilosof, J.F. Cornu, K. Chaisiri, Y. Chaval, J.F. Cosson, J. Claude, T. Feyfant, V. Herbreteau, S. Dupuy and A. Tran

2015 Assessing the distribution of disease-bearing rodents in human-modified tropical landscapes. *Journal of Applied Ecology* 52(3):784-789.

# Norman, F.I.

1975 The murine rodents *Rattus rattus, exulans,* and *norvegicus* as avian predators. *Atoll Research Bulletin* 182. Smithsonian Institution, Washington, DC.

# O'Day, S.J.

2004 The socioeconomics of rank in late prehistoric and contemporary Fiji: An exploration of ethnographic and archaeological indicators. Ph.D. Dissertation, University of Florida, Gainesville.

# Oliver, D.L.

2002 Polynesia in Early Historic Times. Bess Press, Honolulu.

# Pestle, W.J. and M. Colvard

2012 Bone collagen preservation in the tropics: A case study from ancient Puerto Rico. *Journal of Archaeological Science* 39(7):2079-2090.

# Pestle, W.J., B.E. Crowley, and M.T. Weirauch

2014 Quantifying inter-laboratory variability in stable isotope analysis of ancient skeletal remains. *PLoS ONE*: 9(7): e102844. doi:10.1371/journal.pone.0102844.

# Peterson, B.J. and B. Fry

1987 Stable isotopes in ecosystem studies. *Annual Review of Ecological Systems* 18:293-320.

Petchey, F.J., M. Spriggs, S. Bedford, F. Valentin, and H. Buckley

2014 Radiocarbon dating of burials from the Teouma Lapita cemetery, Efate, Vanuatu. *Journal of Archaeological Science* 50:227-242.

## Porch, N.

2008 Invertebrates in landscape archaeology. In *Handbook to Landscape Archaeology*, edited by B. David and J. Thomas, pp. 457-467. Left Coast Press, Walnut Creek.

## Quintus, S., M.S. Allen, and T.N. Ladefoged

2016 In surplus and scarcity: Agricultural development, risk management, and political economy on Ofu Island, American Samoa. *American Antiquity* 81(2):273-293.

## Richards, M.P., B.T. Fuller, and R. Hedges

2001 Sulphur isotopic variation in ancient bone collagen from Europe: Implications from human palaeodiet, residence mobility, and modern pollutant studies. *Earth and Planetary Science Letters* 191:185-190.

Richards, M.P., E. West, B. Rolett, and K. Dobney

2009 Isotope analysis of human and animal diets from the Hanamiai archaeological site (French Polynesia). *Archaeology in Oceania* 44:29-37.

Roberts, M.

1991 Origin, dispersal routes, and geographic distribution of *Rattus exulans* (Peale) with special reference to New Zealand. *Pacific Science* 45:123-130.

# Rolett, B.V.

1992 Faunal extinctions and depletions linked with prehistory and environmental change in the Marquesas Islands (French Polynesia). *Journal of the Polynesian Society* 101(1):86-94.

## Rolett, B.V.

1998 Hanamiai: Prehistoric Colonization and Cultural Change in the Marquesas Islands, East Polynesia. Yale University Press, New Haven.

# Rolett, B.V. and J. Diamond

2004 Environmental predictors of pre-European deforestation on Pacific Islands. *Nature* 431:443-446.

# Ruffino, L., J.C. Russell, B. Pisanu, S. Caut, and E. Vidal

2011 Low individual-level dietary plasticity in an island-invasive generalist forager. *Population Ecology* 53:535-548.

# Ruffino, L., D. Zarzoso-Lacoste, and E. Vidal

2015 Assessment of invasive rodent impacts on island avifauna: Methods, limitations, and the way forward. *Wildlife Research* 42(2):185-195.

# Russell, J.C. and M.N. Clout

2005 Modelling the distribution and interaction of introduced rodents on New Zealand offshore islands. *Global Ecology and Biogeography* 13(6):497-507.

#### Russell, J.C., S. Caut, S.H. Anderson, and M. Lee

2015 Invasive rat interactions and over-invasion on a coral atoll. *Biological Conservation* 185:59-65.

#### Schoeninger, M.J. and M.J. DeNiro

1984 Nitrogen and carbon isotopic composition of bone collagen from marine and terrestrial animals. *Geochimica et Cosmochimica Acta* 48(4):1381-3.

#### Schwarcz, H.P. and M.J. Schoeninger

1991 Stable isotope analyses in human nutritional ecology. *Yearbook of Physical Anthropology* 34:283-321.

# Sealy, J.

2001 Body tissue chemistry and palaeodiet. In *Handbook of Archaeological Sciences*, edited by D. Brothwell and A.M. Pollard, pp. 269-279. John Wiley and Sons, Chichester.

Sealy, J., M. Johnson, M. Richards, and O. Nehlich

2014 Comparison of two methods of extracting bone collagen for stable carbon and nitrogen isotope analysis: Comparing whole bone demineralization with gelatinization and ultrafiltration. *Journal of Archaeological Science* 47:64-69.

#### Sharp, Z.

2007 *Principles of Stable Isotope Geochemistry*. Pearson Education, Inc., Upper Saddle River, New Jersey.

#### Shiels, A.B. and D.R. Drake

- 2015 Barriers to seed and seedling survival of once-common Hawaiian palms: The role of invasive rats and ungulates. *AoB PLANTS* 7: plv057 doi:10.1093/aobpla/plv057.
- Shiels, A.B., C.A. Flores, A. Khamsing, P.D. Krushelnycky, S.M. Mosher, and D.R. Drake
  2013 Dietary niche differentiation among three species of invasive rodents (*Rattus rattus*, *R. exulans, Mus musculus*). *Biological Invasions* 15:1037-1048.

## Sinoto, Y.H.

1996 A tentative prehistoric cultural sequence in the northern Marquesas Islands, French Polynesia. *Journal of the Polynesian Society* 75:287-303.

# Sinoto, Y.H.

1970 An archaeologically based assessment of the Marquesas Islands as a dispersal center in East Polynesia. In *Studies in Oceanic Culture History*, edited by R.C. Green and M. Kelly, pp. 105-132. Pacific Anthropological Records 11. Bernice P. Bishop Museum Press, Honolulu.

# Sinoto, Y.H. and M. Kellum

1965 Preliminary report on excavations in the Marquesas Islands, French Polynesia. Report on file, Bernice P. Bishop Museum, Honolulu.

#### Spennemann, D.H.R.

1997 Distribution of rat species (*Rattus* spp.) on the atolls of the Marshall Islands: Past and present dispersal. *Atoll Research Bulletin* 446:1-18.

#### Spriggs, M. and A. Anderson

1993 Late colonization of East Polynesia. Antiquity 67:200-217.

## Stantis, C., R.L. Kinaston, M.P. Richards, J.M. Davidson, and H.R. Buckley

2015 Assessing human diet and movement in the Tongan maritime kingdom using isotopic analyses. *PLoS ONE* 10(3): e0123156. doi:10.1371/journal.pone.0123156.

#### Steadman, D.W.

1993 Biogeography of Tongan birds before and after human impact. *PNAS* 90:818-822.

#### Steadman, D.W.

1995 Prehistoric extinctions of Pacific Island birds: Biodiversity meets zooarchaeology. *Science* 267(5201):1123-1131.

# Steadman, D.W.

1997 Extinctions of Polynesian birds: Reciprocal impacts of birds and people. In *Historical Ecology in the Pacific Islands: Prehistoric Environmental and Landscape Change*, edited by P.V. Kirch and T.L. Hunt, pp. 51-79. Yale University Press, New Haven.

# Steadman, D.W.

2006 *Extinction & Biogeography of Tropical Pacific Birds*. University of Chicago Press, Chicago.

# Steadman, D.W. and L.J. Justice

1998 Prehistoric exploitation of birds on Mangareva, Gambier Islands, French Polynesia. *Man and Culture in Oceania* 14:81-98.

# Steadman, D.W. and P.V. Kirch

1990 Prehistoric extinction of birds on Mangaia, Cook Islands, Polynesia. *PNAS* 87:9605-9609.

Steadman, D.W. and B.V. Rolett

1996 A chronostratigraphic analysis of landbird extinction on Tahuata, Marquesas Islands. *Journal of Archaeological Science* 23(1):81-94.

# Steadman, D.W., D.S. Pahlavan, and P.V. Kirch

1990 Extinction, biogeography, and human exploitation of birds on Tikopia and Anuta, Polynesian outliers in the Solomon Islands. *Bishop Museum Occasional Papers* 30:118-153.

Steffen, W., J. Grinevald, and P. Crutzen

2011 The Anthropocene: Conceptual and historical perspectives. *Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences* 369:842-867.

Stevens, R.E., R. Jacobi, M. Street, M. Germonpré, N.J. Conard, S.C. Münzel, and R.E.M. Hedges

2008 Nitrogen isotope analyses of reindeer (*Rangifer tarandus*), 45000 BP to 9000 BP: Palaeoenvironmental reconstructions. *Paleogeography, Paleoclimatology, Paleoecology* 262:32-45.

Storey, A.A., M. Spriggs, S. Bedford, S.C. Hawkins, J.H. Robins, L. Huynek, and E. Matisoo-Smith

2010 Mitochondrial DNA from 3000-year old chickens at the Teouma site, Vanuatu. *Journal of Archaeological Science* 37(10):2459-68.

Suggs, R.C.

1961 *The Archaeology of Nuku Hiva, Marquesas Islands, French Polynesia.* Anthropological Papers of the American Museum of Natural History 49, New York.

# Sweeney, M., M.W. Graves, and T.L. Hunt

1993 A reappraisal of evidence for subsistence change at the Hane dune site, Marquesas Islands, French Polynesia. *Asian Perspectives* 32:225-238.

# Szpak, P.

2014 Complexities of nitrogen isotope biogeochemistry in plant-soil systems: Implications for the study of ancient agricultural and animal management practices. *Frontiers in Plant Science* 5:1-19.

Szpak, P., F.J. Longstaffe, J.F. Millaire, and C.D. White

2012 Stable isotope biogeochemistry of seabird guano fertilization: Results from growth chamber studies with maize (*Zea mays*). *PLoS ONE* 7(3): e33741. doi:10.1371/journal.pone.0033741.

# Tamarin, R.H. and S.R. Malecha

1972 Reproductive parameters in *Rattus rattus* and *Rattus exulans* of Hawaii, 1968 to 1970. Journal of Mammalogy 53(3):513-528.

# Tate, G.H.H.

1935 Rodents of the genera *Rattus* and *Mus* from the Pacific Islands, collected by the Whitney South Sea Expedition with a discussion of the origins and races of the Pacific Island rat. *Bulletin of the American Museum of Natural History* 68:145-178.

#### Taylor, R.H.

1975 What limits kiore (*Rattus exulans*) distribution in New Zealand? *New Zealand Journal of Zoology* 2(4):473-77.

#### Taylor, J.M., J.H. Calaby, and H.M. Van Deusen

1982 A revision of the genus *Rattus* (Rodentia: Muridae) in the New Guinean region. *Bulletin of the American Museum of Natural History* 173:177-336.

# Tieszen, L. and T. Fagre

1993 Effect of diet quality on the isotope composition of respiratory CO2, bone collagen, apatite, and soft tissues. In *Molecular Archaeology of Prehistoric Human Bone*, edited by J.B. Lambert and B. Grupe, pp. 121-55. Springer, Berlin.

#### Tieszen, L.L., T.W. Boutton, K.G. Tesdahl, and N.A. Slade

1983 Fractionation and turnover of stable carbon isotopes in animal tissues: Implications for  $\delta^{13}$ C analysis of diet. *Oecologia* 57:32-37.

#### Titcomb, M.

1969 *Dog and Man in the Ancient Pacific*. Bernice P. Bishop Museum Bulletin 59, Honolulu.

Towns, D.R., I.A.E. Atkinson, and C.H. Daugherty

2006 Have the harmful effects of introduced rats on islands been exaggerated? *Biological Invasions* 8:863-891.

# Tuross, N.

2002 Alterations in fossil collagen. Archaeometry 44(3):427-34.

Vaiglova, P., C. Snoeck, E. Nitsch, A. Bogaard, J. Lee-Thorp

2014 Impact of contamination and pre-treatment on stable carbon and nitrogen isotopic composition of charred plant remains. *Rapid Communications in Mass Spectrometry* 28(23):2497-2510.

## Valentin, F., H. Bocherens, B. Gratuze, and C. Sand

2006 Dietary patterns during the late prehistoric/historic period in Cikobia Island (Fiji): Insights from stable isotopes and dental pathologies. *Journal of Archaeological Science* 33:1396-1410.

Valentin, F., H.R. Buckley, E. Herrscher, R. Kinaston, S. Bedford, M. Spriggs, S. Hawkins, and K. Neal

2010 Lapita subsistence strategies and food consumption patterns in the community of Teouma (Efate, Vanuatu). *Journal of Archaeological Science* 37(8):1820-1829.

## Valentin, F., E. Herrscher, F. Petchey, and D.J. Addison

2011 An analysis of the last 1000 years human diet on Tutuila (American Samoa) using carbon and nitrogen stable isotope data. *American Antiquity* 76(3):473-486.

# van Klinken, G.J.

1999 Bone collagen quality indicators for paleodietary and radiocarbon measurements. *Journal of Archaeological Science* 26:687-695.

# Vayda, A.P., A. Leeds, and D.B. Smith

1961 The place of pigs in Melanesian subsistence. In *Proceedings of the 1961 Annual Spring Meeting of the American Ethnological Society*, edited by V.E. Garfield, pp. 69-77. University of Washington Press, Seattle.

## Vitousek, P.M.

2002 Oceanic islands as model systems for ecological studies. *Journal of Biogeography* 29:573-82.

# Vitousek, P.M.

2004 *Nutrient Cycling and Limitation: Hawaii as a Model System.* Princeton University Press, Princeton.

Vitousek, P. and O. Chadwick

2013 Pacific islands in the Anthropocene. *Elementa: Science of the Anthropocene* 1:000011. doi: 10.12952/journal.elementa.000011.

Wainwright, S.C., J.C. Haney, C. Kerr, A.N. Golovkin, and M.V. Flint

1998 Utilization of nitrogen derived from seabird guano by terrestrial and marine plants at St. Paul, Pribilof Islands, Bering Sea, Alaska. *Marine Biology* 131:63-71.

Warren, B.H., D. Simberloff, R.E. Ricklefs, R. Aguilée, F.L. Condamine, D. Gravel, H. Morlon, N. Mouquet, J. Rosindell, J. Casquet, E. Conti, J. Cornuault, J.M. Fernández-Palacios, T. Hengl,

S.J. Norder, K.F. Rijsdijk, I. Sammartin, D. Strasberg, K.A. Triantis, L.M. Valentine, R.J.

Whittaker, R.G. Gillespie, B.C. Emerson, and C. Thébaud

2015 Islands as model systems in ecology and evolution: Prospects fifty years after MacArthur-Wilson. *Ecology Letters* 18:200-17.

Waugh, S., J. Champeau, S. Cranwell, and L. Faulquier

2013 Seabirds of the Gambier Archipelago, French Polynesia, in 2010. *Marine Ornithology* 41:7-12.

Weisler, M.I. and R.C. Green

2013 Fishing strategies in regional context: An analysis of fish bones from five sites excavated in 1959. *Journal of Pacific Archaeology* 4:73-89.

## West, E.

2007 An archaeological investigation of pig husbandry on Tahuata Island, Marquesas, French Polynesia. Ph.D. Dissertation, Department of Anthropology, University of Hawaii Mānoa.

# Whistler, W.A.

2009 Plants of the Canoe People: An Ethnobotanical Voyage through Polynesia. University of Hawaii Press, Honolulu.

## White, J.P., G. Clark, and S. Bedford

2000 Distribution, present and past, of *Rattus praetor* in the Pacific and its implications. *Pacific Science* 54(2):105-117.

# Wickler, S., and M. Spriggs

1988 Pleistocene human occupation of the Solomon Islands, Melanesia. *Antiquity* 62:703-706.

# Williams, J.M.

1973 The ecology of *Rattus exulans* (Peale) reviewed. *Pacific Science* 27(2):120-127.

# Wilmshurst, J.M., A.J. Anderson, T.F.G. Higham, and T.H. Worthy

2008 Dating the late prehistoric dispersal of Polynesians to New Zealand using the commensal Pacific rat. *Proceedings of the National Academy of Sciences* 105:7676-7680.

Wolfe, M.S. and L. Klein

1996 Sex differences in absolute rates of bone resorption in young rats: Appendicular versus axial bones. *Calcified Tissue International* 59:51-57.

Worthy, T.W. and A.J.D. Tennyson

2004 Avifaunal assemblages from Nenega-iti and Onemea. In *Archaeological Investigations in the Mangareva Islands (Gambier Archipelago), French Polynesia*, edited by E. Conte and P.V. Kirch, pp. 122-127. Archaeological Research Facility Contribution 62, Berkeley.

Yen, D.E.

1973 The origins of Oceanic agriculture. *Archaeology and Physical Anthropology in Oceania* 8(1):68-85.

Yoshinaga, J., M. Minagawa, T. Suzuki, R. Ohtsuka, T. Kawabe, T. Hongo, T. Inaoka, and T. Akimichi

1991 Carbon and nitrogen isotopic characterization for Papua New Guinea foods. *Ecology* of Food and Nutrition 26(1):17-25.

Young, H.S., D.J. McCauley, R.B. Dunbar, and R. Dirzo

2010 Plants cause ecosystem nutrient depletion via the interruption of bird-derived spatial subsidies. *PNAS* 107(5):2072-2077.

Zalasiewicz, J., P.J. Crutzen, and W. Steffen

2012 The Anthropocene. In *The Geologic Time Scale 2012*, edited by F.M. Gradstein, J.C. Ogg, and M. Schmitz, pp. 1033-1040. Elsevier, Oxford.

Zangrando, A.F., A. Tessone, A. Ugan, and M.A. Gutiérrez

2014 Applications of stable isotope analysis in Zooarchaeology: An introduction. *International Journal of Osteoarchaeology* 24:127-133.